Filed by Polestar Automotive Holding UK Limited
Pursuant to Rule 425 under the Securities Act of 1933
and deemed filed pursuant to 14a-12 under the
Securities Exchange Act of 1934
Form F-4 File No. (333-260992)
Subject Companies:
Polestar Automotive Holding UK Limited
Gores Guggenheim, Inc.
(Commission File No. 001-40265)

# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

# FORM 8-K

# CURRENT REPORT PURSUANT TO SECTION 13 OR 15(D) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported): December 17, 2021

# GORES GUGGENHEIM, INC.

(Exact name of registrant as specified in its charter)

Delaware
(State or other jurisdiction of incorporation)

001-40265 (Commission File Number) 85-4385646 (I.R.S. Employer Identification No.)

6260 Lookout Road Boulder, CO (Address of principal executive offices)

80301 (Zip Code)

(310) 209-3010 (Registrant's telephone number, including area code)

Not Applicable (Former name or former address, if changed since last report)

Check the appropriate box below if the Form 8-K is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

|   | Written communication pursuant to Rule 425 under the Securities Act (17 CFR 230.425)                    |  |
|---|---|--|
| X   | Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)                  |  |
|   | Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))  |  |
|   | Pre-commencements communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c)) |  |
| curities registered pursuant to Section 12(b) of the Act: |   |  |

| Title of each class  | Trading<br>Symbols | Name of each exchange on which registered |
|----------------------|--------------------|---|
| Class A Common Stock | GGPI               | Nasdaq Capital Market                     |
| Warrants             | GGPIW              | Nasdaq Capital Market                     |
| Units                | GGPIU              | Nasdaq Capital Market                     |

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

#### Item 1.01 Entry into a Material Definitive Agreement

#### **Amendment to Business Combination Agreement**

As previously disclosed, on September 27, 2021, Gores Guggenheim, Inc. (the "Company") entered into a Business Combination Agreement (the "Business Combination Agreement"), by and among the Company, Polestar Automotive Holding Limited, a Hong Kong incorporated company ("Parent"), Polestar Automotive (Singapore) Pte. Ltd., a private company limited by shares in Singapore, Polestar Holding AB, a private limited liability company incorporated under the laws of Sweden, Polestar Automotive Holding UK Limited, a limited company incorporated under the laws of England and Wales and a direct wholly owned subsidiary of Parent ("ListCo"), and PAH UK Merger Sub Inc., a Delaware corporation and a direct wholly owned subsidiary of ListCo.

On December 17, 2021, the parties to the Business Combination Agreement entered into Amendment No. 1 to the Business Combination Agreement (the "BCA Amendment"). The BCA Amendment amends the Business Combination Agreement to: (i) account for the amendments to the Sponsor Subscription Agreement (as defined below) and the Volvo Cars Subscription Agreement (as defined below) and the execution of the New PIPE Subscription Agreements related thereto (as defined below); (ii) reflect the amendment to the Sponsor and Supporting Sponsor Stockholders Lock-Up Agreement (as defined below); (iii) reflect the amendment to the Registration Rights Agreement (as defined below) and (iv) make other administrative and conforming amendments to the Business Combination Agreement.

The foregoing description of the BCA Amendment does not purport to be complete and is qualified in its entirety by the terms and conditions of the BCA Amendment, a copy of which is attached hereto as Exhibit 2.1 and is incorporated herein by reference.

#### New PIPE Subscription Agreements; Amendment of Sponsor Subscription Agreement and Volvo Cars Subscription Agreement

As previously disclosed, on September 27, 2021, the Company and ListCo entered into a subscription agreement (the "Sponsor Subscription Agreement") with Gores Guggenheim Sponsor LLC (the "Sponsor") pursuant to which the Sponsor agreed to subscribe for approximately 9.08 million American depository shares of ListCo (the "ListCo Class A ADSs") for a purchase price of \$9.09 per ListCo Class A ADS on the date of closing (the "Closing") of the transactions contemplated by the Business Combination Agreement and the other transaction documents contemplated thereby (the "Business Combination"), for an aggregate investment amount of approximately \$82.5 million. Pursuant to the Sponsor Subscription Agreement, the Sponsor had the right to assign its commitment to purchase the ListCo Class A ADSs under the Sponsor Subscription Agreement in advance of the Closing. On December 17, 2021, (i) the Sponsor assigned a portion of its commitment to purchase ListCo Class A ADSs, in an aggregate investment amount equaling approximately \$63.0 million (the "Sponsor Assignment"), to certain investors and (ii) the Company, ListCo and Sponsor amended the Sponsor Subscription Agreement to reflect the Sponsor Assignment. As a result, pursuant to the Sponsor Subscription Agreement, as amended, Sponsor has agreed to subscribe for approximately 2.15 million ListCo Class A ADSs for a purchase price of \$9.09 per ListCo Class A ADS on the date of Closing, for an aggregate investment of approximately \$19.5 million. The Sponsor Subscription Agreement, as amended, is substantially similar to the PIPE Subscription Agreements (as defined below), except with regards to purchase price and that the Sponsor has the right to assign its commitment to purchase the ListCo Class A ADSs under the Sponsor Subscription Agreement in advance of the Closing.

As previously disclosed, on September 27, 2021, the Company and ListCo entered into a subscription agreement (the "Volvo Cars Subscription Agreement") with Snita Holding B.V., a corporation organized under the laws of Netherlands ("Snita"), pursuant to which Snita agreed to subscribe for 10.00 million ListCo Class A ADSs for a purchase price of \$10.00 per Class A ADS on the date of Closing, for an aggregate investment of \$100.0 million. Pursuant to the Volvo Cars Subscription Agreement, Snita had the right to assign its commitment to purchase the ListCo Class A ADSs under the Volvo Cars Subscription Agreement in advance of the Closing. On December 17, 2021 (i) Snita assigned a portion of its commitment to purchase ListCo Class A ADSs, in an aggregate investment amount equaling approximately \$73.0 million (the "Volvo Assignment," and together with the Sponsor Assignment, the "PIPE Assignment") to purchase the ListCo Class A ADSs to certain investors (the

investors who were assigned commitments pursuant to the PIPE Assignment, collectively, the "New PIPE Investors") and (ii) the Company, ListCo and Snita amended the Volvo Car Subscription Agreement to reflect the Volvo Assignment. As a result, pursuant to the Volvo Cars Subscription Agreement, as amended, Snita has agreed to subscribe for approximately 2.70 million ListCo Class A ADSs for a purchase price of \$10.00 per ListCo Class A ADS on the date of Closing, for an aggregate investment of approximately \$27.0 million. The Volvo Cars Subscription Agreement, as amended, is substantially similar to the PIPE Subscription Agreements, except with regards to purchase price and that Snita may, in accordance with the terms of the Volvo Cars Subscription Agreement, assign its commitment to purchase the ListCo Class A ADSs under the Volvo Cars Subscription Agreement in advance of the Closing.

As previously disclosed, on September 27, 2021, the Company and ListCo entered into subscription agreements (the "PIPE Subscription Agreements") with certain investors (the "PIPE Investors"), pursuant to which the PIPE Investors have agreed to purchase, substantially concurrently on the date of the Closing, an aggregate of 7.43 million ListCo Class A ADSs for an aggregate amount of approximately \$67.5 million. In connection with the PIPE Assignment, on December 17, 2021, the Company and ListCo entered into subscription agreements (the "New PIPE Subscription Agreements") with the New PIPE Investors, which include certain affiliates and employees of Sponsor. Pursuant to the New PIPE Subscription Agreements, the New PIPE Investors have agreed to collectively subscribe for approximately 14.3 million ListCo Class A ADSs (the "New PIPE Shares") for an average price of approximately \$9.54 per ListCo Class A ADS, reflecting an aggregate investment amount of approximately \$136.0 million. The New PIPE Subscription Agreements are substantially similar to the PIPE Subscription Agreements. The aggregate amount of the PIPE investment and number of ListCo Class A ADSs to be purchased pursuant thereto remains unchanged.

The issuance of the New PIPE Shares pursuant to the New PIPE Subscription Agreements is contingent upon, among other customary closing conditions, the substantially concurrent consummation of the Business Combination. Pursuant to the New PIPE Subscription Agreements, ListCo agreed to file with the U.S. Securities and Exchange Commission (the "SEC") (at ListCo's sole cost and expense), within 30 calendar days after the date of Closing, a registration statement registering the resale of the New PIPE Shares, and to use its commercially reasonable efforts to have the registration statement declared effective as soon as practicable after the filing thereof.

The foregoing description of the New PIPE Subscription Agreements does not purport to be complete and is qualified in its entirety by the terms and conditions of the PIPE Subscription Agreements, a form of which is attached hereto as Exhibit 10.1 and is incorporated herein by reference.

#### Amendment to the Registration Rights Agreement

As previously disclosed, on September 27, 2021, ListCo, Parent, the Parent Shareholders (as defined in the Business Combination Agreement), Sponsor and the independent directors of the Company entered into a Registration Rights Agreement (the "Registration Rights Agreement").

On December 17, 2021, the parties to the Registration Rights Agreement entered into Amendment No. 1 to the Registration Rights Agreement (the "<u>Registration Rights Agreement Amendment</u>"), to provide for certain administrative changes to reflect the BCA Amendment and the New PIPE Subscription Agreements.

The foregoing description of the Registration Rights Agreement Amendment does not purport to be complete and is qualified in its entirety by the terms and conditions of the Registration Rights Agreement Amendment, a copy of which is attached hereto as Exhibit 10.2 and is incorporated herein by reference.

#### Amendment to the Sponsor and Supporting Stockholders Lock-Up Agreement

As previously disclosed, on September 27, 2021, Sponsor, the Company, Parent, ListCo and certain of the Company's directors, executive officers and affiliates entered into the Sponsor and Supporting Sponsor Stockholders Lock-Up Agreement (the "Sponsor and Supporting Sponsor Stockholders Lock-Up Agreement").

On December 17, 2021, the parties to the Sponsor and Supporting Sponsor Stockholders Lock-Up Agreement entered into Amendment No. 1 to the Sponsor and Supporting Sponsor Stockholders Lock-Up Agreement (the "Lock-Up Agreement Amendment"). The Lock-Up Agreement Amendment provides for amendments to the Sponsor and Supporting Sponsor Stockholders Lock-Up Agreement to increase the amount of the Company's Class F common stock ("Company Class F Common Stock") that will be cancelled by the Company in connection with the Closing from 1,501,651 shares of Company Class F Common Stock to 1,533,873 shares of Company Class F Common Stock.

The foregoing description of the Lock-Up Agreement Amendment does not purport to be complete and is qualified in its entirety by the terms and conditions of the Lock-Up Agreement Amendment, a copy of which is attached hereto as Exhibit 10.3 and is incorporated herein by reference.

#### **Forward-Looking Statements**

Certain statements in this Current Report may be considered "forward-looking statements" as defined in the Private Securities Litigation Reform Act of 1995. Forward-looking statements generally relate to future events or the future financial or operating performance of the Company and Polestar Performance AB and its affiliates ("Polestar"). For example, projections of future revenue, volumes and other metrics are forward-looking statements. In some cases, you can identify forward-looking statements by terminology such as "may", "should", "expect", "intend", "will", "estimate", "anticipate", "believe", "predict", "potential", "forecast", "plan", "seek", "future", "propose" or "continue", or the negatives of these terms or variations of them or similar terminology. Such forward-looking statements are subject to risks, uncertainties, and other factors which could cause actual results to differ materially from those expressed or implied by such forward looking statements.

These forward-looking statements are based upon estimates and assumptions that, while considered reasonable by the Company and its management, and Polestar and its management, as the case may be, are inherently uncertain. Factors that may cause actual results to differ materially from current expectations include, but are not limited to: (a) the occurrence of any event, change or other circumstances that could give rise to the termination of definitive agreements with respect to proposed Business Combination; (b) the outcome of any legal proceedings that may be instituted against the Company, the combined company or others following the announcement of the Business Combination and any definitive agreements with respect thereto; (c) the inability to complete the Business Combination due to the failure to obtain approval of the stockholders of the Company, to obtain financing to complete the Business Combination or to satisfy other conditions to Closing; (d) changes to the proposed structure of the Business Combination that may be required or appropriate as a result of applicable laws or regulations or as a condition to obtaining regulatory approval of the Business Combination; (e) the ability to meet stock exchange listing standards following the consummation of the Business Combination; (f) the risk that the Business Combination disrupts current plans and operations of Polestar as a result of the announcement and consummation of the Business Combination; (g) the ability to recognize the anticipated benefits of the Business Combination, which may be affected by, among other things, competition, the ability of the combined company to grow and manage growth profitably, maintain relationships with customers and suppliers and retain its management and key employees; (h) costs related to the Business Combination; (i) risks associated with changes in applicable laws or regulations and Polestar's international operations; (j) the possibility that Polestar or the combined company may be adversely affected by other economic, business, and/or competitive factors; (k) Polestar's estimates of expenses and profitability; (l) Polestar's ability to maintain agreements or partnerships with its strategic partners Volvo Cars and Zhejiang Geely Holding Group Co., Ltd and to develop new agreements or partnerships; (m) Polestar's ability to maintain relationships with its existing suppliers and strategic partners, and source new suppliers for its critical components, and to complete building out its supply chain, while effectively managing the risks due to such relationships; (n) Polestar's reliance on its partnerships with vehicle charging networks to provide charging solutions for its vehicles and its strategic partners for servicing its vehicles and their integrated software; (o) Polestar's ability to establish its brand and capture additional market share, and the risks associated with negative press or reputational harm, including from lithium-ion battery cells catching fire or venting smoke; (p) delays in the design, manufacture, launch and financing of Polestar's vehicles and Polestar's reliance on a limited number of vehicle models to generate revenues; (q) Polestar's ability to continuously and rapidly innovate, develop and market

new products; (r) risks related to future market adoption of Polestar's offerings; (s) increases in costs, disruption of supply or shortage of materials, in particular for lithium-ion cells or semiconductors; (t) Polestar's reliance on its partners to manufacture vehicles at a high volume, some of which have limited experience in producing electric vehicles, and on the allocation of sufficient production capacity to Polestar by its partners in order for Polestar to be able to increase its vehicle production capacities; (u) risks related to Polestar's distribution model; (v) the effects of competition and the high barriers to entry in the automotive industry, and the pace and depth of electric vehicle adoption generally on Polestar's future business; (w) changes in regulatory requirements, governmental incentives and fuel and energy prices; (x) the impact of the global COVID-19 pandemic on the Company, Polestar, Polestar's post business combination's projected results of operations, financial performance or other financial metrics, or on any of the foregoing risks; and (y) other risks and uncertainties set forth in the section entitled "Risk Factors" and "Cautionary Note Regarding Forward-Looking Statements" in the Company's final prospectus relating to its initial public offering (File No. 333-253338) declared effective by the SEC on March 22, 2021, and other documents filed, or to be filed, with the SEC by the Company or ListCo, including the Registration/Proxy Statement (as defined below). There may be additional risks that neither the Company, Polestar nor ListCo presently know or that the Company, Polestar or ListCo currently believe are immaterial that could also cause actual results to differ from those contained in the forward-looking statements. Nothing in this Current Report should be regarded as a representation by any person that the forward-looking statements set forth herein will be achieved or that any of the contemplated results of such forward-looking statements will be achieved. You should not

#### **Additional Information**

In connection with the proposed Business Combination, (a) ListCo has filed with the SEC a registration statement on Form F-4 containing a preliminary proxy statement of the Company and a preliminary prospectus (the "Registration/Proxy Statement") and (b) the Company will file a definitive proxy statement relating to the proposed Business Combination (the "Definitive Proxy Statement") and will mail the Definitive Proxy Statement and other relevant materials to its stockholders after the Registration/Proxy Statement is declared effective. The Registration/Proxy Statement contains and the Definitive Proxy Statement will contain important information about the proposed Business Combination and the other matters to be voted upon at a meeting of the Company's stockholders to be held to approve the proposed Business Combination. This Current Report does not contain all the information that should be considered concerning the proposed Business Combination and is not intended to form the basis of any investment decision or any other decision in respect of the Business Combination. Before making any voting or other investment decisions, securityholders of the Company and other interested persons are advised to read the Registration/Proxy Statement and the amendments thereto and the Definitive Proxy Statement and other documents filed in connection with the proposed Business Combination, as these materials will contain important information about the Company, Polestar, ListCo and the Business Combination. When available, the Definitive Proxy Statement and other relevant materials for the proposed Business Combination will be mailed to stockholders of the Company as of a record date to be established for voting on the proposed Business Combination. Stockholders will also be able to obtain copies of the Registration/Proxy Statement, the Definitive Proxy Statement and other documents filed with the SEC, without charge, once available, at the SEC's website at www.sec.gov, or by directing a request to: Gores Guggenheim, Inc.,

INVESTMENT IN ANY SECURITIES DESCRIBED HEREIN HAS NOT BEEN APPROVED OR DISAPPROVED BY THE SEC OR ANY OTHER REGULATORY AUTHORITY NOR HAS ANY AUTHORITY PASSED UPON OR ENDORSED THE MERITS OF THE OFFERING OR THE ACCURACY OR ADEQUACY OF THE INFORMATION CONTAINED HEREIN. ANY REPRESENTATION TO THE CONTRARY IS A CRIMINAL OFFENSE.

#### **Participants in the Solicitation**

The Company and its directors and executive officers may be deemed participants in the solicitation of proxies from the Company's stockholders with respect to the proposed Business Combination. A list of the names of

those directors and executive officers and a description of their interests in the Company is set forth in the Company's filings with the SEC (including the Company's final prospectus related to its initial public offering (File No. 333-253338) declared effective by the SEC on March 22, 2021), and are available free of charge at the SEC's web site at www.sec.gov, or by directing a request to Gores Guggenheim, Inc., 6260 Lookout Rd., Boulder, CO 80301, attention: Jennifer Kwon Chou. Additional information regarding the interests of such participants is contained in the Registration/Proxy Statement and will be contained in the Definitive Proxy Statement.

Polestar and ListCo, and certain of their directors and executive officers may also be deemed to be participants in the solicitation of proxies from the stockholders of the Company in connection with the proposed Business Combination. A list of the names of such directors and executive officers and information regarding their interests in the proposed Business Combination is included in the Registration/Proxy Statement and will be included in the Definitive Proxy Statement.

#### No Offer and Non-Solicitation

This Current Report is not a proxy statement or solicitation of a proxy, consent or authorization with respect to any securities or in respect of the potential transaction and shall not constitute an offer to sell or a solicitation of an offer to buy the securities of the Company, Polestar or ListCo, nor shall there be any sale of any such securities in any state or jurisdiction in which such offer, solicitation, or sale would be unlawful prior to registration or qualification under the securities laws of such state or jurisdiction. No offer of securities shall be made except by means of a prospectus meeting the requirements of the Securities Act of 1933, as amended.

#### Item 9.01 Financial Statements and Exhibits.

#### (d) Exhibits

| Exhibit<br>No. | <u>Exhibit</u>   |
|----------------|--|
| 2.1            | BCA Amendment, dated December 17, 2021.  |
| 10.1           | Form Subscription Agreement (previously filed as Exhibit 10.1 to the Company's Current Report on Form 8-K, filed with the Securities and Exchange Commission on September 27, 2021). |
| 10.2           | Registration Rights Agreement Amendment, dated December 17, 2021 (included as Annex I to Exhibit 2.1).   |
| 10.3           | Lock-Up Agreement Amendment, dated December 17, 2021 (included as Annex III to Exhibit 2.1).   |
| 104            | Cover Page Interactive Data File (embedded within the Inline XBRL document).   |

# SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Gores Guggenheim, Inc.

Date: December 17, 2021 By: /s/ Andrew McBride

Name: Andrew McBride

Title: Chief Financial Officer and Secretary

#### AMENDMENT NO. 1 TO BUSINESS COMBINATION AGREEMENT

THIS AMENDMENT NO. 1 TO BUSINESS COMBINATION AGREEMENT (this "Amendment No. 1") is made and entered into as of December 17, 2021, by and among Gores Guggenheim, Inc., a Delaware corporation ("GG"), Polestar Automotive Holding Limited, a Hong Kong incorporated company ("Parent"), Polestar Automotive (Singapore) Pte. Ltd., a private company limited by shares in Singapore ("Polestar Singapore"), Polestar Holding AB, a private limited liability company incorporated under the laws of Sweden ("Polestar Sweden"), Polestar Automotive Holding UK Limited, a limited company incorporated under the laws of England and Wales and a direct wholly owned subsidiary of Parent ("ListCo"), and PAH UK Merger Sub Inc., a Delaware corporation and a direct wholly owned subsidiary of ListCo ("Merger Sub," and together with GG, Parent, the Company and ListCo, the "Parties" and each individually as a "Party") and amends that certain Business Combination Agreement, dated as of September 27, 2021, by and among the Parties (the "Business Combination Agreement"). Except as otherwise set forth herein, capitalized terms used herein have the meanings set forth in the Business Combination Agreement.

#### RECITALS

WHEREAS, the Parties entered into the Business Combination Agreement on September 27, 2021;

WHEREAS, concurrently with the execution of this Amendment No. 1, (a) Sponsor is assigning to certain investors certain of its obligations to purchase ListCo Class A ADSs under, and Sponsor and certain of its Affiliates, GG and ListCo are amending, that certain subscription agreement dated September 27, 2021, by and among such parties, (b) Snita Holding B.V. is assigning to certain investors certain of its obligations to purchase ListCo Class A ADSs under, and Snita Holding B.V., GG and ListCo are amending, that certain subscription agreement dated September 27, 2021, by and among such parties and (c) such investors referenced in clauses (a) and (b), GG and ListCo are entering into certain subscription agreements dated December 17, 2021, to reflect such assignment in each case by and among the applicable investors and parties;

WHEREAS, concurrently with the execution of this Amendment No. 1, Sponsor, GG, ListCo and the Supporting Sponsor Stockholders named as parties thereto are entering into that certain Amendment No. 1 to the Sponsor and Supporting Sponsor Stockholders Lock-Up Agreement;

WHEREAS, concurrently with the execution of this Amendment No. 1, ListCo, Parent, Sponsor and the other holders named as parties thereto are entering into that certain Amendment No. 1 to the Registration Rights Agreement; and

WHEREAS, the Parties have determined to amend certain provisions of the Business Combination Agreement and certain exhibits thereto in furtherance of the consummation of the Transactions.

NOW, THEREFORE, in consideration of the covenants, promises and representations set forth herein and in the Business Combination Agreement, and for other good

and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties agree as follows:

#### 1. The Amendments.

- 1.1 <u>Replacement of Exhibit B</u>. Exhibit B to the Business Combination Agreement is hereby replaced in its entirety with (a) a new Exhibit B-1 with Amendment No. 1 to the Sponsor and Supporting Sponsor Stockholders Lock-Up Agreement, dated as of the date hereof and attached to this Amendment No. 1 as <u>Annex I</u> and (b) a new Exhibit B-2 with the Sponsor and Supporting Sponsor Stockholders Lock-Up Agreement, dated September 27, 2021 attached to this Amendment No. 1 as <u>Annex II</u>.
- 1.2 <u>Replacement of Exhibit C</u>. Exhibit C to the Business Combination Agreement is hereby replaced in its entirety with (a) a new Exhibit C-1 with Amendment No. 1 to the Registration Rights Agreement, dated as of the date hereof and attached to this Amendment No. 1 as <u>Annex III</u> and (b) a new Exhibit C-2 with the Registration Rights Agreement, dated September 27, 2021 attached to this Amendment No. 1 as <u>Annex IV</u>.

#### 1.3 Amendment to Exhibit E.

1.3.1 Paragraph 3 of Exhibit E to the Business Combination Agreement is hereby amended and restated in its entirety to read as follows:

Prior to the Closing, ListCo shall (i) amend and restate its articles of association to be in substantially the form attached as Exhibit F to the Agreement, subject to such further amendments as may be agreed between GG and ListCo, (ii) re-register with the Companies House as a public limited company incorporated under the laws of England and Wales and (iii) issue GBP redeemable preferred shares wholly in cash with a nominal value of GBP50,000 to Parent (or such other person as Parent and Gores Guggenheim, Inc. may agree in writing) (collectively, the "ListCo Re-registration").

1.3.2 Paragraph 4 of Exhibit E to the Business Combination Agreement is hereby amended and restated in its entirety to read as follows:

Following the completion of the ListCo Re-registration and prior to the consummation of the PIPE Investment, Parent shall contribute to ListCo all of the issued and outstanding Equity Securities of Polestar Sweden (the "<u>Pre-Closing Consolidation</u>"), and in exchange for the Pre-Closing Consolidation, ListCo shall issue to Parent the Parent Consideration.

- 1.3.3 There shall be an addition of Paragraph 5 to Exhibit E to the Business Combination Agreement as follows:
- 5. The PIPE Investment shall be consummated following the Pre-Closing Consolidation and immediately prior to the Closing.

- 1.4 <u>Replacement of Exhibit F</u>. Exhibit F to the Business Combination Agreement is hereby replaced in its entirety with a new Form of ListCo New Articles of Association attached to this Amendment No. 1 as <u>Annex V</u>.
  - 1.5 Deletion of Exhibit J. Exhibit J to the Business Combination Agreement is hereby deleted in its entirety.
  - 1.6 Amendment to Recitals.
- 1.6.1 The fourth WHEREAS clause of the Recitals of the Business Combination Agreement is hereby amended and restated in its entirety to read as follows:

WHEREAS, concurrently with the execution of this Agreement, Gores Guggenheim Sponsor LLC, a Delaware limited liability company ("Sponsor"), each independent director of GG (collectively, the "Supporting Sponsor Stockholders"), GG, Parent and ListCo are entering into the lock-up agreement (as may be amended from time to time, the "Sponsor and Supporting Sponsor Stockholders Lock-Up Agreement") attached hereto as Exhibits B-1 and B-2, pursuant to which, among other things, each of Sponsor and the Supporting Sponsor Stockholders is agreeing to (a) support and vote in favor of all of the Transaction Proposals, (b) waive all adjustments to the conversion ratio set forth in GG's Governing Documents with respect to its GG Class F Shares, (c) be bound by certain transfer restrictions with respect to their GG Shares and GG Warrants, as applicable, prior to Closing, (d) be bound by certain lock-up provisions during the lock-up period described in the Sponsor and Supporting Sponsor Stockholders Lock-Up Agreement with respect to the ListCo Class A ADSs issued pursuant to this Agreement, subject to the terms and conditions set forth in the Sponsor and Supporting Stockholders Lock-Up Agreement and (e) with respect to Sponsor only, the forfeiture of up to 1,533,873 GG Class F Shares under the circumstances specified therein:

1.6.2 The fifth WHEREAS clause of the Recitals of the Business Combination Agreement is hereby amended and restated in its entirety to read as follows:

WHEREAS, concurrently with the execution of this Agreement, Sponsor and certain of its Affiliates (collectively with Sponsor, the "<u>Sponsor Insiders</u>"), GG and ListCo are entering into a subscription agreement (as may be amended from time to time, the "<u>Sponsor Subscription Agreement</u>") pursuant to which, among other things, the Sponsor Insiders are agreeing to subscribe for and accept, and ListCo is agreeing to issue to the Sponsor Insiders, on the Closing Date, ListCo Class A ADSs in exchange for the subscription price per ListCo Class A ADS set forth in the Sponsor Subscription Agreement, on the terms and subject to the conditions set forth in the Sponsor Subscription Agreement (such investment, the "<u>Sponsor Investment</u>," and the aggregate dollar amount of the Sponsor Investment, the "<u>Sponsor Investment Amount</u>");

1.6.3 The sixth WHEREAS clause of the Recitals of the Business Combination Agreement is hereby amended and restated in its entirety to read as follows:

WHEREAS, concurrently with the execution of this Agreement, Snita Holding B.V., a private limited company organized under the laws of the Netherlands ("Snita") and a direct, wholly owned subsidiary of Volvo Car Corporation, a corporation organized under the laws of Sweden ("VCC", which shall be understood to refer to Snita as the context may suggest), GG and ListCo are entering into a subscription agreement (as may be amended from time to time, the "VCC PIPE Subscription Agreement") pursuant to which, among other things, VCC is agreeing to subscribe for and accept, and ListCo is agreeing to issue to VCC, on the Closing Date, ListCo Class A ADSs in exchange for the subscription price per ListCo Class A ADS set forth in the VCC PIPE Subscription Agreement, on the terms and subject to the conditions set forth in the VCC PIPE Subscription Agreement (such investment, the "VCC PIPE Investment," and the aggregate dollar amount of the VCC PIPE Investment, the "VCC PIPE Investment, the "VCC PIPE Investment, the "VCC PIPE Investment,"

1.6.4 The seventh WHEREAS clause of the Recitals of the Business Combination Agreement is hereby amended and restated in its entirety to read as follows:

WHEREAS, concurrently with and subsequent to the execution of this Agreement, certain investors (collectively, the "PIPE Investors"), GG and ListCo are entering into subscription agreements (each, as may be amended from time to time, a "PIPE Subscription Agreement," and together with the Sponsor Subscription Agreement and the VCC PIPE Subscription Agreement, the "Subscription Agreements") pursuant to which, among other things, the PIPE Investors are agreeing to subscribe for and accept on the Closing Date, and ListCo is agreeing to issue to each such PIPE Investor on the Closing Date, the number of ListCo Class A ADSs set forth in the applicable PIPE Subscription Agreement, in each case, on the terms and subject to the conditions set forth in the applicable PIPE Subscription Agreement (such investment in the aggregate, the "PIPE Investment," and the aggregate dollar amount of the PIPE Investment, the "PIPE Investment Amount");

1.6.5 The twelfth WHEREAS clause of the Recitals of the Business Combination Agreement is hereby amended and restated in its entirety to read as follows:

WHEREAS, concurrently with the execution of this Agreement, ListCo, Sponsor, Parent, the Parent Shareholders and certain other Persons are entering into a registration rights agreement (as may be amended from time to time, the "Registration Rights Agreement"), attached hereto as Exhibits C-1 and C-2, pursuant to which, among other things, Sponsor, the Supporting Sponsor Stockholders, Parent, each Parent Shareholder and certain other Persons shall be granted certain registration rights with respect to their respective ListCo Class A ADSs, ListCo Class A Shares, ListCo AD Warrants or ListCo Class C-2 ADSs

issued to them pursuant to this Agreement or the other Transaction Documents, in each case, on the terms and subject to the conditions set forth in the Registration Rights Agreement;

#### 1.7 Amendments to Article I.

1.7.1 Section 1.01 of the Business Combination Agreement is hereby amended to restate the definition of "Note Conversion Shares" in its entirety as follows:

"Note Conversion Shares" means an aggregate number, rounded down to the nearest whole number with respect to each recipient, of ListCo Class A Shares, to be issued in the form of ListCo Class A ADSs, equal to (a) the outstanding principal amount due in respect of the Convertible Notes, divided by (b) the applicable Conversion Price as defined in the applicable Convertible Note.

1.7.2 Section 1.01 of the Business Combination Agreement is hereby amended to restate the definition of "PIPE Investment Amount" in its entirety as follows:

"PIPE Investment Amount" has the meaning specified in the Recitals hereto.

1.7.3 Section 1.01 of the Business Combination Agreement is hereby amended to restate the definition of "Sponsor Investment Amount" in its entirety as follows.

"Sponsor Investment Amount" has the meaning specified in the Recitals hereto.

1.7.4 Section 1.01 of the Business Combination Agreement is hereby amended to restate the definition of "VCC Preference Shares" in its entirety as follows:

"VCC Preference Shares" means the aggregate number of convertible preference shares of ListCo issued pursuant to the VCC Preference Subscription Agreement, outstanding as of the Closing and convertible into ListCo Class A Shares to be issued the form of ListCo Class A ADSs.

1.8 <u>Amendment to Section 4.04(e)</u>. Section 4.04(e) of the Business Combination Agreement is hereby amended and restated in its entirety to read as follows:

As of (i) December 17, 2021, GG and ListCo have entered into the PIPE Subscription Agreements with PIPE Investors, pursuant to which, and on the terms and subject to the conditions of which, such PIPE Investors have agreed, in connection with the Transactions, to purchase ListCo Class A ADSs for a PIPE Investment equal to the PIPE Investment Amount, (ii) the date of this Agreement, GG, the Sponsor Insiders and ListCo have entered into the Sponsor Subscription Agreement, pursuant to which, and on the terms and subject to the conditions of which, the Sponsor Investment Amount, (iii) the date of this Agreement, GG, VCC and ListCo have entered into the VCC PIPE Subscription Agreement, pursuant to which, and on the terms and subject to the conditions of which, VCC has agreed, in connection with

the Transactions, to purchase ListCo Class A ADSs for a VCC PIPE Investment equal to the VCC PIPE Investment Amount and (iv) the date of this Agreement, ListCo and Snita have entered into the VCC Preference Subscription Agreement, pursuant to which, on the terms and subject to the condition of which, Snita has agreed, in connection with the Transactions, to subscribe for VCC Preference Shares. The Subscription Agreements and the VCC Preference Subscription Agreement are in full force and effect with respect to, and binding on, ListCo and, to Parent's knowledge, on each other party thereto, in accordance with their terms.

 $1.9 \frac{Amendment to Section 5.04(c)}{5.04(c)}$ . Section 5.04(c) of the Business Combination Agreement is hereby amended and restated in its entirety to read as follows:

As of (i) December 17, 2021, GG and ListCo have entered into the PIPE Subscription Agreements with PIPE Investors, pursuant to which, and on the terms and subject to the conditions of which, such PIPE Investors have agreed, in connection with the Transactions, to purchase ListCo Class A ADSs for a PIPE Investment equal to the PIPE Investment Amount, (ii) the date of this Agreement, GG, the Sponsor Insiders and ListCo have entered into a Sponsor Subscription Agreement, pursuant to which, and on the terms and subject to the conditions of which, the Sponsor Investment Amount and (iii) the date of this Agreement, GG, VCC and ListCo have entered into a VCC PIPE Subscription Agreement, pursuant to which, and on the terms and subject to the conditions of which, VCC has agreed, in connection with the Transactions, to purchase ListCo Class A ADSs for a VCC PIPE Investment equal to the VCC PIPE Investment Amount. The Subscription Agreements are in full force and effect with respect to, and binding on, GG and, to the knowledge of GG, on each PIPE Investor party thereto, in accordance with their terms.

1.10 Amendment to Section 6.08. Section 6.08 of the Business Combination Agreement is hereby amended and restated in its entirety to read as follows:

Prior to the Closing, Parent shall, and shall cause ListCo, the Company and their respective Subsidiaries to, complete a reorganization, in accordance with the steps and transactions set forth on <a href="Exhibit E">Exhibit E</a> attached hereto, with such modifications as Parent and GG may agree in writing (not to be unreasonably withheld, conditioned or delayed by either Party) (such transactions, the "<a href="Pre-Closing Reorganization">Pre-Closing Reorganization</a>"). Pursuant to, and in accordance with, the Pre-Closing Reorganization, Parent shall cause ListCo to adopt new articles of association to be in substantially the form attached hereto as Exhibit F (the "<a href="ListCo New Articles of Association">ListCo New Articles of Association</a>"), subject to such further amendments as may be agreed between GG and ListCo.

2. Effect of Amendment. This Amendment No. 1 shall be effective as of the date first written above. For the avoidance of any doubt, all references: (a) in the Business Combination Agreement to "this Agreement" and (b) to the Business Combination Agreement in any other agreements, exhibits and schedules will, in each case, be deemed to be references to the Business Combination Agreement as amended by this Amendment No. 1. Except as amended hereby, the Business

Combination Agreement will continue in full force and effect and shall be otherwise unaffected hereby. This Amendment No. 1 shall be binding upon and shall inure to the benefit of the Parties and their respective successors and assigns. This Amendment No. 1 only may be amended by the Parties by execution of an instrument in writing signed on behalf of each of the Parties.

3. <u>Incorporation by Reference</u>. Sections 1.02 (Construction); 11.03 (Assignment); 11.06 (Governing Law); 11.07 (Captions; Counterparts); 11.09 (Entire Agreement); 11.10 (Amendments); 11.11 (Severability); 11.12 (Jurisdiction); 11.13 (Waiver of Jury Trial) and 11.14 (Enforcement) of the Business Combination Agreement are incorporated herein and shall apply to this Amendment No. 1 *mutatis mutandis*.

[Signature pages follow]

IN WITNESS WHEREOF, the Parties have caused this Amendment No. 1 to be executed as of the date first written above.

# GORES GUGGENHEIM, INC.

By: /s/ Mark Stone

Name: Mark Stone

Title: Chief Executive Officer

#### PAH UK MERGER SUB INC.

By: /s/ Mikael Alkmark
Name: Jan Mikael Alkmark

Title: Director

#### POLESTAR AUTOMOTIVE HOLDING LIMITED

By: /s/ Lars Denielson
Name: Lars Danielson
Title: Director

# POLESTAR AUTOMOTIVE (SINGAPORE) PTE. LTD.

By: /s/ Alex Ang
Name: Alex Ang
Title: Director

By: /s/ Mikael Alkmark
Name: Jan Mikael Alkmark

Title: Director

[Signature Page to Amendment No. 1 to Business Combination Agreement]

# POLESTAR HOLDING AB (SWEDEN)

By: /s/ Johan Malmqvist
Name: Johan Malmqvist

Title: Director

By: /s/ Anna Rudensjö

Name: Anna Rudensjö Title: Director

# POLESTAR AUTOMOTIVE HOLDING UK LIMITED

By: /s/ Mikael Alkmark
Name: Jan Mikael Alkmark

Title: Director

[Signature Page to Amendment No. 1 to Business Combination Agreement]

# ANNEX I

# Amendment No. 1 to the

 $\underline{Sponsor\ and\ Supporting\ Sponsor\ Stockholders\ Lock-Up\ Agreement}$ 

#### AMENDMENT NO. 1 TO THE SPONSOR AND SUPPORTING SPONSOR STOCKHOLDERS LOCK-UP AGREEMENT

This AMENDMENT NO. 1 TO THE SPONSOR AND SUPPORTING SPONSOR STOCKHOLDERS LOCK-UP AGREEMENT (this "Amendment No. 1"), dated as of December 17, 2021, is made by and among Gores Guggenheim Sponsor LLC, a Delaware limited liability company ("Sponsor"), Gores Guggenheim, Inc., a Delaware corporation ("GG"), Polestar Automotive Holding Limited, a Hong Kong Company ("Parent"), Polestar Automotive Holding UK Limited, a limited company incorporated under the laws of England and Wales and a direct wholly owned subsidiary of Parent ("ListCo"), and each of the undersigned directors of GG (each a "Supporting Sponsor Stockholder" and collectively, the "Supporting Sponsor Stockholders") and amends that certain Sponsor and Supporting Sponsor Stockholders Lock-Up Agreement, dated as of September 27, 2021 (the "Lock-Up Agreement"), by and among Sponsor, GG, Parent, ListCo and the Supporting Sponsor Stockholders (the "Parties"). Capitalized terms used but not otherwise defined herein shall have the meanings ascribed to such terms in the Business Combination Agreement (as defined in the Lock-Up Agreement).

#### RECITALS

WHEREAS, the Parties entered into the Lock-Up Agreement on September 27, 2021;

WHEREAS, the Parties have determined to amend certain provisions of the Lock-Up Agreement.

WHEREAS, concurrently with the execution of this Amendment No. 1, (a) Sponsor is assigning to certain investors its obligations to purchase ListCo Class A ADSs under, and Sponsor and certain of its Affiliates, GG and ListCo are amending, that certain subscription agreement dated September 27, 2021, by and among such parties, (b) Snita is assigning to certain investors its obligations to purchase ListCo Class A ADSs under, and Snita, GG and ListCo are amending, that certain subscription agreement dated September 27, 2021, by and among such parties and (c) such investors referenced in clauses (a) and (b), GG and ListCo are entering into certain subscription agreements dated December 17, 2021, to reflect such assignment in each case by and among the applicable investors and parties; and

WHEREAS, concurrently with the execution and delivery hereof, the Parties are entering into that certain Amendment No. 1 to Business Combination Agreement dated as of December 17, 2021, by and among the Parties.

NOW THEREFORE, in consideration of the covenants and agreements contained herein, and certain other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

#### 1. The Amendments.

a. The third recital of the Lock-Up Agreement is hereby replaced in its entirety as follows:

WHEREAS, the Business Combination Agreement contemplates that the Parties will enter into this Agreement concurrently with the entry into the Business Combination Agreement by the parties thereto, pursuant to which, among other things, each of Sponsor and each Supporting Sponsor Stockholder is agreeing to (a) support and vote in favor of all of the Transaction Proposals, (b) waive all adjustments to the conversion ratio set forth in GG's Governing Documents with respect to its GG Class F Shares, (c) be bound by certain transfer restrictions with respect to their GG Shares and GG Warrants, as applicable, prior to Closing, (d) be bound by certain lock-up provisions during the Lock-Up Period (as defined herein) with respect to the ListCo Class A ADSs issued pursuant to the Business Combination Agreement in respect of the GG Class F Shares held by the Sponsor or the Supporting Sponsor Stockholders immediately prior to the Closing (the "ListCo Covered Shares"), subject to the terms and conditions set forth herein and (e) with respect to Sponsor only, the forfeiture of up to 1,533,873 GG Class F Shares under the circumstances specified herein.

b. Section 4 is replaced in its entirety as follows:

Section 4. <u>Surrender of Shares</u>. "Sponsor agrees that, subject to the satisfaction or waiver of each of the conditions to Closing set forth in Section 9.01 and 9.03 of the Business Combination Agreement, immediately prior to the Closing, Sponsor shall surrender 1,533,873 shares of GG Class F Shares (the "<u>Surrendered Shares</u>"), which Surrendered Shares will be cancelled by GG."

- 2. Effect of Amendment. This Amendment No. 1 shall be effective as of the date first written above. For the avoidance of any doubt, all references: (a) in the Lock-Up Agreement to "this Agreement" and (b) to the Lock-Up Agreement in any other agreements, exhibits and schedules will, in each case, be deemed to be references to the Lock-Up Agreement as amended by this Amendment No. 1. Except as amended hereby, the Lock-Up Agreement will continue in full force and effect and shall be otherwise unaffected hereby. This Amendment No. 1 shall be binding upon and shall inure to the benefit of the Parties and their respective successors and assigns. This Amendment only may be amended by the Parties by execution of an instrument in writing signed on behalf of each of the Parties.
- 3. Incorporation by Reference. Sections 1.02 (Construction); 11.03 (Assignment); 11.06 (Governing Law); 11.07 (Captions; Counterparts); 11.09 (Entire Agreement); 11.10 (Amendments); 11.11 (Severability); 11.12 (Jurisdiction); 11.13 (Waiver of Jury Trial) and 11.14 (Enforcement) of the Business Combination Agreement are incorporated herein and shall apply to this Amendment No. 1 *mutatis mutandis*.

[Signature pages follow]

**IN WITNESS WHEREOF**, each of the Parties has caused this Amendment No. 1 to be duly executed on its behalf as of the day and year first above written.

#### GORES GUGGENHEIM SPONSOR LLC

By: GG SPONSOR, LLC Its: Managing Member

By: AEG Holdings, LLC

Its: Manager

By: /s/ Alex Gores
Name: Alec Gores

Name: Alec Gores Title: Chairman

#### GORES GUGGENHEIM SPONSOR LLC

By: GGP SPONSOR HOLDINGS, LLC

Its: Managing Member

By: /s/ Andrew Rosenfield

Name: Andrew Rosenfield

Title: Manager

# GORES GUGGENHEIM, INC.

By: /s/ Mark Stone

Name: Mark Stone

Title: Chief Executive Officer

 $[Signature\ Page\ to\ Amendment\ No.\ 1\ to\ Sponsor\ and\ Supporting\ Sponsor\ Stockholders\ Lock-Up\ Agreement]$ 

# POLESTAR AUTOMOTIVE HOLDING LIMITED

By: /s/ Lars Danielson

Name: Lars Danielson Title: Director

#### POLESTAR AUTOMOTIVE HOLDING UK LIMITED

By: /s/ Mikael Alkmark

Name: Jan Mikael Alkmark

Title: Director

 $[Signature\ Page\ to\ Amendment\ No.\ 1\ to\ Sponsor\ and\ Supporting\ Sponsor\ Stockholders\ Lock-Up\ Agreement]$ 

# SUPPORTING SPONSOR STOCKHOLDER

|       |                  | Address for Notice: |
|-------|------------------|---------------------|
| By:   | /s/ Randall Bort |                     |
| Name: | Randall Bort     |                     |
|       |                  | Attention:          |
|       |                  | Facsimile:          |
|       |                  | E-mail·             |

[Signature Page to Amendment No. 1 to Sponsor and Supporting Sponsor Stockholders Lock-Up Agreement]

# SUPPORTING SPONSOR STOCKHOLDER

| By: | /s/ Elizabeth Marcellino | Address for Notice: |
|-----|--------------------------|---------------------|
|     | Elizabeth Marcellino     |                     |
|     |                          | Attention:          |
|     |                          | Facsimile:          |
|     |                          | E-mail:             |

[Signature Page to Amendment No. 1 to Sponsor and Supporting Sponsor Stockholders Lock-Up Agreement]

# SUPPORTING SPONSOR STOCKHOLDER

|       |                  | Address for Notice: |
|-------|------------------|---------------------|
| By:   | /s/ Nancy Tellem |                     |
| Name: | Nancy Tellem     |                     |
|       |                  | Attention:          |
|       |                  | Facsimile:          |
|       |                  | E-mail:             |

[Signature Page to Amendment No. 1 to Sponsor and Supporting Sponsor Stockholders Lock-Up Agreement]

# ANNEX II

<u>Sponsor and Supporting Sponsor Stockholders</u> <u>Lock-Up Agreement, dated September 27, 2021</u>

#### SPONSOR AND SUPPORTING SPONSOR STOCKHOLDERS LOCK-UP AGREEMENT

This SPONSOR AND SUPPORTING SPONSOR STOCKHOLDERS LOCK-UP AGREEMENT (this "Agreement"), dated as of September 27, 2021, is made by and among Gores Guggenheim Sponsor LLC, a Delaware limited liability company ("Sponsor"), Gores Guggenheim, Inc., a Delaware corporation ("GG"), Polestar Automotive Holding Limited, a Hong Kong Company ("Parent"), Polestar Automotive Holding UK Limited, a limited company incorporated under the laws of England and Wales and a direct wholly owned subsidiary of Parent ("ListCo"), and each of the undersigned directors of GG (each a "Supporting Sponsor Stockholder" and collectively, the "Supporting Sponsor Stockholders"). Sponsor, GG, Parent, ListCo and the Supporting Sponsor Stockholders shall be referred to herein from time to time collectively as the "Parties" and each individually as a "Party". Capitalized terms used but not otherwise defined herein shall have the meanings ascribed to such terms in the Business Combination Agreement (as defined below).

WHEREAS, concurrently with the execution of this Agreement, GG, Parent, Polestar Automotive (Singapore) Pte. Ltd., a private company limited by shares in Singapore, Polestar Holding AB, a private limited liability company incorporated under the laws of Sweden, ListCo and PAH UK Merger Sub Inc., a Delaware corporation and a direct wholly owned subsidiary of ListCo, are entering into that certain Business Combination Agreement, dated as of the date hereof (as it may be amended, restated or otherwise modified from time to time in accordance with its terms, the "Business Combination Agreement"), pursuant to which the parties thereto will consummate the Transactions on the terms and subject to the conditions set forth therein; and

WHEREAS, the Business Combination Agreement contemplates that the Parties will enter into this Agreement concurrently with the entry into the Business Combination Agreement by the parties thereto, pursuant to which, among other things, each of Sponsor and each Supporting Sponsor Stockholder is agreeing to (a) support and vote in favor of all of the Transaction Proposals, (b) waive all adjustments to the conversion ratio set forth in GG's Governing Documents with respect to its GG Class F Shares, (c) be bound by certain transfer restrictions with respect to their GG Shares and GG Warrants, as applicable, prior to Closing, (d) be bound by certain lock-up provisions during the Lock-Up Period (as defined herein) with respect to the ListCo Class A ADSs issued pursuant to the Business Combination Agreement in respect of the GG Class F Shares held by the Sponsor or the Supporting Sponsor Stockholders immediately prior to the Closing (the "ListCo Covered Shares"), subject to the terms and conditions set forth herein and (e) with respect to Sponsor only, the forfeiture of up to 1,501,651 GG Class F Shares under the circumstances specified herein.

NOW, THEREFORE, in consideration of the premises and the mutual promises contained herein and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties, each intending to be legally bound, hereby agree as follows:

Section 1. <u>Agreement to Vote</u>; <u>Agreement to Consent</u>. Prior to the Termination Date (as defined herein), each of Sponsor and each Supporting Sponsor Stockholder, in its capacity as a stockholder of GG, irrevocably and unconditionally agrees that: at the meeting of GG's

stockholders to be convened for the purpose of obtaining the requisite stockholder approval of the Transaction Proposals in connection with the Transactions or any other meeting of GG's stockholders (whether annual or special and whether or not an adjourned or postponed meeting, however called and including any adjournment or postponement thereof), such Sponsor and Supporting Sponsor Stockholder shall (a) if and when such meeting is held, appear at such meeting or otherwise cause all Covered Shares (as defined herein) owned by such Sponsor and Supporting Sponsor Stockholder as of the record date of such meeting to be counted as present thereat for the purpose of establishing a quorum; (b) vote, or cause to be voted, at such meeting all of such Sponsor and Supporting Sponsor Stockholder's Covered Shares owned as of the record date for such meeting in favor of each of the Transaction Proposals; (c) vote, or cause to be voted, at such meeting all of such Sponsor and Supporting Sponsor Stockholder's Covered Shares and against any other action that would reasonably be expected to (i) prevent, materially impede, interfere with, delay, postpone, nullify or adversely affect the Transactions, (ii) change in any manner the capitalization of, including the voting rights of any class of Equity Securities of GG other than as contemplated by the Business Combination Agreement or (iii) result in any liquidation, dissolution or other change in GG's corporate structure or business other than as contemplated by the Business Combination Agreement.

The obligations of each of Sponsor and each Supporting Sponsor Stockholder specified in this <u>Section 1</u> shall apply whether or not the Transactions or any action described above are recommended by the GG Board or there is, or is reasonably expected to be, a GG Change of Recommendation or GG Intervening Event. For purposes of this Agreement, "<u>Covered Shares</u>" means all GG Class F Shares held by such Sponsor and Supporting Sponsor Stockholder, as applicable, as of the date hereof together with any GG Class F Shares acquired by such Sponsor and Supporting Sponsor Stockholder, as applicable, after the date hereof.

Section 2. Waiver of Anti-dilution Protection. With respect to its Covered Shares, each of Sponsor and each Supporting Sponsor Stockholder hereby waives and shall refrain from asserting or perfecting, subject to, conditioned upon and effective as of immediately prior to the occurrence of the Closing (for itself and for its successors and assigns), to the fullest extent permitted by Law and the Governing Documents of GG, any rights to adjustment of the conversion ratio with respect to the GG Class F Shares owned by Sponsor set forth in the Governing Documents of GG or otherwise (including the rights set forth in Section 4.3(b) of the Certificate of Incorporation of GG, dated as of December 21, 2020). Notwithstanding anything to the contrary contained herein, Sponsor shall not be prohibited from waiving, asserting or perfecting any of the foregoing rights in the event the Business Combination Agreement is validly terminated in accordance with its terms. If the Business Combination Agreement is so terminated, then this Section 2 shall be deemed null and void *ab initio*.

#### Section 3. Transfer of Shares.

(a) Each of Sponsor and each Supporting Sponsor Stockholder agrees that, during the period from the date hereof through the Termination Date, except as contemplated by this Agreement and the Business Combination Agreement, it shall not, and shall cause its Affiliates not to, without the prior written consent of both GG and Parent (which consent may be given or

withheld by GG and/or Parent in their sole discretion): (i) offer for sale, sell (including short sales), transfer, tender, hypothecate, pledge, convert, encumber, assign or otherwise dispose of, directly or indirectly (including by gift, merger, tendering into any tender offer or exchange offer or otherwise, for avoidance of doubt, including by any other Transfer, as defined in those certain Letter Agreements, dated as of March 25, 2021, by and between GG and each of Sponsor and each Supporting Sponsor Stockholder (the "Letter Agreement")), or enter into any contract, option, derivative, swap, hedging or other agreement or arrangement or understanding (including any profit sharing arrangement) with respect to, or consent to, a transfer to another, in whole or in part, any of the economic consequences of ownership (collectively, a "Transfer"), any or all of its Covered Shares and GG Warrants, (ii) grant any proxies or powers of attorney with respect to any or all of its Covered Shares and GG Warrants held by it (except in connection with voting by proxy at a meeting of stockholders of GG as contemplated in Section 1), or (iii) permit to exist any Lien with respect to any or all of its Covered Shares and GG Warrants other than those created by this Agreement; provided, that any Lien with respect to Covered Shares and GG Warrants that would not prevent, impair or delay its ability to comply with the terms and conditions of this Agreement shall be permitted and shall not be deemed to violate the restrictions contained above. Notwithstanding the foregoing, this Section 3(a) shall not prohibit a Transfer by each of Sponsor and each Supporting Sponsor Stockholder of any of its Covered Shares, (A) in the case of an individual, by bona fide gift to a member of one of the individual's immediate family, to a trust, the beneficiary of which is a member of the individual's immediate family or an Affiliate of such individual; (B) in the case of an individual, by virtue of laws of descent and distribution upon death of the individual; (C) in the case of an individual, pursuant to a qualified domestic relations order; or (D) in the case of an entity, by virtue of such entity's Governing Documents upon liquidation or dissolution of such entity; provided, that, in each case, such Transfer shall be permitted only if, prior to or in connection with such Transfer, the transferee agrees in writing to assume all of the obligations of such Sponsor or such Supporting Sponsor Stockholder hereunder and to be bound by the terms of this Agreement.

(b) Subject to the exceptions set forth herein, each of Sponsor and each Supporting Sponsor Stockholder agrees that, without the prior written consent of the ListCo Board, for a period from the Closing Date through the date that is 180 days after the Closing Date (the "Lock-Up Period"), it shall not, and shall cause its Affiliates not to, (i) Transfer any or all of the ListCo Covered Shares held by Sponsor or such Supporting Sponsor Stockholder, as applicable, immediately after the Closing or (ii) publicly announce any intention to effect any Transfer of ListCo Covered Shares during the Lock-Up Period. Notwithstanding the foregoing, this Section 3(b) shall not apply to, (A) in the case of an entity, pro rata distributions from Sponsor or such Supporting Sponsor Stockholder to any direct or indirect stockholder, partner, member or Affiliate of Sponsor or such Supporting Sponsor Stockholder or to any investment fund or other entity controlling, controlled by, managing or managed by or under common control or management with Sponsor or such Supporting Sponsor Stockholder, (B) in the case of an individual, Transfers by bona fide gift to members of the individual's immediate family or to a trust, the beneficiary of which is a member of such individual's immediate family, to an affiliate of such person or to a charitable organization, (C) in the case of an individual, Transfers by will or by virtue of laws of descent and distribution upon death of the individual, (D) in the case of an individual, Transfers pursuant to a qualified domestic relations order or divorce settlement, (E) in the case of an entity, Transfers by virtue of such entity's Governing Documents upon liquidation or dissolution of such entity, (F) exercise of

any options or warrants to purchase ListCo Covered Shares (which exercises may be effected on a cashless basis to the extent the instruments representing such options or warrants permit exercises on a cashless basis) and (G) acceptance, or execution and delivery, of an irrevocable commitment to or undertaking to accept (without any further agreement to Transfer any Covered Shares or interest therein) a tender offer or similar transaction made by a third party to all of holders of ListCo's Class A Shares in accordance with applicable Law, including Securities Laws, to acquire greater than 50% (which minimum condition shall be non-waivable) of the outstanding ListCo Class A for cash, securities or other property; provided, that in the case of clauses (A) through (E) and clause (G), such Transfers shall be permitted only if, prior to or in connection with such Transfer, the transferee agrees in writing to assume all of the obligations of Sponsor or such Supporting Sponsor Stockholder hereunder and to be bound by the terms of this Section 3(b).

(c) Any Transfer in violation of this Section 3 shall be null and void *ab initio*.

Section 4. <u>Surrender of Shares</u>. Sponsor agrees that, subject to the satisfaction or waiver of each of the conditions to Closing set forth in Sections 9.01 and 9.03 of the Business Combination Agreement, immediately prior to the Closing, Sponsor shall surrender 1,501,651 shares of GG Class F Shares (the "<u>Surrendered Shares</u>"), which Surrendered Shares will be cancelled by GG.

#### Section 5. Redemption; Other Covenants.

- (a) Unless this Agreement shall have been terminated in accordance with <u>Section 7</u>, each Supporting Sponsor Stockholder hereby agrees that such Supporting Sponsor Stockholder shall not effect a GG Stockholder Redemption.
- (b) Each of Sponsor and each Supporting Sponsor Stockholder hereby agrees to be bound by and subject to (and the following sections are hereby incorporated herein, mutatis mutandis) (i) Section 8.04 (Exclusive Dealing) of the Business Combination Agreement to the same extent as such provision applies to GG and (ii) Sections 8.06(a), (d) and (e) (Confidentiality; Publicity) of the Business Combination Agreement, to the same extent as such provision applies to the parties to the Business Combination Agreement.
- (c) Each of GG, Sponsor and each Supporting Sponsor Stockholder agrees that during the period from the date hereof through the Termination Date, it shall not further modify, amend or waive the performance of any provision under the Letter Agreement.

### Section 6. Amendment to the Letter Agreement

- (a) Each of GG, Sponsor and each Supporting Sponsor Stockholder hereby agrees that, effective as of the Closing Date (and not before):
  - (i) paragraph (a) of Section 7 of the Letter Agreement shall be amended and restated in its entirety as follows:

"7. (a) Reserved."

- (ii) paragraph (c) of Section 7 of the Letter Agreement shall be amended to remove all references to paragraph (a) of Section 7 and all references to the Founder Shares (as such term is defined therein); and
- (iii) any provisions related relating to the Private Placement Warrants (as defined therein) or Common Stock (as defined therein) underlying the Private Placement Warrants shall refer to the ListCo Class C-2 ADSs or ListCo AD Warrants, as applicable, issued by ListCo to the holders of Private Placement Warrants in exchange for such warrants as contemplated by the Business Combination Agreement.
- (b) For the avoidance of doubt, the amendment set forth in this <u>Section 6</u> shall be void and of no force and effect with respect to the Letter Agreement if the Business Combination Agreement shall be terminated for any reason in accordance with its terms.

Section 7. Parent Lock-Up Agreement. Notwithstanding anything to the contrary set forth herein, in the event that, prior to the expiration of the Lock-Up Period, either (a) the transfer restrictions set forth in Section 1 of that certain Parent Lock-Up Agreement, dated as of the date hereof, by and among Parent, ListCo and the other parties signatory thereto (the "Parent Lock-Up Agreement"), are waived by ListCo or breached by Parent or another party thereto or (b) ListCo otherwise releases any of the holders party to the Parent Lock-Up Agreement of their respective obligations under Section 1 thereof, then ListCo shall (i) automatically be deemed to have waived or released the obligations of each of the Sponsor and the Supporting Sponsor Stockholders (each such party, a "Sponsor Holder") under Section 3 hereunder, on the same terms as such waiver or release (or breach) under the Parent Lock-Up Agreement, solely, with respect to each Sponsor Holder, for a number of ListCo Covered Shares (rounded down to the nearest whole share) equal to (A)(1) the total number of ListCo Covered Shares held by such Sponsor Holder immediately prior to such waiver or release (or breach), divided by (2) the total number of ListCo Covered Shares multiplied by (B) the total number of ListCo Covered Shares (as defined in the Parent Lock-Up Agreement) subject to the waiver or release (or breach) under the Parent Lock-Up Agreement, and (ii) promptly notify each Sponsor Holder of such event.

Section 8. <u>Termination</u>. This Agreement shall terminate and be void and of no further force and effect, and all rights and obligations of the Parties hereunder shall terminate without any further liability on the part of any Party in respect thereof, upon the earlier to occur of (the "<u>Termination Date</u>") (a) the Closing, (b) such date and time as the Business Combination Agreement is validly terminated in accordance with its terms and (c) the mutual written agreement of the Parties hereto; <u>provided</u>, that nothing herein shall relieve any Party from liability for any breach of this Agreement prior to the Termination Date, and each Party shall be entitled to any remedies at law or in equity to recover Liabilities arising from any such breach. GG shall promptly notify each of Sponsor and each Supporting Sponsor Stockholder of the termination of the Business Combination Agreement promptly after the termination thereof. Notwithstanding the foregoing or anything to the contrary in this Agreement, <u>Sections 2, 3(b) and 7</u> (in each case, solely in the event that this Agreement terminates at Closing as a result of the Closing occurring), <u>5(b)(ii)</u> (solely in the event that this Agreement terminates at Closing as a result of the Parties hereof, and of the Business

Combination Agreement that survive following the Closing) and Sections 12 through 16, shall survive the termination of this Agreement.

Section 9. No Recourse. Subject in all respects to the last sentence of this Section 9, this Agreement may only be enforced against, and any claim or cause of action based upon, arising out of, or relating to, this Agreement may only be brought against, the entities that are expressly named as Parties and then only with respect to the specific obligations set forth herein with respect to such Party. Except to the extent a Party (and then only to the extent of the specific obligations undertaken by such Party in this Agreement), (a) no past, present or future director, officer, employee, incorporator, member, partner, shareholder, Affiliate, agent, attorney, advisor or representative or Affiliate of any Party and (b) no past, present or future director, officer, employee, incorporator, member, partner, shareholder, Affiliate, agent, attorney, advisor or representative or Affiliate of any of the foregoing shall have any Liability (whether in contract, tort, equity or otherwise) for any one or more of the representations, warranties, covenants, agreements or other obligations or liabilities of any one or more of Sponsor, the Supporting Sponsor Stockholders, GG, Parent or ListCo under this Agreement of or for any claim based upon, arising out of, or relating to, this Agreement. Notwithstanding the foregoing, nothing in this Section 9 shall limit, amend or waive any rights or obligations of any party to any other Transaction Document with respect to the other parties thereto.

Section 10. <u>Fiduciary Duties</u>. Notwithstanding anything in this Agreement to the contrary, (a) Sponsor makes no agreement or understanding herein in any capacity other than in Sponsor's capacity as a record holder and beneficial owner of GG Class F Shares, ListCo Class A ADSs, ListCo Class C-1 ADSs, ListCo Class C-2 ADSs, and/or ListCo AD Warrants; (b) each Supporting Sponsor Stockholder makes no agreement or understanding herein in any capacity other than in such Supporting Sponsor Stockholder's capacity as a record holder and beneficial owner of GG Class F Shares and (c) nothing herein shall be construed to limit or affect any action or inaction by any representative of Sponsor or Supporting Sponsor Stockholder serving as a member of the board of directors (or other similar governing body) of GG or as an officer, employee or fiduciary of GG, in each case, acting in such person's capacity as a director, officer, employee or fiduciary of GG.

#### Section 11. Representations and Warranties.

- (a) Each of the Parties represents and warrants that (i) it has the power and authority, or capacity, as the case may be, to enter into this Agreement and to carry out its obligations hereunder, (ii) the execution and delivery of this Agreement and the performance of its obligations hereunder have been, as applicable, duly and validly authorized by all corporate or similar action on its part and (iii) this Agreement has been duly and validly executed and delivered by each of the Parties and constitutes, a legal, valid and binding obligation of each such Party enforceable in accordance with its terms subject to the Enforceability Exceptions.
- (b) Each of Sponsor and each Supporting Sponsor Stockholder hereby represents and warrants as of the date hereof to GG, Parent and ListCo (solely with respect to itself, himself or herself and not with respect to any other Party):

- (i) The execution and delivery of this Agreement by such Person does not, and the performance by such Person of its, his or her obligations hereunder shall not, (A) if such Person is not an individual, result in any breach of any provision of the Governing Documents of such Person or (B) require any consent, approval or authorization of, or designation, declaration or filing with, any Governmental Authority that has not been given, except for (1) the filing with the SEC of such reports under Section 13(a) or 15(d) of the Exchange Act as may be required in connection with this Agreement or the transactions contemplated hereby, (2) such filings with and approvals of the Stock Exchange to permit ListCo Class A ADSs and ListCo Class C-2 ADSs or ListCo AD Warrants, as applicable, to be issued in accordance with the Business Combination Agreement to be listed on the Stock Exchange, (3) filing of the Certificate of Merger under the applicable law of the State of Delaware, (4) the GG Shareholder Approval or (5) any consents, approvals, authorizations, designations, declarations, waivers or filings, the absence of which would not reasonably expected to be, individually or in the aggregate, material to such Person, as applicable, in each case, to the extent such consent, approval or authorization of, or designation, declaration or filing with, any Governmental Authority would prevent, enjoin or materially delay the performance by such Person of its, his or her obligations under this Agreement.
- (ii) Such Person is the record and beneficial owner (within the meaning of Rule 13d-3 under the Exchange Act) of, and has good title to, all of the GG Class A Shares, GG Class F Shares and/or the GG Warrants as set forth in its respective beneficial ownership reports filed with the SEC, and there exist no Liens or any other limitation or restriction (other than transfer restrictions under the Securities Act, GG's Governing Documents, Permitted Liens, this Agreement, the Business Combination Agreement, the Letter Agreement or any other applicable securities Laws), in each case, that could reasonably be expected to (A) impair the ability of such Person to perform its obligations under this Agreement or (B) prevent, impede or delay the consummation of any of the transactions contemplated by this Agreement. The Equity Securities set forth in such beneficial ownership reports filed with the SEC are the only Equity Securities in GG owned of record or beneficially by such Person on the date of this Agreement, and none of such Equity Securities are subject to any proxy, voting trust or other agreement or arrangement with respect to the voting of such Equity Securities, except as provided hereunder and under the Letter Agreement.
- (iii) As of the date hereof, there are no Proceedings pending against such Person, or to the knowledge of such Person threatened against it, before (or, in the case of threatened Proceedings, that would be before) any Governmental Authority, which in any manner challenges or seeks to prevent, enjoin or materially delay the performance by such Person of its, his or her obligations under this Agreement or the Letter Agreement.
- (iv) No broker, finder, financial advisor, investment banker or other Person is entitled to any brokerage fee, finders' fee or other commission in connection with the transactions contemplated by the Business Combination

Agreement, other than as disclosed in the GG Disclosure Schedules, based upon arrangements made by such Person and for which GG or any of its Affiliates may become liable.

(v) Such Person understands and acknowledges that each of GG, Parent and ListCo is entering into the Business Combination Agreement in reliance upon such Person's execution and delivery of this Agreement.

Section 12. <u>Rights of Third Parties</u>. This Agreement shall be binding upon and inure solely to the benefit of each Party and its successors and permitted assigns and, except as set forth in <u>Section 9</u>, nothing in this Agreement, express or implied, is intended to or shall confer upon any other Person any rights, benefits or remedies of any nature whatsoever under or by reason of this Agreement.

Section 13. <u>Further Assurances</u>. Each of the Parties hereto is entitled to rely upon this Agreement and is irrevocably authorized to produce this Agreement or a copy hereof to any interested party in any administrative or legal proceeding or official inquiry with respect to the matters covered hereby. Each of the Parties hereto shall pay all of their respective expenses in connection with this Agreement and the transactions contemplated herein. Each of the Parties hereto shall take, or cause to be taken, all actions and do, or cause to be done, all things necessary, proper or advisable to consummate the transactions contemplated by this Agreement on the terms and conditions described therein no later than immediately prior to the consummation of the Transactions.

Section 14. <u>Notices</u>. All notices, requests, claims, demands and other communications hereunder shall be in writing and shall be given (and shall be deemed to have been duly given) by delivery in person, by email (having obtained electronic delivery confirmation thereof (i.e., an electronic record of the sender that the email was sent to the intended recipient thereof without an "error" or similar message that such email was not received by such intended recipient)), or by registered or certified mail (postage prepaid, return receipt requested) (upon receipt thereof) to the other Parties as follows:

### If to Sponsor:

Gores Guggenheim Sponsor LLC 6260 Lookout Road Boulder, CO 80301 Attention: Andrew McBride E-mail: amcbride@gores.com

with a required copy (which copy shall not constitute notice) to:

Weil, Gotshal & Manages LLP 201 Redwood Shores Parkway Redwood Shores, CA 94065

Attention: Kyle C. Krpata; James Griffen

Email: kyle.krpata@weil.com; james.griffen@weil.com

#### If to GG:

Gores Guggenheim, Inc. 6260 Lookout Road Boulder, CO 80301

Attention: Andrew McBride E-mail: amcbride@gores.com

with a required copy (which copy shall not constitute notice) to:

Weil, Gotshal & Manages LLP 201 Redwood Shores Parkway Redwood Shores, CA 94065

Attention: Kyle C. Krpata; James Griffen

Email: kyle.krpata@weil.com; james.griffen@weil.com

#### If to Parent and ListCo:

Polestar Automotive Holding Limited 13/F, Gloucester Tower, The Landmark 15 Queen's Road Central, Central Hong Kong

Attn: Yaru Li

Email: yaru.li@polestar.com

with a required copy (which shall not constitute notice) to:

Polestar Automotive Holding UK Limited 13/F, Gloucester Tower, The Landmark 15 Queen's Road Central, Central Hong Kong

Attn: Mikael Alkmark

Email: Mikael.alkmark@polestar.com

with a required copy (which shall not constitute notice) to:

Kirkland & Ellis LLP 601 Lexington Avenue New York, NY 10022

Attention: Eric L. Schiele, P.C., David B. Feirstein, P.C., Marshall P. Shaffer, P.C.

Email: eric.schiele@kirkland.com; david.feirstein@kirkland.com;

marshall.shaffer@kirkland.com

If to any Supporting Sponsor Stockholder, to such Stockholder's address as set forth opposite its name in the signature page hereto.

Section 15. <u>Waiver</u>. No (a) extension of time for the performance of any of the obligations or other acts of a Party set forth herein or (b) provision of this Agreement may be granted or waived, as applicable, unless such extension or waiver, as applicable, is in writing and signed by or on behalf of the Party or Parties granting such waiver. Any waiver of any term or condition shall not be construed as a waiver of any subsequent breach or a subsequent waiver of the same term or condition, or a waiver of any other term or condition of this Agreement. The failure of any Party to assert any of its rights hereunder shall not constitute a waiver of such rights.

Section 16. <u>Incorporation by Reference</u>. Sections 1.02 (Construction); 11.03 (Assignment); 11.06 (Governing Law); 11.07 (Captions; Counterparts); 11.09 (Entire Agreement); 11.10 (Amendments); 11.11 (Severability); 11.12 (Jurisdiction); 11.13 (Waiver of Jury Trial); 11.14 (Enforcement) and 11.16 (Nonsurvival of Representations, Warranties and Covenants) of the Business Combination Agreement are incorporated herein and shall apply to this Agreement *mutatis mutandis*.

[Signature pages follow]

**IN WITNESS WHEREOF**, each of the Parties has caused this Agreement to be duly executed on its behalf as of the day and year first above written.

## GORES GUGGENHEIM SPONSOR LLC

By: /s/ Alec Gores

Name: Alec Gores Title: Chairman

### GORES GUGGENHEIM, INC.

By: /s/ Mark Stone

Name: Mark Stone

Title: Chief Executive Officer

### POLESTAR AUTOMOTIVE HOLDING LIMITED

By: /s/ Lars Danielson

Name: Lars Danielson Title: Director

### POLESTAR AUTOMOTIVE HOLDING UK LIMITED

By: /s/ Mikael Alkmark

Name: Jan Mikael Alkmark

Title: Director

# SUPPORTING SPONSOR STOCKHOLDER

| By: | /s/ Randall Bort   | Address for Notice: |
|-----|--------------------|---------------------|
|     | Name: Randall Bort |                     |
|     |                    | Attention:          |
|     |                    | Facsimile:          |
|     |                    | E-mail:             |

# SUPPORTING SPONSOR STOCKHOLDER

| /s/ Elizabeth Marcellino   | Address for Notice: |
|----------------------------|---------------------|
| Name: Elizabeth Marcellino |                     |
|                            | Attention:          |
|                            | Facsimile:          |
|                            | E-mail:             |

# SUPPORTING SPONSOR STOCKHOLDER

| By: | /s/ Nancy Tellem   | Address for Notice: |
|-----|--------------------|---------------------|
|     | Name: Nancy Tellem |                     |
|     |                    | Attention:          |
|     |                    | Facsimile:          |
|     |                    | E-mail:             |
|     |                    |                     |

# Annex III

Amendment No. 1 to the Registration Rights Agreement

#### AMENDMENT NO. 1 TO THE REGISTRATION RIGHTS AGREEMENT

This AMENDMENT NO. 1 TO THE REGISTRATION RIGHTS AGREEMENT (this "Amendment No. 1"), dated as of December 17, 2021, is made by and among (a) Polestar Automotive Holding UK Limited, a limited company organized under the laws of England and Wales (the "Company"), (b) Polestar Automotive Holding Limited, a Hong Kong incorporated company ("Parent"), (c) Zibo High-Tech Industrial Investment Co., Ltd., Zibo Financial Holding Group Co., Ltd, Chongqing Liangjiang (重庆承星股权投资基金合伙企业(有限合伙)), Northpole GLY 1 LP, GLY New Mobility 1. LP, SNITA Holding B.V. ("Snita"), PSINV AB, PSD Investment Limited (collectively, the "Parent Holders") and (d) Gores Guggenheim Sponsor LLC, a Delaware limited liability company (the "Sponsor"), Randall Bort, Elizabeth Marcellino and Nancy Tellem (collectively, the "Gores Holders," and together with the Company, Parent, Zibo High-Tech Industrial Investment Co., Ltd., Zibo Financial Holding Group Co., Ltd, Chongqing Liangjiang (重庆承星股权投资基金合伙企业(有限合伙)), Northpole GLY 1 LP, GLY New Mobility 1. LP, Snita, Parent Holders and Sponsor, the "RRA Parties"). Capitalized terms used but not otherwise defined herein shall have the meanings ascribed to such terms in the Business Combination Agreement (as defined below).

#### **RECITALS**

WHEREAS, the RRA Parties entered into that certain Registration Rights Agreement on September 27, 2021 (the "Registration Rights Agreement");

WHEREAS, concurrently with the execution of this Amendment No. 1, (a) Sponsor is assigning to certain investors its obligations to purchase ListCo Class A ADSs under, and Sponsor and certain of its Affiliates, GG and the Company are amending, that certain subscription agreement dated September 27, 2021, by and among such parties, (b) Snita is assigning to certain investors its obligations to purchase ListCo Class A ADSs under, and Snita, GG and the Company are amending, that certain subscription agreement dated September 27, 2021, by and among such parties and (c) such investors referenced in clauses (a) and (b), GG and the Company are entering into certain subscription agreements dated December 17, 2021, to reflect such assignment in each case by and among the applicable investors and parties; and

WHEREAS, concurrently with the execution and delivery hereof, the Parties are entering into that certain Amendment No. 1 to Business Combination Agreement (the "BCA Amendment No. 1") dated as of December 17, 2021, by and among Gores Guggenheim, Inc., a Delaware corporation ("GG"), Parent, Polestar Automotive (Singapore) Pte. Ltd., a private company limited by shares in Singapore ("Polestar Singapore"), Polestar Holding AB, a private limited liability company incorporated under the laws of Sweden ("Polestar Sweden"), the Company, and PAH UK Merger Sub Inc., a Delaware corporation and a direct wholly owned subsidiary of ListCo ("Merger Sub"), which amends and restates that certain Business Combination Agreement, dated as of September 27, 2021, by and among GG, Parent, Polestar Singapore, Polestar Sweden, the Company and Merger Sub (such agreement, as amended by BCA Amendment No. 1, and as may be further amended from time to time, the "Business Combination Agreement"); and

NOW THEREFORE, in consideration of the covenants and agreements contained herein, and certain other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

#### 1. The Amendments.

- a. The first recital of the Registration Rights Agreement is hereby amended and restated in its entirety as follows:
  - WHEREAS, upon the closing of the transactions contemplated by that certain Business Combination Agreement, dated as of September 27, 2021 (as may be amended from time to time, the "Business Combination Agreement"), by and among Gores Guggenheim, Inc., a Delaware corporation ("GG"), the Company, PAH UK Merger Sub Inc., a Delaware corporation ("Merger Sub"), Parent, Polestar Automotive (Singapore) Pte. Ltd., a private company limited by shares in Singapore and Polestar Holding AB, a private limited liability company incorporated under the laws of Sweden, Merger Sub shall merge with and into GG (the "Merger"), with GG surviving the Merger as a wholly owned subsidiary of the Company, on the terms and subject to the conditions in the Business Combination Agreement (the "Transactions");
- b. Section 1.1 of the Registration Rights Agreement is hereby amended to restate the definition of "Subscription Agreements" in the Registration Rights Agreement in its entirety as follows:
  - "Subscription Agreements" means those certain subscription agreements dated September 27, 2021 and December 17, 2021, as may be amended from time to time, by and between the Company and certain subscribers to purchase Shares in connection with the Transactions.
- 2. Effect of Amendment. This Amendment No. 1 shall be effective as of the date first written above. For the avoidance of any doubt, all references: (a) in the Registration Rights Agreement to "this Agreement" and (b) to the Registration Rights Agreement in any other agreements, exhibits and schedules will, in each case, be deemed to be references to the Registration Rights Agreement as amended by this Amendment No. 1. Except as amended hereby, the Registration Rights Agreement will continue in full force and effect and shall be otherwise unaffected hereby. This Amendment No. 1 shall be binding upon and shall inure to the benefit of the Parties and their respective successors and assigns. This Amendment No. 1 only may be amended by the Parties by execution of an instrument in writing signed on behalf of each of the Parties.
- 3. <u>Incorporation by Reference</u>. Sections 1.02 (Construction); 11.03 (Assignment); 11.06 (Governing Law); 11.07 (Captions; Counterparts); 11.09 (Entire Agreement); 11.10 (Amendments); 11.11 (Severability); 11.12 (Jurisdiction); 11.13 (Waiver of Jury Trial) and 11.14 (Enforcement) of the Business Combination Agreement are incorporated herein and shall apply to this Amendment No. 1 *mutatis mutandis*.

[Signature pages follow]

IN WITNESS WHEREOF, the undersigned have caused this Amendment No. 1 to be executed as of the date first written above.

## **COMPANY:**

# POLESTAR AUTOMOTIVE HOLDING UK LIMITED

By: /s/ Mikael Almark

Name: Jan Mikael Alkmark

Title: Director

[Signature Page to Registration Rights Agreement]

IN WITNESS WHEREOF, the undersigned have caused this Amendment No. 1 to be executed as of the date first written above.

## **GORES HOLDERS:**

## GORES GUGGENHEIM SPONSOR LLC

By: **GG Sponsor, LLC**Its: Managing Member

By: AEG Holdings, LLC

Its: Manager

By: /s/ Alex Gores

Name: Alec Gores Title: Chairman

Address:

Email:

By: /s/ Andrew Rosenfield

Name: Andrew Rosenfield

Title: Manager

Address:

Email:

By: /s/ Randall Bort

Name: Randall Bort

Address:

Email:

By: /s/ Elizabeth Marcellino

Name: Elizabeth Marcellino

Address:

Email:

By: /s/ Nancy Tellem

Name: Nancy Tellem

Address:

Email:

[Signature Page to Registration Rights Agreement]

IN WITNESS WHEREOF, the undersigned have caused this Amendment No. 1 to be executed as of the date first written above.

### PARENT HOLDERS:

ZIBO HIGH-TECH INDUSTRIAL INVESTMENT CO., LTD.

By: /s/ Ning Chen
Name: Ning Chen

Address:

Email:

ZIBO FINANCIAL HOLDING GROUP CO., LTD

By: /s/ Dapeng Li

Name: Dapeng Li

Address:

Email:

CHONGQING LIANGJIANG (重庆承星股权投资基金合伙企业 (有限合伙))

By: /s/ Yimin Ai

Name: Yimin Ai

Address:

Email:

NORTHPOLE GLY 1 LP

By: /s/ Hrvoje Krkalo

Name: Hrvoje Krkalo

Address:

Email:

 $[Signature\ Page\ to\ Amendment\ No.\ 1\ to\ Registration\ Rights\ Agreement]$ 

## **GLY NEW MOBILITY 1. LP**

By: /s/ Hrvoje Krkalo

Name: Hrvoje Krkalo

Address:

Email:

## SNITA HOLDING B.V.

By: /s/ Lex Kerssemakers

Name: Lex Kerssemakers

Address:

Email:

By: /s/ Per Ansgar

Name: Per Ansgar

Address:

Email:

# PSINV AB, PSD INVESTMENT LIMITED

By: /s/ Fredrik Aaben

Name: Fredrik Aaben

Address:

Email:

By: /s/ Per Ansgar

Name: Per Ansgar

Address:

Email:

#### PSD INVESTMENT LIMITED

By: /s/ Shufu Li

Name: Shufu Li

Address:

Email:

[Signature Page to Amendment No. 1 to Registration Rights Agreement]

IN WITNESS WHEREOF, the undersigned have caused this Amendment No. 1 to be executed as of the date first written above.

## PARENT:

# POLESTAR AUTOMOTIVE HOLDING LIMITED

By: /s/ Lars Danielson

Name: Lars Danielson Title: Director

[Signature Page to Amendment No. 1 to Registration Rights Agreement]

# ANNEX IV

Registration Rights Agreement, dated September 27, 2021

#### REGISTRATION RIGHTS AGREEMENT

This REGISTRATION RIGHTS AGREEMENT (this "Agreement") is made and entered into as of September 27, 2021, by and among (a) Polestar Automotive Holding UK Limited, a limited company organized under the laws of England and Wales (the "Company"), (b) Polestar Automotive Holding Limited, a Hong Kong incorporated company ("Parent"), (c) Zibo High-Tech Industrial Investment Co., Ltd., Zibo Financial Holding Group Co., Ltd, Chongqing Liangjiang (重庆承星股权投资基金合伙企业(有限合伙)), Northpole GLY 1 LP, GLY New Mobility 1. LP, SNITA Holding B.V. ("Snita"), PSINV AB, PSD Investment Limited (collectively, the "Parent Holders") and (d) Gores Guggenheim Sponsor LLC, a Delaware limited liability company (the "Sponsor"), Randall Bort, Elizabeth Marcellino and Nancy Tellem (collectively, the "Gores Holders"). Parent, the Parent Holders, the Gores Holders and any person or entity who hereafter becomes a party to this Agreement pursuant to Section 5.2 of this Agreement are each referred to herein as a "Holder" and collectively as the "Holders".

WHEREAS, upon the closing of the transactions contemplated by that certain Business Combination Agreement, dated as of September 27, 2021 (the "<u>Business Combination Agreement</u>"), by and among Gores Guggenheim, Inc., a Delaware corporation ("<u>GG</u>"), the Company, PAH UK Merger Sub Inc., a Delaware corporation ("<u>Merger Sub</u>"), Parent, Polestar Automotive (Singapore) Pte. Ltd., a private company limited by shares in Singapore and Polestar Holding AB, a private limited liability company incorporated under the laws of Sweden, Merger Sub shall merge with and into GG (the "<u>Merger</u>"), with GG surviving the Merger as a wholly owned subsidiary of the Company, on the terms and subject to the conditions in the Business Combination Agreement (the "Transactions");

WHEREAS, in connection with the Merger, (a) each GG Unit (as defined in the Business Combination Agreement) shall be automatically separated, and the holder thereof shall be deemed to hold one GG Class A Share (as defined in the Business Combination Agreement) and one-fifth (1/5) of a Public Warrant (as defined in the Business Combination Agreement) in accordance with the terms of the applicable GG Unit, (b) each GG Share (as defined in the Business Combination Agreement) shall be automatically cancelled and extinguished and converted into the right to receive one Class A American depository share of the Company (the "Class A ADS") duly and validly issued against the deposit of an underlying Class A ordinary share of the Company (such Class A ordinary shares, collectively, the "Shares"), (c) at the Effective Time (as defined in the Business Combination Agreement), (i) each Public Warrant shall be automatically cancelled and extinguished and converted into the right to receive either one ListCo AD Warrant (as defined in the Business Combination Agreement) is obtained prior to the Effective Time, one Class C-1 American depository share of the Company (the "Class C-1 ADS") duly and validly issued against the deposit of an underlying Class C-1 ordinary share of the Company (such Class C-1 ordinary shares, collectively, the "Class C-1 Shares") and (ii) each Private Placement Warrant (as defined in the Business Combination Agreement) shall be automatically cancelled and extinguished and converted into the right to receive either one ListCo AD Warrant or, in the event the Requisite GG Warrantholder Approval is obtained prior to the Effective Time, one Class C-2 American depository share of the Company (the "Class C-2 Shares") duly and validly issued against the deposit of an underlying Class C-2 Ordinary share of the Company (such Class C-2 Shares"), in each case, on the terms and subject to the conditions set forth in the Business Combination Agreement;

WHEREAS, upon the closing of the Transactions, (a) all Shares issued to holders of GG Shares shall be automatically deposited by the Company with the Depository Bank (as defined in the Business Combination Agreement) in accordance with the Deposit Agreement (as defined in the Business Combination Agreement) and the Depository Bank shall issue Class A ADSs to such holders, (b) in the event that the Requisite GG Warrantholder Approval is obtained prior to the Effective Time, all Class C-1

Shares issued to holders of Public Warrants or, in the event that the Requisite GG Warrantholder Approval is not obtained prior to the Effective Time, all Listco Warrants (as defined in the Business Combination Agreement) issued to the holders of Public Warrants, shall automatically be deposited by the Company with the Depository Bank in accordance with the Deposit Agreement and the Depository Bank shall issue Class C-1 ADSs or Listco AD Warrants, as applicable, to such holders, and (c) in the event that the Requisite GG Warrantholder Approval is obtained prior to the Effective Time, all Class C-2 Shares issued to holders of Private Placement Warrants or, in the event that the Requisite GG Warrantholder Approval is not obtained prior to the Effective Time, all Listco Warrants issued to the holders of Private Placement Warrants, shall be automatically deposited by the Company with the Depository Bank in accordance with the Deposit Agreement and the Depository Bank shall issue Class C-2 ADSs or Listco AD Warrants, as applicable, to such holders;

WHEREAS, as of the closing of the Transactions, (a) Parent will hold Class A ADSs (the "Parent ADSs") and Class B ordinary shares of the Company (the "Class B ordinary shares" and together with the Parent ADSs, the "Parent Shares"), (b) Parent Holders will have an interest in the Parent Shares, and (c) the Gores Holders will hold Class A ADSs (the "Founder Shares") and Class C-2 ADSs or, if the requisite GG Warrantholder Approval is not obtained prior to the Effective Time, ListCo AD Warrants (the "GG C Shares");

WHEREAS, Parent will use its commercially reasonable efforts to dissolve or liquidate in the 2022 calendar year and distribute all of the Parent Shares to the Parent Holders:

WHEREAS, in the event the Requisite GG Warrantholder Approval is obtained, the Sponsor will, in accordance with the ListCo New Articles of Association (as defined in the Business Combination Agreement) that will be adopted in connection with the closing of the Transactions, have the option to convert its Class C-2 ADSs into Class C-1 ADSs (such Class C-1 ADSs, collectively, the "Converted Class C-1 ADSs");

WHEREAS, pursuant to the Business Combination Agreement, upon the occurrence of certain Earn Out Triggering Events (as defined in the Business Combination Agreement) the Company will issue a certain number of Earn Out Class A Shares (as defined in the Business Combination Agreement) and Earn Out Class B Shares (as defined in the Business Combination Agreement) from time to time after the closing of the Transactions to Parent or, in the event Parent has been dissolved or liquidated, to the Parent Holders in accordance with the terms set forth in the Business Combination Agreement;

WHEREAS, Snita has subscribed for 49,803,900 convertible preference shares of ListCo (the "<u>VCC Preference Shares</u>") pursuant to a subscription agreement dated September 27, 2021 by and between the Company and Snita (the "<u>VCC Preference Shares Subscription Agreement</u>");

WHEREAS, the VCC Preference Shares will be issued at closing of the Transactions and convertible into Shares in accordance with the terms of the VCC Preference Shares Subscription Agreement and the ListCo New Articles of Association that will be adopted at closing of the Transactions;

WHEREAS, Parent, each of the Parent Holders and the Sponsor have entered into lock-up agreements (as amended from time to time in accordance with the terms thereof, the "Lock-Up Agreements"), pursuant to which Parent, each of the Parent Holders and the Sponsor agreed not to transfer certain securities of the Company as set forth in the applicable Lock-Up Agreements during the period commencing the date of the closing of the Transaction and ending 180 days following the date of the closing of the Transaction; and

WHEREAS, on March 25, 2021, GG and the Gores Holders entered into that certain Registration Rights Agreement, which, pursuant to the terms of the Business Combination Agreement, was terminated effective as of the consummation of the Transactions.

NOW, THEREFORE, in consideration of the foregoing, and the mutual promises herein made, and in consideration of the representations, warranties and covenants herein contained, the receipt and sufficiency of which the parties hereto hereby acknowledge, the parties hereto hereby agree as follows:

# ARTICLE I DEFINITIONS

Section 1.1 <u>Definitions</u>. Capitalized terms used in this Agreement have the meanings set forth below.

"Adverse Disclosure" means any public disclosure of material non-public information (including information with respect to a potential financing, acquisition, disposition, merger, reorganization or similar transaction), which disclosure, in the good faith judgment of the Board or Chief Executive Officer or Chief Financial Officer of the Company, after consultation with counsel to the Company, (a) would be required to be made in any Registration Statement or Prospectus in order for the applicable Registration Statement or Prospectus not to contain any Misstatement, (b) would not be required to be made at such time if the Registration Statement were not being filed, declared effective or used, as the case may be, and (c) the Company has a bona fide business purpose for not making such information public.

- "Agreement" has the meaning given in the Recitals.
- "Blackout Period" has the meaning given in Section 2.7(d).
- "Block Trade" means an offering or sale of Registrable Securities by any Holder on an underwritten basis (whether firm commitment or otherwise) without substantial marketing efforts prior to pricing and is commonly known as a "block trade", including a same day trade, overnight trade or similar transaction.
  - "Board" means the Board of Directors of the Company.
  - "Business Combination Agreement" has the meaning given in the Recitals.
- "Business Day" means a day other than (a) a Saturday or Sunday or (b) any other day on which banks located in (a) New York, USA, (b) Hong Kong, China, (c) Singapore, (d) Stockholm, Sweden or (e) London, England are required or authorized by law to be closed for business.
  - "Claims" has the meaning given in Section 4.1(a).
  - "Class A ADS" has the meaning given in the Recitals.
  - "Class B ordinary shares" has the meaning given in the Recitals.
  - "Class C-1 ADS" has the meaning given in the Recitals.
  - "Class C-2 ADS" has the meaning given in the Recitals.
  - "Class C Shares" has the meaning given in the Recitals.

- "Class C-1 Shares" has the meaning given in the Recitals.
- "Class C-2 Shares" has the meaning given in the Recitals.
- "Commission" means the United States Securities and Exchange Commission.
- "Commission Guidance" means (a) any publicly-available written or oral guidance of the Commission staff, or any comments, requirements or requests of the Commission staff and (b) the Securities Act.
  - "Company" has the meaning given in the Recitals.
  - "Company Shelf Take Down Notice" has the meaning given in Section 2.1(c).
  - "Converted Class C-1 ADSs" has the meaning given in the Recitals.
  - "Demand Registration" has the meaning given in Section 2.2(a).
- "<u>Demanding Holder</u>" means, as applicable, (a) the applicable Holders making a written demand for the Registration of Registrable Securities pursuant to <u>Section 2.2(a)</u> or (b) the applicable Holders making a written demand for a Shelf Underwritten Offering of Registrable Securities pursuant to <u>Section 2.1(c)</u>.
  - "Deposit Agreement" has the meaning given in the Recitals.
  - "Depository Bank" has the meaning given in the Recitals.
  - "Earn Out Class A Shares" has the meaning given in the Business Combination Agreement.
  - "Earn Out Class B Shares" has the meaning given in the Business Combination Agreement.
  - "Earn Out Triggering Events" has the meaning given in the Business Combination Agreement.
  - "Effective Time" has the meaning given in the Business Combination Agreement.
  - "Effectiveness Deadline" has the meaning given in Section 2.1(a).
  - "Exchange Act" means the United States Securities Exchange Act of 1934.
- "Exempted Registration Statement" means a Registration Statement (a) filed in connection with any employee stock option or other benefit plan, including a Registration Statement on Form S-8 (or similar successor form), (b) for an exchange offer or offering of securities solely to the Company's existing shareholders, (c) filed on a Registration Statement on Form S-4 or F-4 (or similar successor forms), (d) for an offering of debt that is convertible into equity securities of the Company, (e) for a dividend reinvestment plan, or (f) filed pursuant to Section 2.1(a).
  - "Form F-1 Shelf" has the meaning given in Section 2.1(a).
  - "Form F-3 Shelf" has the meaning given in Section 2.1(b).

- "Founder Shares" has the meaning given in the Recitals.
- "GG" has the meaning given in the Recitals.
- "GG C Shares" has the meaning given in the Recitals.
- "GG Unit" has the meaning given in the Recitals.
- "GG Share" has the meaning given in the Recitals.
- "GG Warrants" has the meaning given in the Business Combination Agreement.
- "Gores Holders" has the meaning given in the Recitals.
- "Holders" has the meaning given in the Recitals.
- "ListCo AD Warrant" has the meaning given in the Business Combination Agreement.
- "ListCo New Articles of Association" has the meaning given in the Business Combination Agreement.
- "ListCo Warrant" has the meaning given in the Business Combination Agreement.
- "Lock-Up Agreements" has the meaning given in the Recitals.
- "Maximum Number of Securities" has the meaning given in Section 2.2(d).
- "Merger Sub" has the meaning given in the Recitals.
- "Minimum Amount" has the meaning given in Section 2.1(c).
- "<u>Misstatement</u>" means an untrue statement of a material fact or an omission to state a material fact required to be stated therein, or necessary to make the statements therein (in the case of any Prospectus and any preliminary Prospectus, in the light of the circumstances under which they were made) not misleading.
  - "Parent" has the meaning given in the Recitals.
  - "Parent ADSs" has the meaning given in the Recitals.
  - "Parent Holders" has the meaning given in the Recitals.
  - "Parent Shares" has the meaning given in the Recitals.
- "Permitted Transferee" means a person or entity to whom a Holder is permitted to Transfer such Registrable Securities pursuant to the Lock-Up Agreements.
  - " $\underline{Piggyback\ Registration}$ " has the meaning given in Section 2.4(a).
  - "Private Placement Warrant" has the meaning given in the Recitals.

"Prospectus" means the prospectus included in any Registration Statement, as supplemented by any and all prospectus supplements and as amended by any and all post-effective amendments and including all material incorporated by reference in such prospectus.

"Public Warrant" has the meaning given in the Recitals.

## "Registrable Security" means:

- (a) (i) any Shares, Class C-2 Shares or ListCo Warrants issued at the closing of the Transactions pursuant to the Business Combination Agreement (including, without limitation, any Shares issued pursuant to a Subscription Agreement) to Holders and that have, in each case, been deposited with the Depository Bank in accordance with the Deposit Agreement,
- (ii) any Shares to be represented by Class A ADSs issued in accordance with the Deposit Agreement and that are hereafter acquired by a Holder upon exercise of any rights associated with their Converted Class C-1 ADSs, Class C-2 ADSs or ListCo AD Warrants (as the case may be),
- (b) any Shares issuable upon the conversion of the VCC Preference Shares, the Class B ordinary shares or Earn Out Class B Shares and to be represented by Class A ADSs issued in accordance with the Deposit Agreement,
- (c) any Earn Out Class A Shares issued pursuant to the Business Combination Agreement and to be represented by Class A ADSs upon deposit of such Earn Out Class A Shares with the Depository Bank in accordance with the Deposit Agreement,
- (d) any Shares, Class C-1 Shares, Class C-2 Shares, or ListCo Warrants (as the case may be) issued or issuable with respect to the securities referred to in the foregoing clauses (a), (b) and (c) by way of a stock dividend or stock split or in connection with a combination of shares, recapitalization, merger, consolidation or other reorganization or otherwise to be represented by Class A ADSs, Converted Class C-1 ADSs, Class C-2 ADSs or ListCo AD Warrants (as the case may be) issued in accordance with the Deposit Agreement, and
- (e) any Class A ADSs, Converted Class C-1 ADSs, Class C-2 ADSs or ListCo AD Warrants (as the case may be), representing Shares, Class C-1 Shares, Class C-2 Shares or ListCo Warrants, VCC Preference Shares, Class B ordinary shares, Earn Out Class B Shares and Earn Out Class A Shares described in the foregoing clauses (a), (b), (c) and (d) (as the case may be);

<u>provided</u>, <u>however</u>, that, as to any particular Registrable Security, such securities shall cease to be Registrable Securities when: (i) a Registration Statement with respect to the sale of such securities shall have become effective under the Securities Act and such securities shall have been sold, transferred, disposed of or exchanged in accordance with such Registration Statement; (ii) the Shares, Class C-1 Shares, Class C-2 Shares, ListCo Warrants, VCC Preference Shares, Class B ordinary shares, Earn Out Class B Shares, Earn

Out Class A Shares, Class A ADSs, Converted Class C-1 ADSs, Class C-2 ADSs or ListCo AD Warrants described in foregoing clauses (a), (b), (c), (d) and (e) shall have been otherwise transferred, new certificates for such securities not bearing (or book entry positions not subject to) a legend restricting further transfer shall have been delivered by the Company and subsequent public distribution of such securities shall not require registration under the Securities Act; (iii) such securities shall have ceased to be outstanding; or (iv) such securities have been sold to, or through, a broker, dealer or underwriter in a public distribution or other public securities transaction.

"Registration" means a registration effected by preparing and filing a registration statement or similar document in compliance with the requirements of the Securities Act, and the applicable rules and regulations promulgated thereunder, and such registration statement becoming effective.

"Registration Expenses" means the out-of-pocket expenses of a Registration, including the following:

- (a) all registration and filing fees (including fees with respect to filings required to be made with the United States Financial Industry Regulatory Authority, Inc.) and any securities exchange on which the Registrable Securities are then listed;
- (b) fees and expenses of compliance with securities or blue sky laws (including reasonable fees and disbursements of counsel for the Underwriters in connection with "blue sky" qualifications of Registrable Securities);
  - (c) printing, messenger, telephone, delivery and reasonable road show or other reasonable marketing expenses;
  - (d) reasonable fees and disbursements of counsel for the Company;
- (e) reasonable fees and disbursements of all independent registered public accountants of the Company incurred specifically in connection with such Registration;
- (f) reasonable fees and expenses of one (1) legal counsel selected by either (i) the majority-in-interest of the Demanding Holders (and any local or foreign counsel) initiating a Demand Registration or Shelf Underwritten Offering (including a Block Trade), or (ii) of a majority-in-interest of participating Holders under Section 2.4 if the Registration was initiated by the Company for its own account or that of a Company stockholder other than pursuant to rights under this Agreement, in each case to be registered for offer and sale in the applicable Registration; and
  - (g) American Depository issuance fees.

"Registration Statement" means any registration statement that covers the Registrable Securities pursuant to the provisions of this Agreement, including the Prospectus included in such registration statement, amendments (including post-effective amendments) and supplements to such registration statement, and all exhibits to and all material incorporated by reference in such registration statement. The term Registration Statement shall include any registration statement on Form F-6, or any successor form, necessary to register Class A ADSs, Converted Class C-1 ADSs, Class C-2 ADSs and ListCo AD Warrants (as the case may be) representing Shares, Class C-1 Shares, Class C-2 Shares or ListCo Warrants issued in accordance with the Deposit Agreement).

- "Removed Securities" has the meaning given in Section 2.6.
- "Requesting Holder" has the meaning given in Section 2.2(a).
- "Requisite GG Warrantholder Approval" has the meaning given in the Business Combination Agreement.
- "Securities Act" means the United States Securities Act of 1933.
- "Shares" has the meaning given in the Recitals.
- "Shelf Take Down Notice" has the meaning given in Section 2.1(c).
- "Shelf Underwritten Offering" has the meaning given in Section 2.1(c).
- "Snita" has the meaning given in the Recitals.
- "Sponsor" has the meaning given in the Recitals.
- "Subscription Agreements" means those certain subscription agreements dated September 27, 2021 by and between the Company and certain subscribers to purchase Shares in connection with the Transactions.
  - "Transactions" has the meaning given in the Recitals.
- "Transfer" means to, directly or indirectly, sell, transfer, assign, pledge, encumber, hypothecate or similarly dispose of, either voluntarily or involuntarily, or to enter into any contract, option or other arrangement or understanding with respect to the sale, transfer, assignment, pledge, encumbrance, hypothecation or similar disposition of, any interest owned by a person or any interest (including a beneficial interest) in, or the ownership, control or possession of, any interest owned by a person.
- "<u>Underwriter</u>" means a securities dealer who purchases any Registrable Securities as principal in an Underwritten Offering and not as part of such dealer's market-making activities.
- "<u>Underwritten Offering</u>" means a Registration in which securities of the Company are sold to an Underwriter in a firm commitment underwriting for distribution to the public.
  - "VCC Preference Share" has the meaning given in the Recitals.
  - "VCC Preference Shares Subscription Agreement" has the meaning given in the Recitals.

#### ARTICLE II REGISTRATIONS

# Section 2.1 Shelf Registration.

(a) The Company shall, as soon as practicable, but in any event no later than the date that is thirty (30) calendar days after the closing of the Transactions, file a Registration Statement under the Securities Act to permit the public resale by the Holders of all the Registrable Securities held by the Holders from time to time as permitted by Rule 415 under the Securities Act (or any successor or similar provision adopted by the Commission then in effect) on the terms and conditions specified in this Section

2.1(a) and shall use its reasonable best efforts to cause such Registration Statement to be declared effective as soon as practicable after the filing thereof, but in no event later than sixty (60) calendar days following the filing deadline (the "Effectiveness Deadline"); provided that the Effectiveness Deadline shall be extended to ninety (90) calendar days after the filing deadline if the Registration Statement is reviewed by, and receives comments from, the Commission. The Registration Statement filed with the Commission pursuant to this Section 2.1(a) shall be on a shelf Registration Statement on Form F-1 (a "Form F-1 Shelf") or such other form of Registration Statement as is then available to effect a registration for resale of such Registrable Securities, covering such Registrable Securities, and shall contain a Prospectus in such form as to permit any Holder to sell such Registrable Securities pursuant to Rule 415 under the Securities Act (or any successor or similar provision adopted by the Commission then in effect) at any time beginning on the effective date for such Registration Statement. A Registration Statement filed pursuant to this Section 2.1(a) shall provide for the resale pursuant to any method or combination of methods legally available to, and requested by, the Holders. The Company shall use its reasonable best efforts to cause a Registration Statement filed pursuant to this Section 2.1(a) to remain effective, and to be supplemented and amended to the extent necessary to ensure that such Registration Statement is available or, if not available, that another Registration Statement is available, for the resale of all the Registrable Securities held by the Holders until all such Registrable Securities have ceased to be Registrable Securities. When effective, a Registration Statement filed pursuant to this Section 2.1(a) (including the documents incorporated therein by reference) will comply as to form in all material respects with all applicable requirements of the Securities Act and the Exchange Act and will not contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading (in the case of any Prospectus contained in such Registration Statement, in the light of the circumstances under which such statement is made).

(b) The Company shall use its reasonable best efforts to convert the Form F-1 Shelf filed pursuant to Section 2.1(a) to a shelf Registration Statement on Form F-3 (a "Form F-3 Shelf") as promptly as practicable after the Company is eligible to use a Form F-3 Shelf and have the Form F-3 Shelf declared effective as promptly as practicable and to cause such Form F-3 Shelf to remain effective, and to be supplemented and amended to the extent necessary to ensure that such Registration Statement is available or, if not available, that another Registration Statement is available, for the resale of all the Registrable Securities held by the Holders until all such Registrable Securities have ceased to be Registrable Securities.

(c) Subject to the limitations set forth in Section 2.7, at any time and from time to time following the effectiveness of the shelf Registration Statement required by Section 2.1(a) or Section 2.1(b), each of the Sponsor, Parent or a majority in interest of the then-outstanding number of Registrable Securities held by the Parent Holders (the "Parent Demanding Holders") may request to sell all or a portion of their Registrable Securities in an underwritten offering that is registered pursuant to such shelf Registration Statement, including a Block Trade (a "Shelf Underwritten Offering") provided that the Sponsor, Parent or the Parent Demanding Holders, as the case may be, (i) reasonably expects to sell Registrable Securities yielding aggregate gross proceeds in excess of \$50,000,000 from such Shelf Underwritten Offering or (ii) reasonably expects to sell all of the Registrable Securities held by such Holder in such Shelf Underwritten Offering (the amount of Registrable Securities pursuant to the foregoing clause (i) or (ii), as applicable, the "Minimum Amount"). All requests for a Shelf Underwritten Offering shall be made by giving written notice to the Company (the "Shelf Take Down Notice"). Each Shelf Takedown Notice shall specify the approximate number of Registrable Securities proposed to be sold in the Shelf Underwritten Offering and the expected price range (net of underwriting discounts and commissions) of such Shelf Underwritten Offering. Within five (5) Business Days after receipt of any Shelf Take Down Notice, the Company shall give written notice of such requested Shelf Underwritten Offering to all other Holders of Registrable Securities (the "Company Shelf Takedown Notice") and, subject to the provisions of Section 2.2(d) shall include in such Shelf Underwritten Offering all

Registrable Securities with respect to which the Company has received written requests for inclusion therein, within five (5) Business Days after sending the Company Shelf Takedown Notice, or, in the case of a Block Trade, as provided in Section 2.5. The Company shall enter into an underwriting agreement in a form as is customary in Underwritten Offerings of securities by the Company with the managing Underwriter or Underwriters selected by the Holders and reasonably acceptable to the Company and shall take all such other reasonable actions as are requested by the managing Underwriter or Underwriters in order to expedite or facilitate the disposition of such Registrable Securities. In connection with any Shelf Underwritten Offering contemplated by this Section 2.1(c), subject to Section 3.3 and Article IV, the underwriting agreement into which each Holder and the Company shall enter shall contain such representations, covenants, indemnities and other rights and obligations as are customary in underwritten offerings of securities by the Company. Any Shelf Underwritten Offering effected pursuant to this Section 2.1(c) shall be counted as a Registration for purposes of the limit on the number of Registrations that can be effected under Section 2.2.

#### Section 2.2 Demand Registration.

(a) Subject to the provisions of Section 2.7, at any time and from time to time on or after the the closing of the Transactions, each of (i) the Gores Holders of at least a majority in interest of the then-outstanding number of Registrable Securities held by the Gores Holders (the "Gores Demanding Holders, (ii) the Parent Demanding Holders, and (iii) Parent (together with the Gores Demanding Holders, the Parent Demanding Holders, the "Demanding Holders"), may make a written demand for Registration of all or part of their Registrable Securities on (1) Form F-1 or (2) if available, Form F-3, which in the case of either clause (1) or (2), may be a shelf Registration Statement filed pursuant to Rule 415 under the Securities Act, which written demand shall describe the amount and type of securities to be included in such Registration and the intended method(s) of distribution thereof (such written demand a "Demand Registration"). The Company shall, promptly following the Company's receipt of a Demand Registration (and, in any event, within twenty days of its receipt of such Demand Registration, notify, in writing all other Holders of Registrable Securities of such demand, and each Holder of Registrable Securities who thereafter wishes to include all or a portion of such Holder's Registrable Securities in a Registration pursuant to a Demand Registration (each such Holder that includes all or a portion of such Holder's Registrable Securities in such Registration, a "Requesting Holder") shall so notify the Company, in writing, within ten (10) days after the receipt by the Holder of the notice from the Company. For the avoidance of doubt, to the extent a Requesting Holder also separately possesses Demand Registration rights pursuant to this Section 2.2, but is not the Holder who exercises such Demand Registration rights, the exercise by such Requesting Holder of its rights pursuant to the foregoing sentence shall not count as the exercise by it of one of its Demand Registration rights. Upon receipt by the Company of any such written notification from a Requesting Holder(s) to the Company, subject to Section 2.2(d), such Requesting Holder(s) shall be entitled to have their Registrable Securities included in a Registration pursuant to a Demand Registration and the Company shall effect, as soon thereafter as practicable, the Registration of all Registrable Securities requested by the Demanding Holders and Requesting Holders pursuant to such Demand Registration.

(b) Notwithstanding the provisions of Section 2.2(a) or any other part of this Agreement, a Registration pursuant to a Demand Registration shall not count as a Registration unless and until (i) the Registration Statement filed with the Commission with respect to a Registration pursuant to a Demand Registration has been declared effective by the Commission and (ii) the Company has complied with all of its obligations under this Agreement with respect thereto; provided, further, that if, after such Registration Statement has been declared effective, an offering of Registrable Securities in a Registration pursuant to a Demand Registration is subsequently interfered with by any stop order or injunction of the Commission, federal or state court or any other governmental agency, the Registration Statement with respect to such Registration shall be deemed not to have been declared effective, unless and until, (1) such

stop order or injunction is removed, rescinded or otherwise terminated, and (2) a majority-in-interest of the Demanding Holders initiating such Demand Registration thereafter affirmatively elect to continue with such Registration and accordingly notify the Company in writing, but in no event later than ten (10) days after the removal, rescission or other termination of such stop order or injunction, of such election; provided, further, that the Company shall not be obligated or required to file another Registration Statement until the Registration Statement that has been previously filed with respect to a Registration pursuant to a Demand Registration by the same Demand Holder becomes effective or is subsequently terminated.

(c) Subject to the provisions of Section 2.2(d) and Section 2.7, if a majority-in-interest of the Demanding Holders so advise the Company as part of their Demand Registration that the offering of the Registrable Securities pursuant to such Demand Registration shall be in the form of an Underwritten Offering, then the right of such Demanding Holder or Requesting Holder (if any) to include its Registrable Securities in such Registration shall be conditioned upon such Holder's participation in such Underwritten Offering and the inclusion of such Holder's Registrable Securities in such Underwritten Offering to the extent provided herein. All such Holders proposing to distribute their Registrable Securities through an Underwritten Offering under this Section 2.2(c), subject to Section 3.3 and Article IV, shall enter into an underwriting agreement in customary form with the Company and the Underwriter(s) selected for such Underwritten Offering by the majority-in-interest of the Demanding Holders initiating the Demand Registration, which Underwriter(s) shall be reasonably satisfactory to the Company.

(d) If a Demand Registration is to be an Underwritten Offering and the managing Underwriter or Underwriters, in good faith, advises the Company, the Demanding Holders and the Requesting Holders (if any) in writing that, in its opinion, the dollar amount or number of Registrable Securities that the Demanding Holders and the Requesting Holders (if any) desire to sell, taken together with all other Registrable Securities or other equity securities that the Company desires to sell for its own account and the Registrable Securities, if any, as to which a Registration has been requested pursuant to separate written contractual piggy-back registration rights held by any other stockholders of the Company who desire to sell, exceeds the maximum dollar amount or maximum number of equity securities that can be sold in such Underwritten Offering without adversely affecting the proposed offering price, the timing, the distribution method, or the probability of success of such offering (such maximum dollar amount or maximum number of such securities, as applicable, the "Maximum Number of Securities"), then the Company shall include in such Underwritten Offering, as follows: (i) first, the Registrable Securities of the Demanding Holders and the Requesting Holders (if any) (pro rata based on the total amount of Registrable Securities held by each such Demanding Holder and Requesting Holder (if any) (such proportion is referred to herein as "Pro Rata")) that can be sold without exceeding the Maximum Number of Securities; (ii) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (i), the Registrable Securities or other equity securities that the Company desires to sell for its own account, which can be sold without exceeding the Maximum Number of Securities; and (iii) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i) and (ii), the Registrable Securities or other equity securities of other persons or entities that the Company is obligated to register in a Registration pursuant to separate written contractual arrangements with such persons and that can be sold without exceeding the Maximum Number of Securities.

**Section 2.3** Withdrawal of Securities. A Demanding Holder or a Requesting Holder shall have the right to withdraw all or a portion of its Registrable Securities included in a Demand Registration pursuant to Section 2.2(a) or a Shelf Underwritten Offering pursuant to Section 2.1(c) for any or no reason whatsoever upon written notification to the Company and the Underwriter or Underwriters (if any) of its intention to so withdraw at any time prior to (i) in the case of a Demand

Registration not involving an Underwritten Offering, the effectiveness of the applicable Registration Statement or (ii) in the case of any Demand Registration involving an Underwritten Offering or any Shelf Underwritten Offering, prior to the pricing of such Underwritten Offering or Shelf Underwritten Offering; provided, however, that upon withdrawal by a majority-in-interest of the Demanding Holders initiating a Demand Registration (or in the case of a Shelf Underwritten Offering, withdrawal of an amount of Registrable Securities included by the Holders in such Shelf Underwritten Offering, in their capacity as Demanding Holders, being less than the Minimum Amount), the Company shall cease all efforts to secure effectiveness of the applicable Registration Statement or complete the Underwritten Offering, as applicable. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with a Registration pursuant to a Demand Registration or a Shelf Underwritten Offering prior to and including its withdrawal under this Section 2.3.

#### Section 2.4 Piggyback Registration.

(a) If the Company proposes to file a Registration Statement under the Securities Act with respect to an offering of equity securities, or securities or other obligations exercisable or exchangeable for, or convertible into equity securities, for its own account or for the account of stockholders of the Company (or by the Company and by the stockholders of the Company including pursuant to Section 2.2), other than an Exempted Registration Statement, then the Company shall give written notice of such proposed filing to all of the Holders of Registrable Securities as soon as practicable but not less than ten (10) days (or, in the case of a Block Trade, five (5) Business Days) before the anticipated filing date of such Registration Statement, which notice shall (i) describe the amount and type of securities to be included in such offering, the intended method(s) of distribution (including whether such registration will be pursuant to a shelf Registration Statement), and the proposed price and name of the proposed managing Underwriter or Underwriters, if any, in such offering, (ii) such Holders' rights under this Section 2.4 and (iii) offer to all of the Holders of Registrable Securities the opportunity to register the sale of such number of Registrable Securities as such Holders may request in writing within five (5) days after receipt of such written notice (or in the case of a Block Trade, within two (2) Business Days) (such Registration a "Piggyback Registration"). The Company shall, in good faith, cause such Registrable Securities identified in a Holder's response notice described in the foregoing sentence to be included in such Piggyback Registration and shall use its reasonable best efforts to cause the managing Underwriter or Underwriters of a proposed Underwritten Offering, if any, to permit the Registrable Securities requested by the Holders pursuant to this Section 2.4(a) to be included in a Piggyback Registration on the same terms and conditions as any similar securities of the Company or Company stockholder(s) for whose account the Registration Statement is to be filed included in such Registration and to permit the sale or other disposition of such Registrable Securities in accordance with the intended method(s) of distribution thereof. All such Holders proposing to distribute their Registrable Securities through an Underwritten Offering under this Section 2.4(a), subject to Section 3.3 and Article IV, shall enter into an underwriting agreement in customary form with the Underwriter(s) selected for such Underwritten Offering by the Company or Company stockholder(s) for whose account the Registration Statement is to be filed. For purposes of this Section 2.4, the filing by the Company of an automatic shelf Registration Statement for offerings pursuant to Rule 415(a) that omits information with respect to any specific offering pursuant to Rule 430B shall not trigger any notification or participation rights hereunder until such time as the Company amends or supplements such Registration Statement to include information with respect to a specific offering of Securities (and such amendment or supplement shall trigger the notice and participation rights provided for in this Section 2.4).

(b) If a Piggyback Registration is to be an Underwritten Offering and the managing Underwriter or Underwriters, in good faith, advises the Company and the Holders of Registrable Securities participating in the Piggyback Registration in writing that, in its opinion, the dollar amount or number of the Registrable Securities that the Company desires to sell, taken together with (x) the

Registrable Securities, if any, as to which Registration has been demanded pursuant to separate written contractual arrangements with persons or entities other than the Holders of Registrable Securities hereunder, (y) the Registrable Securities as to which registration has been requested pursuant Section 2.4, and (z) the Registrable Securities, if any, as to which Registration has been requested pursuant to separate written contractual piggy-back registration rights of other stockholders of the Company, exceeds the Maximum Number of Securities, then:

- (i) if the Registration is undertaken for the Company's account, the Company shall include in any such Registration (1) first, the Registrable Securities or other equity securities that the Company desires to sell for its own account, which can be sold without exceeding the Maximum Number of Securities, (2) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (1), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to Section 2.4(a), Pro Rata, which can be sold without exceeding the Maximum Number of Securities and (3) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (1) and (2), the Registrable Securities, if any, as to which Registration has been requested pursuant to written contractual piggy-back registration rights of other stockholders of the Company, which can be sold without exceeding the Maximum Number of Securities; and
- (ii) if the Registration is pursuant to a request by persons or entities other than the Holders of Registrable Securities, then the Company shall include in any such Registration (1) first, the Registrable Securities or other equity securities, if any, of such requesting persons or entities, other than the Holders of Registrable Securities, which can be sold without exceeding the Maximum Number of Securities of Holders exercising their rights to register their Registrable Securities pursuant to this Section 2.4(a), Pro Rata, which can be sold without exceeding the Maximum Number of Securities, (3) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (1) and (2), the Registrable Securities or other equity securities that the Company desires to sell for its own account, which can be sold without exceeding the Maximum Number of Securities and (4) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (1), (2) and (3), the Registrable Securities or other equity securities for the account of other persons or entities that the Company is obligated to register pursuant to separate written contractual arrangements with such persons or entities, which can be sold without exceeding the Maximum Number of Securities.
- (c) Any Holder of Registrable Securities shall have the right to withdraw all or any portion of its Registrable Securities in a Piggyback Registration for any or no reason whatsoever upon written notification to the Company and the Underwriter or Underwriters (if any) of his, her or its intention to withdraw such Registrable Securities from such Piggyback Registration prior to (i) in the case of a Piggyback Registration not involving an Underwritten Offering or Shelf Underwritten Offering, the effectiveness of the applicable Registration Statement or (ii), in the case of any Piggyback Registration involving an Underwritten Offering or any Shelf Underwritten Offering, prior to the pricing of such Underwritten Offering or Shelf Underwritten Offering. The Company (whether on its own good faith determination or as the result of a request for withdrawal by persons pursuant to separate written contractual obligations) may withdraw a Registration Statement filed with the Commission in connection with a Piggyback Registration at any time prior to the effectiveness of such Registration Statement. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the

Registration Expenses incurred in connection with the Piggyback Registration prior to and including its withdrawal under this Section 2.4(c).

(d) For purposes of clarity, any Registration effected pursuant to <u>Section 2.4</u> shall not be counted as a Registration pursuant to a Demand Registration effected under <u>Section 2.2</u> or a Shelf Underwritten Offering effected under <u>Section 2.1(c)</u>.

Section 2.5 <u>Block Trades</u>. Subject to Section 2.7, if the Holders desire to effect a Block Trade, then notwithstanding any other time periods in this Article II, the Holders shall provide written notice to the Company at least five (5) Business Days prior to the date such Block Trade will commence. As expeditiously as possible, the Company shall use its reasonable best efforts to facilitate such Block Trade. The Holders shall use reasonable best efforts to work with the Company and the Underwriters (including by disclosing the maximum number of Registrable Securities proposed to be the subject of such Block Trade) in order to facilitate preparation of the Registration Statement, Prospectus and other offering documentation related to the Block Trade and any related due diligence and comfort procedures. In the event of a Block Trade, and after consultation with the Company, the Demanding Holders and the Requesting Holders (if any) shall determine the Maximum Number of Securities, the underwriter or underwriters and share price of such offering.

Section 2.6 Rule 415; Removal. If at any time the Commission takes the position that the offering of some or all of the Registrable Securities in a Registration Statement on Form F-3 filed pursuant to this Article II is not eligible to be made on a delayed or continuous basis under the provisions of Rule 415 under the Securities Act (provided, however, the Company shall be obligated to use diligent efforts to advocate with the Commission for the registration of all of the Registrable Securities in accordance with the Commission Guidance, including Compliance and Disclosure Interpretation 612.09) or requires a Holder to be named as an "underwriter," the Company shall (a) promptly notify each holder of Registrable Securities thereof (or in the case of the Commission requiring a Holder to be named as an "underwriter," the Holder) and (b) use reasonable best efforts to persuade the SEC that the offering contemplated by such Registration Statement is a valid secondary offering and not an offering "by or on behalf of the issuer" as defined in Rule 415 and that none of the Holders is an "underwriter." The Holders shall have the right to select one legal counsel designated by the holders of a majority of the Registrable Securities subject to such Registration Statement to review and oversee any registration or matters pursuant to this Section 2.6, including participation in any meetings or discussions with the Commission regarding the Commission's position and to comment on any written submission made to the Commission with respect thereto. No such written submission with respect to this matter shall be made to the Commission to which the applicable Holders' counsel reasonably objects. In the event that, despite the Company's reasonable best efforts and compliance with the terms of this Section 2.6, the Commission refuses to alter its position, the Company shall (a) remove from such Registration Statement such portion of the Registrable Securities (the "Removed Securities") or (b) agree to such restrictions and limitations on the registration and resale of the Registrable Securities as the Commission may require to assure the Company's compliance with the requirements of Rule 415; provided, however, that the Company shall not agree to name any Holder as an "underwriter" in such Registration Statement without the prior written consent of such Holder. In the event of a share removal pursuant to this Section 2.6, the Company shall give the applicable Holders at least five (5) days prior written notice along with the calculations as to such Holder's allotment. Subject to the Company's obligations under the Subscription Agreements, any removal of shares of the Holders pursuant to this Section 2.6 shall first be applied to Holders other than the Holders with securities registered for resale under the applicable Registration Statement and thereafter allocated between the Holders on a pro rata basis based on the aggregate amount of Registrable Securities held by the Holders. In the event of a share removal of the Holders pursuant to this Section 2.6, the Company shall promptly register the resale of any Removed Securities pursuant to Section 2.1(b). In the case of a Form F-1 Shelf filed to register the resale of Removed Securities, upon such date as the

Company becomes eligible to register all of the Removed Securities for resale on a Form F-3 Shelf pursuant to the Commission Guidance and, if applicable, without a requirement that any of the Holders be named as an "underwriter" therein, the Company shall use its reasonable best efforts to file a Form F-3 Shelf as promptly as practicable to replace the applicable Form F-1 Shelf and have the Form F-3 Shelf declared effective as promptly as practicable and to cause such Form F-3 Shelf to remain effective, and to be supplemented and amended to the extent necessary to ensure that such Registration Statement is available or, if not available, that another Registration Statement is available, for the resale of all the Registrable Securities thereunder held by the applicable Holders until all such Registrable Securities have ceased to be Registrable Securities.

# Section 2.7 Restrictions on Registration and Block Trade Rights; Suspension of Sales; Adverse Disclosure.

- (a) Notwithstanding anything in this Agreement to the contrary, in no event will the Gores Holders be entitled, on a collective basis, to initiate more than an aggregate of three (3) Registrations pursuant to a Demand Registration or a Shelf Underwritten Offering (including a Block Trade) under Section 2.1(c) (including with respect to a Block Trade) or Section 2.2(a), as the case may be; provided, however, that a Registration shall not be counted for such purposes unless a Registration Statement that may be available at such time has become effective.
- (b) Notwithstanding anything in this Agreement to the contrary, in no event will the Parent Holders be entitled, on a collective basis, to initiate more than (i) an aggregate of ten (10) Registrations pursuant to a Demand Registration, (ii) more than four (4) Shelf Underwritten Offerings in any twelve month period or (iii) more than four (4) Block Trades in any twelve month period, provided that the estimated proceeds of any such Block Trade shall not be less than \$25,000,000, in each of the foregoing clauses (i), (ii) and (iii) pursuant to Section 2.1(c) or Section 2.2(a), as the case may be; provided, however, that a Registration shall not be counted for such purposes unless a Registration Statement that may be available at such time has become effective.
- (c) Notwithstanding anything to the contrary contained herein, upon receipt of written notice from the Company that a Registration Statement or Prospectus contains a Misstatement, or in the opinion of counsel for the Company it is necessary to supplement or amend such Prospectus to comply with law, each of the Holders shall forthwith discontinue disposition of Registrable Securities until it has received copies of a supplemented or amended Prospectus correcting the Misstatement or including the information counsel for the Company believes to be necessary to comply with law (it being understood that the Company hereby covenants to prepare and file such supplement or amendment as soon as practicable after the time of such notice such that the Registration Statement or Prospectus, as so amended or supplemented, as applicable, will not include a Misstatement and complies with law), or until it is advised in writing by the Company that the use of the Prospectus may be resumed. The Company shall promptly notify the Holders of the expiration of any period during which it exercised its rights under this Section 2.7(b).
- (d) Notwithstanding anything to the contrary contained herein, the Company shall not be obligated to (i) effect any Demand Registration or Underwritten Offering or (ii) file a Registration Statement (or any amendment thereto) or effect an Underwritten Offering (or, if the Company has filed a Registration Statement that includes Registrable Securities, the Company shall be entitled to suspend the offer and sale of Registrable Securities pursuant to such Registration Statement) for a period of up to forty-five (45) days if the Company has determined that the sale of Registrable Securities pursuant a Registration Statement would require the Company to make an Adverse Disclosure, would require the inclusion in such Registration Statement of financial statements that are unavailable to the Company for reasons beyond the Company's control or in the good faith judgment of the Board would be materially

detrimental to the Company (any such period, a "Blackout Period"); provided, however, that in no event shall any Blackout Period together with other Blackout Periods exceed an aggregate of one hundred and twenty (120) days in any twelve (12)-month period. In the event the Company exercises its rights under this Section 2.7(c), the Holders agree to suspend, immediately upon their receipt of the notice referred to above, their use of any Prospectus relating to any Registration in connection with any sale or offer to sell Registrable Securities. The Company shall promptly notify the Holders of the expiration of any period during which it exercised its rights under this Section 2.7(d). The Holders receiving notice of a Blackout Period (and the expiration thereof) shall maintain the confidentiality of the existence (and circumstances, to the extent known) of a Blackout Period. In connection with any notice of a Blackout Period, the Company may not deliver any material non-public information, and, for avoidance of doubt, solely the receipt of notice of a Blackout Period without additional information shall not constitute material non-public information.

# ARTICLE III COMPANY PROCEDURES

- **Section 3.1** <u>General Procedures</u>. If the Company is required to effect the Registration of Registrable Securities, the Company shall use its reasonable best efforts to effect such Registration to permit the sale of such Registrable Securities in accordance with the intended method of distribution thereof, and pursuant thereto the Company shall, as expeditiously as reasonably possible:
- (a) prepare and file with the Commission as soon as reasonably practicable a Registration Statement with respect to such Registrable Securities and use its reasonable best efforts to cause such Registration Statement to become effective and remain effective until all Registrable Securities covered by such Registration Statement have been sold;
- (b) prepare and file with the Commission such amendments and post-effective amendments to the Registration Statement, and such supplements to the Prospectus as may be required by the Securities Act or rules and regulations thereunder to keep the Registration Statement effective until all Registrable Securities covered by such Registration Statement are sold in accordance with the intended method of distribution set forth in such Registration Statement or supplement to the Prospectus;
- (c) prior to filing a Registration Statement or Prospectus, or any amendment or supplement thereto, furnish without charge to the Underwriters, if any, and the Holders of Registrable Securities included in such Registration, and such Holders' legal counsel, copies of such Registration Statement as proposed to be filed, each amendment and supplement to such Registration Statement (in each case including all exhibits thereto and documents incorporated by reference therein), the Prospectus included in such Registration Statement (including each preliminary Prospectus), and such other documents as the Underwriters and the Holders of Registrable Securities included in such Registration or the legal counsel for any such Holders may reasonably request in order to facilitate the disposition of the Registrable Securities owned by such Holders;
- (d) prior to any public offering of Registrable Securities, but in any case no later than the effective date of the applicable Registration Statement, use its reasonable best efforts to (i) register or qualify the Registrable Securities covered by the Registration Statement under such securities or "blue sky" laws of such jurisdictions in the United States as the Holders of Registrable Securities included in such Registration Statement (in light of their intended method of distribution) may reasonably request and to keep such registration or qualification in effect for so long as such Registration Statement remains in effect and (ii) take such action as may be necessary to cause such Registrable Securities covered by the Registration Statement to be registered with or approved by such other governmental authorities as may be necessary by virtue of the business and operations of the Company or otherwise and do any and all

other acts and things that may be necessary to enable the Holders of Registrable Securities included in such Registration Statement to consummate the disposition of such Registrable Securities in such jurisdictions; <u>provided</u>, <u>however</u>, that the Company shall not be required to qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify or take any action to which it would be subject to general service of process or taxation in any such jurisdiction where it is not then otherwise so subject;

- (e) cause all such Registrable Securities to be listed on Nasdaq no later than the effective date of such Registration Statement;
- (f) provide a transfer agent or warrant agent, as applicable, and registrar for all such Registrable Securities no later than the effective date of such Registration Statement;
- (g) furnish to each seller of Registrable Securities covered by such Registration Statement such number of conformed copies of such Registration Statement and of each such amendment and supplement thereto (in each case including all exhibits), such number of copies of the Prospectus contained in such Registration Statement (including each preliminary Prospectus and any summary Prospectus) and any other Prospectus filed under Rule 424 under the Securities Act, in conformity with the requirements of the Securities Act, and such other documents as such seller may reasonably request;
- (h) advise each seller of such Registrable Securities, promptly after it shall receive notice or obtain knowledge thereof, of the issuance of any stop order by the Commission suspending the effectiveness of such Registration Statement or Prospectus the initiation or threatening of any proceeding for such purpose and promptly use its reasonable best efforts to amend or supplement such Registration Statement or Prospectus or prevent the issuance of any stop order or to obtain its withdrawal if such stop order should be issued, as applicable;
- (i) advise each Holder of Registrable Securities covered by such Registration Statement, promptly after the Company receives notice thereof, of the time when such registration statement has been declared effective or a supplement to any Prospectus forming a part of such registration statement has been filed;
- (j) notify the Holders at any time when a Prospectus relating to such Registration Statement is required to be delivered under the Securities Act, of the happening of any event or the existence of any condition as a result of which the Prospectus included in such Registration Statement, as then in effect, includes a Misstatement, or in the opinion of counsel for the Company it is necessary to supplement or amend such Prospectus to comply with law, and then to correct such Misstatement or include such information as is necessary to comply with law, in each case as set forth in Section 2.7, at the request of any such Holder promptly prepare and furnish to such Holder a reasonable number of copies of a supplement to or an amendment of such Prospectus as may be necessary so that, as thereafter delivered to the purchasers of such securities, such Prospectus shall not include a Misstatement or such Prospectus, as supplemented or amended, shall comply with law;
- (k) make available for inspection, at such place and in such manner as determined by the Company in its sole discretion, permit a representative of the Holders, the Underwriters, if any, and any attorney or accountant retained by such Holders or Underwriter access to the Company's books and records and such opportunities to discuss the business, finances and accounts of the Company and its subsidiaries with its officers, directors and the independent public accountants who have certified its financial statements as shall be necessary, in the opinion of such Holders' and such Underwriters' respective counsel, to conduct a reasonable investigation within the meaning of the Securities Act, and will cause the Company's officers, directors and employees to supply all information reasonably

requested by any such representative, Underwriter, attorney or accountant in connection with the Registration; <u>provided</u>, <u>however</u>, that (<u>i</u>) if requested by the Company, such representatives, Holders or Underwriters enter into a confidentiality agreement, in form and substance reasonably satisfactory to the Company, prior to the release or disclosure of any such information and (<u>ii</u>) any records, information or documents that are furnished by the Company and that are non-public shall be used only in connection with the Registration;

- (l) obtain a "comfort" letter from the Company's independent registered public accountants in the event of an Underwritten Offering, in customary form and covering such matters of the type customarily covered by "comfort" letters as the managing Underwriter may reasonably request, and reasonably satisfactory to a majority-in-interest of the participating Holders and any Underwriter;
- (m) on the date the Registrable Securities are delivered for sale pursuant to such Registration, obtain an opinion and negative assurance letter, dated such date, of counsel representing the Company for the purposes of such Registration, addressed to the placement agent or sales agent, if any, and the Underwriters, if any, covering such legal matters with respect to the Registration as are customarily included in such opinions and negative assurance letters;
- (n) in the event of any Underwritten Offering, enter into an underwriting agreement, in usual and customary form, with the managing Underwriter of such offering;
- (o) otherwise use its reasonable best efforts to comply with all applicable rules and regulations of the Commission, and to make available to its security holders, as soon as reasonably practicable, an earnings statement covering a period of at least twelve (12) months which satisfies the provisions of Section 11(a) of the Securities Act and the rules and regulations thereunder, including Rule 158 thereunder (or any successor rule promulgated thereafter by the Commission);
- (p) use its reasonable efforts to make available senior executives of the Company to participate in customary "road show" presentations that may be reasonably requested by the Underwriter in an Underwritten Offering;
- (q) otherwise, in good faith, cooperate reasonably with, and take such customary actions as may reasonably be requested by the Holders, in connection with such Underwritten Offering; and
- (r) assist the Depository Bank to maintain an effective registration of the Class A ADSs, Converted Class C-1 ADSs, Class C-2 ADSs or ListCo AD Warrants on Form F-6 in accordance with the Deposit Agreement and cooperate with the Depositary Bank in filing amendments to such Form F-6 sufficient to allow the Holders to exercise its rights hereunder and under the Deposit Agreement to cover the Registerable Securities then outstanding.

**Section 3.2 Registration Expenses**. The Registration Expenses of all Registrations shall be borne by the Company. It is acknowledged by the Holders that the Holders shall bear all incremental selling expenses relating to the sale of Registrable Securities, such as Underwriters' commissions and discounts, brokerage fees and, other than as set forth in the definition of "Registration Expenses," all fees and expenses of any legal counsel representing the Holders. In addition, the Company will pay its internal expenses (including all salaries and expenses of its officers and employees performing legal or accounting duties), the expense of any annual audit or quarterly review, the expense of any liability insurance obtained by the Company and the expenses and fees for listing the securities to be registered on each securities exchange.

#### Section 3.3 Participation in Underwritten Offerings.

- (a) No person may participate in any Underwritten Offering for equity securities of the Company pursuant to a Registration initiated by the Company hereunder unless such person (i) agrees to sell such person's securities on the basis provided in any underwriting arrangements approved by the Company and (ii) completes and executes all customary questionnaires, indemnities, lock-up agreements, underwriting agreements and other customary documents as may be reasonably required under the terms of such underwriting arrangements.
- (b) The Company will use its reasonable best efforts to ensure that no Underwriter shall require any Holder to make any representations or warranties to or agreements with the Company or the Underwriters other than representations, warranties or agreements regarding such Holder and such Holder's intended method of distribution and any other representation required by law, and if, despite the Company's reasonable best efforts, an Underwriter requires any Holder to make additional representation or warranties to or agreements with such Underwriter, such Holder may elect not to participate in such Underwritten Offering (but shall not have any claims against the Company as a result of such election). Any liability of such Holder to any Underwriter or other person under such underwriting agreement shall be limited to an amount equal to the proceeds (net of expenses and underwriting discounts and commissions) that it derives from such registration.

**Section 3.4** Covenants of the Company. As long as any Holder shall own Registrable Securities, the Company hereby covenants and agrees:

- (a) the Company will not file any Registration Statement or Prospectus included therein with the Commission which refers to any Holder of Registrable Securities by name or otherwise without the prior written approval of such Holder, which may not be unreasonably withheld, conditioned or delayed;
- (b) at all times while it shall be a reporting company under the Exchange Act, to file timely (or obtain extensions in respect thereof and file within the applicable grace period) all reports required to be filed by the Company after the closing of the Transactions pursuant to Sections 13(a) or 15(d) of the Exchange Act and to promptly furnish the Holders with true and complete copies of all such filings. The Company further covenants that it shall take such further action as any Holder may reasonably request, all to the extent required from time to time to enable such Holder to sell Registerable Securities held by such Holder without registration under the Securities Act within the limitation of the exemptions provided by Rule 144 promulgated under the Securities Act (or any successor rule promulgated thereafter by the Commission), including providing any legal opinions. Upon the request of any Holder, the Company shall deliver to such Holder a written certification of a duly authorized officer as to whether it has complied with such requirements; and
- (c) promptly following the effectiveness of the shelf Registration Statement required by Section 2.1(a), the Company shall cause the transfer agent to remove any restrictive legends (including any electronic transfer restrictions) from any Shares, Class C-1 Shares, Class C-2 Shares, ListCo Warrants, Class A ADSs, Converted Class C-1 ADSs, Class C-2 ADSs or ListCo AD Warrants held by such Holder and provide or cause any customary opinions of counsel to be delivered to the transfer agent in connection with such removal.

# ARTICLE IV INDEMNIFICATION AND CONTRIBUTION

Section 4.1 Indemnification.

- (a) The Company agrees to indemnify, to the extent permitted by law, each Holder of Registrable Securities, its officers, directors, partners, stockholders or members, employees, agents, investment advisors and each person who controls such Holder (within the meaning of the Securities Act and Exchange Act) from and against all losses, claims, damages, liabilities and expenses (including attorneys' fees), joint or several (or actions or proceedings, whether commenced or threatened, in respect thereof) (collectively, "Claims"), to which any such Holder or other persons may become subject, insofar as such Claims arise out of or are based on any untrue or alleged untrue statement of any material fact contained in any Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, and the Company will reimburse such Holder or other person for any legal or any other expenses reasonably incurred by them in connection with investigating or defending any such Claim; except insofar as the Claim or expense arises out of or is based upon an untrue statement or alleged untrue statement or omission or alleged omission made in such filing in reliance upon and in conformity with information furnished in writing to the Company by such Holder expressly for use therein. The Company shall indemnify the Underwriters, their officers and directors and each person who controls such Underwriters (within the meaning of the Securities Act and Exchange Act) to the same extent as provided in the foregoing with respect to the indemnification of the Holder.
- (b) In connection with any Registration Statement in which a Holder of Registrable Securities is participating, as a condition to including any Registrable Securities in any Registration Statement, the Company shall have received an undertaking reasonably satisfactory to it from such Holder, to indemnify the Company, its directors and officers and agents and each person who controls the Company (within the meaning of the Securities Act and Exchange Act) from and against any Claims, to which any the Company or such other persons may become subject, insofar as such Claims arise out of or are based on any untrue statement of any material fact contained in the Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, but only to the extent that such untrue statement or omission is contained in any information furnished in writing by such Holder expressly for use therein; provided, however, that the obligation to indemnify shall be several, not joint and several, among such Holders of Registrable Securities, and the liability of each such Holder of Registrable Securities shall be in proportion to and limited to the net proceeds received by such Holder from the sale of Registrable Securities pursuant to such Registration Statement. The Holders of Registrable Securities shall indemnify the Underwriters, their officers, directors and each person who controls such Underwriters (within the meaning of the Securities Act and Exchange Act) to the same extent as provided in the foregoing with respect to indemnification of the Company and the Company shall use its reasonable best efforts to ensure that no Underwriter shall require any Holder of Registrable Securities to provide any indemnification other than that provided hereinabove in this Section 4.1(b), and, if, despite the Company's reasonable best efforts, an Underwriter requires any Holder of Registrable Securities to provide additional indemnification, such Holder may elect not to participate in such Underwritten Offering (but shall not have any claim against the Company as a result of such election).
- (c) Any person entitled to indemnification herein shall (i) give prompt written notice to the indemnifying party of any Claim with respect to which it seeks indemnification (provided that the failure to give prompt notice shall not impair any person's right to indemnification hereunder to the extent such failure has not materially prejudiced the indemnifying party) and (ii) unless in such indemnified party's reasonable judgment a conflict of interest between such indemnified and indemnifying parties may exist with respect to such Claim, permit such indemnifying party to assume the defense of such Claim with counsel reasonably satisfactory to the indemnified party. If such defense is assumed, the indemnifying party shall not be subject to any liability for any settlement made by the indemnified party without its consent (but such consent shall not be unreasonably withheld). An indemnifying party who is

not entitled to, or elects not to, assume the defense of a claim shall not be obligated to pay the fees and expenses of more than one (1) counsel for all parties indemnified by such indemnifying party with respect to such claim, unless in the reasonable judgment of any indemnified party a conflict of interest may exist between such indemnified party and any other of such indemnified parties with respect to such claim. No indemnifying party shall, without the consent of the indemnified party, consent to the entry of any judgment or enter into any settlement which cannot be settled in all respects by the payment of money (and such money is so paid by the indemnifying party pursuant to the terms of such settlement) and which settlement includes a statement or admission of fault or culpability on the part of such indemnified party or does not include as an unconditional term thereof the giving by the claimant or plaintiff to such indemnified party of a release from all liability in respect to such claim or litigation.

- (d) The indemnification provided for under this Agreement shall remain in full force and effect regardless of any investigation made by or on behalf of the indemnified party or any officer, director, partners, stockholders or members, employees, agents, investment advisors or controlling person of such indemnified party and shall survive the Transfer of Registrable Securities.
- (e) If the indemnification provided under Section 4.1 from the indemnifying party is unavailable or insufficient to hold harmless an indemnified party in respect of any Claims, then the indemnifying party, in lieu of indemnifying the indemnified party, shall contribute to the amount paid or payable by the indemnified party as a result of such Claims (i) in such proportion as is appropriate to reflect the relative benefits received by the indemnifying party or parties on the one hand and the indemnified party or parties on the other hand from the offering of the Registrable Securities or (ii) if the allocation provided by clause (i) above is not permitted by applicable law, in such proportion as is appropriate to reflect not only the relative benefits referred to in clause (ii) above but also to reflect the relative fault of the indemnifying party or parties on the other hand in connection with the statements or omissions that resulted in such Claims, as well as any other relevant equitable considerations; provided, however, that the liability of any Holder or any director, officer, employee, agent, investment advisor or controlling person thereof under this Section 4.1(e) shall be limited to the amount of the net proceeds received by such Holder in such offering giving rise to such liability. The amount paid or payable by a party as a result of the losses or other liabilities referred to above shall be deemed to include, subject to the limitations set forth in Section 4.1(e), Section 4.1(b) and Section 4.1(c) above, any legal or other fees, charges or expenses reasonably incurred by such party in connection with any investigation or proceeding. The parties hereto agree that it would not be just and equitable if contribution pursuant to this Section 4.1(e) were determined by pro rata allocation or by any other method of allocation, which does not take account of the equitable considerations referred to in this Section 4.1(e). No person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) o

# ARTICLE V MISCELLANEOUS

**Section 5.1** <u>Notices</u>. All notices, requests, claims, demands and other communications hereunder shall be in writing and shall be deemed to have been given when delivered (a) in person or (b) by e-mail or other means of electronic transmission (so long as confirmation of transmission is electronically or mechanically generated or sent and kept on file by the sending party, and no "bounceback" or notice of non-delivery is received), and the sender may, in its sole discretion, deliver a copy by mail (postage prepaid) or by an internationally-recognized courier service (postage prepaid). Notices shall be given to the respective Parties at the following addresses (or at such other address for a Party as shall be specified in a notice given in accordance with this Section 5.1).

(a) If to the Company or Parent:

Assar Gabrielssons Väg 9

405 31 Göteborg, Sweden Attention: Anna Rudensjö

Email: anna.rudensjo@polestar.com

with a copy (which shall not constitute notice) to:

Kirkland & Ellis LLP 601 Lexington Avenue New York, NY 10022, USA Attention: Christian O. Nagler Timothy Cruickshank Alex Lloyd

Email: cnagler@kirkland.com tim.cruickshank@kirkland.com alex.lloyd@kirkland.com

If to any Gores Holder or Parent Holder, to such Holder's address as set forth on the signature page hereto.

#### Section 5.2 Assignment; No Third Party Beneficiaries.

- (a) This Agreement and the rights, duties and obligations of the Company hereunder may not be assigned or delegated by the Company in whole or in part.
- (b) Prior to the expiration of any lock-up as described in the Lock-Up Agreements, no Holder may assign or delegate such Holder's rights, duties or obligations under this Agreement, in whole or in part, except in connection with a transfer of Registrable Securities by such Holder to a Permitted Transferee (but subject to such Permitted Transferee, if required pursuant to the Lock-Up Agreements, agreeing in writing, in form and substance reasonably satisfactory to the Company, to be bound by the provisions of this Agreement).
- (c) This Agreement and the provisions hereof shall be binding upon and shall inure to the benefit of each of the parties and its successors and the permitted assigns of the applicable Holders, which shall include Permitted Transferees.
- (d) This Agreement shall not confer any rights or benefits on any persons that are not parties hereto, other than as expressly set forth in this Agreement and this Section 5.2.
- (e) No assignment by any party hereto of such party's rights, duties and obligations hereunder shall be binding upon or obligate the Company unless and until the Company shall have received (i) written notice of such assignment as provided in Section 5.1 and (ii) the written agreement of the assignee, in a form reasonably satisfactory to the Company, to be bound by the terms and provisions of this Agreement (which may be accomplished by an addendum or certificate of joinder to this Agreement). Any Transfer or assignment made other than as provided in this Section 5.2 shall be null and void.

**Section 5.3** Counterparts. This Agreement may be executed in multiple counterparts, each of which when executed and delivered shall thereby be deemed to be an original and all of which taken together shall constitute one and the same instrument. Any party hereto may execute

and deliver signed counterparts of this Agreement to the other Parties by electronic mail or other electronic transmission in portable document format (.PDF) or any other electronic signature complying with the United States ESIGN Act of 2000 (including www.docusign.com), each of which shall be deemed an original.

**Section 5.4** <u>Governing Law</u>. This Agreement, and all claims or causes of action based upon, arising out of, or related to this Agreement or the transactions contemplated herein, shall be governed by, and construed in accordance with, the laws of the State of Delaware, without giving effect to principles or rules of conflict of laws to the extent such principles or rules would require or permit the application of laws of another jurisdiction.

**Section 5.5** <u>Jurisdiction</u>. In any action or proceeding arising out of or relating to this Agreement or any of the transactions contemplated herein: (a) each of the parties hereto hereby irrevocably and unconditionally consents and submits to the exclusive jurisdiction and venue of the Court of Chancery of the State of Delaware (unless the federal courts have exclusive jurisdiction over the matter, in which case the United States District Court for the District of Delaware); (b) each of the parties hereto irrevocably waives and agrees that it will not attempt to deny or defeat such jurisdiction by motion or other request for leave from such court; and (c) each of the parties hereto agrees that it will not bring any such action in any court other than the Court of Chancery of the State of Delaware (unless the federal courts have exclusive jurisdiction over the matter, in which case the United States District Court for the District of Delaware).

Section 5.6 WAIVER OF TRIAL BY JURY. TO THE EXTENT PERMITTED BY LAW, EACH PARTY HEREBY KNOWINGLY, VOLUNTARILY AND INTENTIONALLY WAIVES (AND SHALL CAUSE ITS SUBSIDIARIES AND AFFILIATES TO WAIVE) THE RIGHT TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION BASED HEREON, ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREIN OR ANY COURSE OF CONDUCT, COURSE OF DEALINGS, STATEMENTS (WHETHER VERBAL OR WRITTEN) OR ACTIONS OF ANY PARTY IN CONNECTION HEREWITH. EACH PARTY ACKNOWLEDGES THAT (A) THIS WAIVER IS A MATERIAL INDUCEMENT FOR THE OTHER PARTIES TO ENTER INTO THIS AGREEMENT, AND (B) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THAT FOREGOING WAIVER.

**Section 5.7** Severability. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced by any rule of law, or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated herein is not affected in any manner materially adverse to any party hereto. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the parties hereto shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties hereto as closely as possible in a mutually acceptable manner in order that the transactions contemplated herein be consummated as originally contemplated to the fullest extent possible.

**Section 5.8** <u>Modification or Amendment</u>. Upon the written consent of the Company and the Holders of at least a majority in interest of the Registrable Securities at the time in question, compliance with any of the provisions, covenants and conditions set forth in this Agreement may be waived, or any of such provisions, covenants or conditions may be amended or modified; <u>provided</u>, <u>however</u>, that notwithstanding the foregoing, (a) any amendment hereto or waiver hereof that adversely affects one Holder, solely in its capacity as a holder of Registrable Securities, in a manner that

is adverse and different from the other Holders (in such capacity) shall require the consent of the Holder so affected, (b) any amendment hereto or waiver hereof that adversely affects the Gores Holders solely in their capacity as Gores Holders in a manner that is adverse and different from the other Holders, shall require the consent of a majority-in-interest of the then-outstanding number of Registrable Securities held by the Gores Holders; and (c) any amendment hereto or waiver hereof that adversely affects Parent or the Parent Holders, as applicable, solely in their capacity as Parent or Parent Holders in a manner that is adverse and different from the other Holders, shall require the consent of a majority-in-interest of the then-outstanding number of Registrable Securities held by the Parent or the Parent Holders. No course of dealing between any Holder or the Company and any other party hereto or any failure or delay on the part of a Holder or the Company in exercising any rights or remedies under this Agreement shall operate as a waiver of any rights or remedies of any Holder or the Company. No single or partial exercise of any rights or remedies under this Agreement by a party shall operate as a waiver or preclude the exercise of any other rights or remedies hereunder or thereunder by such party.

**Section 5.9** Other Registration Rights. Other than pursuant to the terms of the Subscription Agreements, the Company represents and warrants that no person, other than a Holder of Registrable Securities, has any right to require the Company to register any securities of the Company for sale or to include such securities of the Company in any Registration filed by the Company for the sale of securities for its own account or for the account of any other person.

Further, the Company represents and warrants that this Agreement supersedes any other registration rights agreement or agreement with similar terms and conditions among the parties thereto and in the event of a conflict between any such agreement or agreements and this Agreement, the terms of this Agreement shall prevail.

**Section 5.10 Term.** This Agreement shall terminate upon the earlier of the date as of which (a) all of the Registrable Securities have been sold pursuant to a Registration Statement (but in no event prior to the applicable period referred to in Section 4(a)(3) of the Securities Act and Rule 174 thereunder (or any successor rule promulgated thereafter by the Commission)) or (b) as to any Holder individually, such Holder is permitted to sell all of such Holder's Registrable Securities under Rule 144 (or any similar provision) under the Securities Act without limitation on the amount of securities sold or the manner of sale and the reporting requirements of Rule 144(i)(2) are not applicable or has otherwise sold all of the Registrable Securities held by such Holder. The provisions of Article IV shall survive any termination to the extent related to a Claim arising prior to the termination of this Agreement.

Section 5.11 Interpretation. The parties hereto have participated jointly in negotiating and drafting this Agreement. If an ambiguity or a question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties hereto, and no presumption or burden of proof shall arise favoring or disfavoring any party hereto by virtue of the authorship of any provision of this Agreement. Whenever the words "include", "includes" or "including" are used in this Agreement they shall be deemed to be followed by the words "without limitation." Words in the singular form will be construed to include the plural and vice versa, unless the context requires otherwise. Reference to any law means such law as amended, modified, codified, replaced or re-enacted, in whole or in part, from time to time, including rules, regulations, enforcement procedures and any interpretations promulgated thereunder. Underscored references to Articles or Sections shall refer to those portions of this Agreement, and any underscored references to a clause shall, unless otherwise identified, refer to the appropriate clause within the same Section in which such reference occurs. The headings for this Agreement are for reference purposes only and do not affect in any way the meaning or interpretation of this Agreement. The use of the terms "hereunder", "hereto", "hereto" and words of similar import shall refer to this Agreement as a whole and not to any particular Article, Section or clause of this Agreement. The word "or" is not exclusive and is deemed to have the meaning "and/or" unless expressly indicated

otherwise. Any reference to "days" means calendar days unless Business Days are expressly specified. When calculating the period of time before which, within which or following which any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded, and if the last day of such period is not a Business Day, the period shall end on the immediately following Business Day. References to "writing", "written" and comparable terms refer to printing, typing and other means of reproducing words (including electronic media) in a visible form. References to a Person are also to its successors and permitted assigns.

[The Remainder of This Page Is Intentionally Left Blank]

IN WITNESS WHEREOF, the undersigned have caused this Agreement to be executed as of the date first written above.

## **COMPANY:**

## POLESTAR AUTOMOTIVE HOLDING UK LIMITED

By: /s/ Mikael Alkmark

Name: Jan Mikael Alkmark

Title: Director

IN WITNESS WHEREOF, the undersigned have caused this Agreement to be executed as of the date first written above.

## PARENT:

## POLESTAR AUTOMOTIVE HOLDING LIMITED

By: <u>/s/ Lars D</u>anielson

Name: Lars Danielson Title: Director

IN WITNESS WHEREOF, the undersigned have caused this Agreement to be executed as of the date first written above.

## PARENT HOLDERS:

ZIBO HIGH-TECH INDUSTRIAL INVESTMENT CO., LTD.

By: /s/ Ning Chen
Name: Ning Chen
Address:

Email:

IN WITNESS WHEREOF, the undersigned have caused this Agreement to be executed as of the date first written above.

## PARENT HOLDERS:

## ${\bf ZIBO\ FINANCIAL\ HOLDING\ GROUP\ CO.,\ LTD.}$

By: /s/ Dapeng Li Name: Dapeng Li Address:

Email:

IN WITNESS WHEREOF, the undersigned have caused this Agreement to be executed as of the date first written above.

## PARENT HOLDERS:

CHONGQING LIANGJIANG (重庆承星股权投资基金合伙企业(有限合伙)

By: <u>/s/ Yimin Ai</u> Name: Yimin Ai Address:

Email:

IN WITNESS WHEREOF, the undersigned have caused this Agreement to be executed as of the date first written above.

## PARENT HOLDERS:

## NORTHPOLE GLY 1 LP

By: /s/ Hrvoje Krkalo Name: Hrvoje Krkalo Address:

Email:

IN WITNESS WHEREOF, the undersigned have caused this Agreement to be executed as of the date first written above.

## PARENT HOLDERS:

## **GLY NEW MOBILITY 1, LP**

By: /s/ Hrvoje Krkalo Name: Hrvoje Krkalo Address:

Email:

IN WITNESS WHEREOF, the undersigned have caused this Agreement to be executed as of the date first written above.

## PARENT HOLDERS:

## SNITA HOLDING B.V.

By: /s/ Lex Kerssemakers

Name: Lex Kerssemakers

Address:

Email:

By: /s/ Carla De Geyseleer

Name: Carla De Geyseleer

Address:

Email:

IN WITNESS WHEREOF, the undersigned have caused this Agreement to be executed as of the date first written above.

## PARENT HOLDERS:

**PSINV AB** 

By: /s/ Fredrik Aaben

Name: Fredrik Aaben

Address:

Email:

By: /s/ Per Ansgar

Name: Per Ansgar

Address:

Email:

IN WITNESS WHEREOF, the undersigned have caused this Agreement to be executed as of the date first written above.

## PARENT HOLDERS:

## PSD INVESTMENT LIMITED

By: /s/ Shufu Li Name: Shufu Li Address:

Email:

IN WITNESS WHEREOF, the undersigned have caused this Agreement to be executed as of the date first written above.

## **GORES HOLDERS:**

## GORES GUGGENHEIM SPONSOR LLC

By: /s/ Alec Gores

Name: Alec Gores Title: Chairman

Address:

Email:

By: /s/ Randall Bort

Name: Randall Bort

Address:

Email:

By: /s/ Elizabeth Marcellino

Name: Elizabeth Marcellino

Address:

Email:

By: /s/ Nancy Tellem

Name: Nancy Tellem

Address:

Email:

## ANNEX V

Form of ListCo New Articles of Association

## THE COMPANIES ACT 2006

## PUBLIC COMPANY LIMITED BY SHARES

## ARTICLES OF ASSOCIATION

OF

## POLESTAR AUTOMOTIVE HOLDING UK PLC

Company No 13624182

1

# CONTENTS

42. No fees on registration

46. Rights on transmission

48. Sub-division

49. <u>Fractions</u>

47. <u>Destruction of documents</u>

43. Other powers in relation to transfers

45. <u>Election of person entitled by transmission</u>

44. Transmission of shares on death

| CLA | CLAUSE   |  |  |
|-----|--|--|--|
| 1.  | Exclusion of model articles (and any other prescribed regulations) |  |  |
| 2.  | <u>Interpretation</u>  |  |  |
| 3.  | Form of resolution   |  |  |
| 4.  | <u>Limited liability</u>   |  |  |
| 5.  | Change of name   |  |  |
| 6.  | The Office   |  |  |
| 7.  | Shares   |  |  |
| 8.  | Right to Return of Assets and Distributions                        |  |  |
| 9.  | Rights and Conversion of Convertible Preferred Shares              |  |  |
| 10. | Rights and Redemption of GBP Redeemable Preferred Shares           |  |  |
| 11. | Rights of Deferred Shares  |  |  |
| 12. | Rights and Conversion of Ordinary Shares                           |  |  |
| 13. | Rights and Conversion of Class C Shares                            |  |  |
| 14. | Power to attach rights to shares                                   |  |  |
| 15. | Allotment of shares and pre-emption                                |  |  |
| 16. | Redeemable shares  |  |  |
| 17. | <u>Pari passu issues</u>   |  |  |
| 18. | <u>Variation of rights</u>   |  |  |
| 19. | Payment of commission  |  |  |
| 20. | <u>Trusts not recognised</u>                                       |  |  |
| 21. | Share certificates   |  |  |
| 22. | Replacement certificates   |  |  |
| 23. | <u>Lien on shares not fully paid</u>                               |  |  |
| 24. | Enforcement of lien by sale  |  |  |
| 25. | <u>Application of proceeds of sale</u>                             |  |  |
| 26. | <u>Calls</u>   |  |  |
| 27. | <u>Liability of joint holders</u>                                  |  |  |
| 28. | <u>Interest on calls</u>   |  |  |
| 29. | Power to differentiate   |  |  |
| 30. | Payment of calls in advance  |  |  |
| 31. | Notice if call or instalment not paid                              |  |  |
| 32. | Forfeiture for non-compliance                                      |  |  |
| 33. | Notice after forfeiture  |  |  |
| 34. | <u>Forfeiture may be annulled</u>                                  |  |  |
| 35. | <u>Surrender</u>   |  |  |
| 36. | Sale of forfeited shares   |  |  |
| 37. | Effect of forfeiture   |  |  |
| 38. | Evidence of forfeiture   |  |  |
| 39. | Form of transfer   |  |  |
| 40. | Right to refuse registration of transfer                           |  |  |
| 41. | Notice of refusal to register a transfer                           |  |  |

| 50.  | Annual general meetings   | 33 |
|------|---|----|
| 51.  | Convening of general meetings   | 33 |
| 52.  | Simultaneous attendance and participation by electronic facilities                      | 34 |
| 53.  | Notice of general meetings  | 34 |
| 54.  | Contents of notice of general meetings  | 34 |
| 55.  | Omission to give notice and non-receipt of notice                                       | 36 |
| 56.  | Postponement of general meeting   | 36 |
| 57.  | Quorum at general meeting   | 36 |
| 58.  | Procedure if quorum not present   | 36 |
| 59.  | Chair of general meeting  | 37 |
| 60.  | Entitlement to attend, speak and participate  | 37 |
| 61.  | Adjournments  | 37 |
| 62.  | Notice of adjournment   | 38 |
| 63.  | Business of adjourned meeting   | 38 |
| 64.  | Accommodation of members, security arrangements and orderly conduct at general meetings | 38 |
| 65.  | Amendment to resolutions  | 39 |
| 66.  | Members' resolutions  | 39 |
| 67.  | Method of voting  | 39 |
| 68.  | Objection to error in voting  | 40 |
| 69.  | Procedure on a poll   | 40 |
| 70.  | <u>Votes of members</u>   | 40 |
| 71.  | No right to vote where sums overdue on shares   | 41 |
| 72.  | <u>Voting by Proxy</u>  | 41 |
| 73.  | Receipt of proxy  | 42 |
| 74.  | Revocation of proxy   | 43 |
| 75.  | <u>Corporate representatives</u>  | 43 |
| 76.  | <u>Failure to disclose interests in shares</u>  | 44 |
| 77.  | Power of sale of shares of untraced members   | 45 |
| 78.  | <u>Application of proceeds of sale of shares of untraced members</u>                    | 46 |
| 79.  | Number of directors   | 46 |
| 80.  | Power of company to appoint directors   | 46 |
| 81.  | Power of board to appoint directors   | 47 |
| 82.  | Eligibility of new directors  | 47 |
| 83.  | Retirement of directors   | 47 |
| 84.  | Removal of directors  | 48 |
| 85.  | <u>Vacation of office by director</u>   | 49 |
| 86.  | Resolution as to vacancy conclusive   | 49 |
| 87.  | Appointment of alternate directors  | 49 |
| 88.  | Alternate directors' participation in board meetings                                    | 50 |
| 89.  | Alternate director responsible for own acts   | 50 |
| 90.  | <u>Interests of alternate director</u>  | 50 |
| 91.  | Revocation of alternate director  | 50 |
| 92.  | <u>Directors' fees</u>  | 50 |
| 93.  | <u>Expenses</u>   | 51 |
| 94.  | Additional remuneration   | 51 |
| 95.  | Remuneration of executive directors   | 51 |
| 96.  | Pensions and other benefits   | 51 |
| 97.  | Powers of the board   | 52 |
| 98.  | Powers of directors if less than minimum number   | 52 |
| 99.  | Powers of executive directors   | 52 |
| 100. | <u>Delegation to committees</u>   | 52 |
| 101. | <u>Local management</u>   | 52 |

| 102. | <u>Power of attorney</u>                                   | 53 |
|------|--|----|
| 103. | Exercise of voting power                                   | 53 |
| 104. | <u>Provision for employees on cessation of business</u>    | 53 |
| 105. | <u>Overseas registers</u>                                  | 53 |
| 106. | Borrowing powers   | 53 |
| 107. | Board meetings   | 54 |
| 108. | Notice of board meetings                                   | 54 |
| 109. | Quorum   | 54 |
| 110. | <u>Chair</u>   | 54 |
| 111. | <u>Voting</u>  | 55 |
| 112. | Participation by telephone or other form of communication  | 55 |
| 113. | Resolution in writing                                      | 55 |
| 114. | <u>Proceedings of committees</u>                           | 55 |
| 115. | Minutes of proceedings                                     | 55 |
| 116. | Validity of proceedings                                    | 55 |
| 117. | <u>Transactions or other arrangements with the company</u> | 56 |
| 118. | Authorisation of Directors' conflicts of interest          | 56 |
| 119. | <u>Directors' permitted interests</u>                      | 57 |
| 120. | General General  | 58 |
| 121. | Power to authenticate documents                            | 59 |
| 122. | Use of seals   | 59 |
| 123. | Declaration of dividends                                   | 59 |
| 124. | <u>Interim dividends</u>                                   | 59 |
| 125. | <u>Calculation and currency of dividends</u>               | 59 |
| 126. | Amounts due on shares can be deducted from dividends       | 60 |
| 127. | Dividends not in cash                                      | 60 |
| 128. | No interest on dividends                                   | 60 |
| 129. | Method of payment  | 60 |
| 130. | <u>Uncashed dividends</u>                                  | 61 |
| 131. | Unclaimed dividends  | 61 |
| 132. | Scrip dividends  | 61 |
| 133. | Capitalisation of reserves                                 | 63 |
| 134. | Record dates   | 64 |
| 135. | <u>Inspection of records</u>                               | 64 |
| 136. | Account to be sent to members                              | 64 |
| 137. | Service of Notices   | 64 |
| 138. | Notice on person entitled by transmission                  | 65 |
| 139. | Record date for service                                    | 66 |
| 140. | Evidence of service  | 66 |
| 141. | Notice when post not available                             | 66 |
| 142. | Indemnity and insurance                                    | 67 |
| 143. | Forum Selection  | 67 |
| 144. | Winding up   | 68 |
|      | <del></del>  | -  |

THE COMPANIES ACT 2006

PUBLIC COMPANY LIMITED BY SHARES

ARTICLES OF ASSOCIATION

**OF** 

#### POLESTAR AUTOMOTIVE HOLDING UK PLC

(Adopted by special resolution passed on the [●] day of [●], 2021 and effective on such date)

#### 1. Exclusion of model articles (and any other prescribed regulations)

No regulations or articles set out in any statute, or in any statutory instrument or other subordinate legislation made under any statute, concerning companies (including the regulations in the Companies (Model Articles) Regulations 2008 (SI 2008/3229)) shall apply as the articles of the Company. The following shall be the articles of association of the Company.

#### 2. Interpretation

2.1 In these articles, unless the context otherwise requires:

Act means the Companies Act 2006.

address includes any number or address used for the purposes of sending or receiving documents or information by electronic means.

**Articles** means these articles of association as altered from time to time and "Article" shall be construed accordingly.

**Board** means the board of Directors for the time being of the Company or the Directors present or deemed to be present at a duly convened quorate meeting of the Directors.

**Business Combination** means the merger of a newly formed, wholly-owned subsidiary of the Company with and into the SPAC, which SPAC shall survive, pursuant to that certain business combination agreement dated September 27 2021

**Business Day** means a day other than a day on which banking institutions or trust companies are authorized or obligated by law to close in Hong Kong, the People's Republic of China, the British Virgin Islands, Sweden, New York City, or London, England, a Saturday or a Sunday.

**call** includes any instalment of a call and, in the application of provisions of these Articles to forfeiture of shares, a sum which, by the terms of issue of a share, is payable at a fixed time.

capital means the share capital from time to time of the Company.

**Chair** means the Chair presiding at any meeting of members or the Board.

**Class A Ordinary Share** means an ordinary share of USD 0.01 each in the share capital of the Company which is designated as a Class A Ordinary Share and having the rights provided for in these Articles.

**Class B Ordinary Share** means an ordinary share of USD 0.01 each in the share capital of the Company which is designated as a Class B Ordinary Share and having the rights provided for in these Articles.

Class C Shares means Class C-1 Shares and Class C-2 Shares.

Class C Share Conversion Ratio means one, subject to adjustment in accordance with these Articles;

**Class C-1 Share** means a class C-1 ordinary share of USD 0.10 each in the share capital of the Company which is designated as a Class C-1 Share and having the rights provided for in these Articles.

**Class C-2 Share** means a class C-2 ordinary share of USD 0.10 each in the share capital of the Company which is designated as a Class C-2 Share and having the rights provided for in these Articles.

**clear days** in relation to a period of notice means that period excluding the day when the notice is served or deemed to be served and the day for which it is given or on which it is to take effect.

**Closing** means closing of the Business Combination Agreement.

**Companies Acts** means the Act, the Companies Act 1985 and, where the context requires, every other statute from time to time in force concerning companies and affecting the Company.

Company means POLESTAR AUTOMOTIVE HOLDING UK PLC.

**Conversion Subscription Value** means an amount equal to the product of the number of Convertible Preferred Shares which have at the relevant time been converted in accordance with these Articles and the initial subscription price (including any premium) paid per each Convertible Preferred Share.

**Convertible Preferred Shares**: means the convertible preferred shares of USD 10.00 each in the capital of the Company having the rights and restrictions set out in these Articles.

Date of Adoption means the date of adoption of these Articles.

**Deferred Shares** means deferred shares of USD 0.01 each in the capital of the Company having the rights and restrictions set out in these Articles.

**Depositary** means any depositary, custodian or nominee approved by the Board that holds legal title to shares in the capital of the Company for the purposes of facilitating beneficial ownership of such shares by other persons;

**Director** means a director for the time being of the Company.

**electronic communication** means a communication sent, transmitted, conveyed and received by wire, by radio, by optical means, by electronic means or by other electromagnetic means in any form through any medium.

**electronic facility** includes, without limitation, website addresses and conference call systems, and any device, system, procedure, method or other facility whatsoever providing an electronic means of attendance at or participation in (or both attendance at and participation in) a general meeting determined by the Board pursuant to Article 52.

**electronic form** has the meaning given to it in section 1168 of the Act.

**electronic means** has the meaning given to it in section 1168 of the Act.

**electronic meeting** means a general meeting held and conducted wholly and exclusively by virtual attendance and participation by members and/or proxies by means of electronic facilities.

**Exchange** means any recognised investment exchange or other stock market or public securities exchange showing quotations for the Company's ordinary shares.

Exchange Act means Securities Exchange Act of 1934 (USA), as amended;

Geely means Zhejiang Geely Holding Co Ltd.

**Group** means the Company and its subsidiaries.

**GBP Redeemable Preferred Shares** means redeemable preference shares of GBP 1.00 each in the capital of the Company having the rights and restrictions set out in these Articles.

**hybrid meeting** means a general meeting held and conducted by (i) physical attendance and participation by members and/or proxies at one or more places of meeting or meeting locations; and (ii) virtual attendance and participation by members and/or proxies by means of electronic facilities.

**Independent Directors** means a director that both (i) satisfies the requirements to qualify as an "independent director" under the stock exchange rules of the stock exchange on which the Ordinary Shares

are then-currently listed and (ii) is not affiliated (as a director, employee, shareholder or otherwise) with Parent, Volvo or Geely, provided that an individual shall not be precluded from being appointed, or continuing to act, as an Independent Director solely on the basis of holding, directly or indirectly, up to 0.01% of the share capital of any publically traded affiliate of Parent, Volvo or Geely.

Initial Liquidation Preference means, in respect of the Convertible Preferred Shares, an aggregate amount equal to USD 498,039,000.

member means a member of the Company, or where the context requires, a member of the Board or of any committee.

month means calendar month.

Office means the registered office from time to time of the Company.

Ordinary Shares means the Class A Ordinary Shares and the Class B Ordinary Shares.

paid up means paid up or credited as paid up.

Parent means Polestar Automotive Holding Limited, a Hong Kong incorporated company.

#### Permitted Transfer means:

- (a) to the Company's officers or directors, any affiliates or family members of any of the Company's officers or directors, any members of the Sponsor, or any affiliates of the Sponsor;
- (b) in the case of an individual, by gift to a member of one of the members of the individual's immediate family or to a trust, the beneficiary of which is a member of one of the individual's immediate family, an affiliate of such person or to a charitable organization;
- (c) in the case of an individual, by virtue of the laws of descent and distribution upon death of the individual;
- (d) in the case of an individual, pursuant to a qualified domestic relations order;
- (e) by private sales or transfers made in connection with the consummation of the Business Combination at prices no greater than the price at which the Class C-2 Shares were originally purchased;
- (f) by virtue of the laws of Delaware or the Sponsor's limited liability company agreement upon dissolution of the Sponsor; or
- (g) in the event of the Company's liquidation, merger, capital stock exchange, reorganization or other similar transaction which results in all of the Company's shareholders having the right to exchange their ordinary shares for cash, securities or other property subsequent to the Closing,

(each such transferee being a "Permitted Transferee").

**physical meeting** means a general meeting held and conducted by physical attendance and participation by members and/or proxies at one or more Meeting Location(s).

**Pound** and £ means pounds in the lawful currency of England.

Preference Conversion Ratio means 1 Class A Ordinary Share for each Convertible Share as adjusted in accordance with Article 9.12.

**Register** means the register of members of the Company to be maintained under the Act or as the case may be any overseas branch register maintained under Article 105.

Seal means the common seal of the Company or, where the context allows, any official seal kept by the Company under section 50 of the Act.

Secretary means the person or persons appointed for the time being to perform for the Company the duties of a secretary.

Securities Act means the Securities Act of 1933 (USA), as amended;

**share** means a Class A Ordinary Share, a Class B Ordinary Share, Class C-1 Share, a Class C-2 Share, a Convertible Preferred Share, a GBP Redeemable Preferred Share, a Deferred Share or any other share in the capital of the Company from time to time.

shareholder or member or holder means a duly registered holder of share(s) from time to time.

**Solvent Reorganisation** means any solvent reorganisation of the Company, including without limitation by merger, consolidation, recapitalisation, transfer or sale of all or substantially all of the shares or assets of the Company, scheme of arrangement, exchange of securities, conversion of entity, migration of entity, formation of new entity, or any other transaction or group of transactions.

SPAC means Gores Guggenheim, Inc., a Delaware corporation

special resolution shall have the meaning ascribed thereto in section 283 of the Companies Act 2006.

Sponsor means Gores Guggenheim Sponsor LLC.

Subsidiary has the meaning given in section 1159 of the Companies Act 2006.

**United States** means the United States of America, its territories, its possessions and all areas subject to its jurisdiction.

**USD** or \$ means the lawful currency of United States.

Volvo means Volvo Car Corporation.

**in writing** and **written** means the representation or reproduction of words, symbols or other information in a visible form by any method or combination of methods, whether sent or supplied in electronic form or otherwise.

- 2.2 Headings shall not affect the interpretation of these Articles.
- 2.3 Unless the context otherwise requires, a **person** includes a natural person, corporate or unincorporated body (whether or not having separate legal personality) acting, where applicable, by their duly authorized representatives.
- 2.4 Unless the context otherwise requires, words in the singular shall include the plural and in the plural shall include the singular.
- 2.5 A reference to one gender shall include a reference to the other genders.
- 2.6 A reference to a statute or statutory provision is a reference to it as amended, extended or re-enacted from time to time, and shall include all subordinate legislation made from time to time under that statute or supporting provision.
- Any words or expressions defined in the Companies Acts in force when these Articles or any part of these Articles are adopted shall (if not inconsistent with the subject or context in which they appear) have the same meaning in these Articles or that part, save that **company** shall include any company, corporation or other body corporate, wherever and however incorporated or established.
- 2.8 A reference to a document **being signed** or to **signature** includes references to its being executed under hand or under seal or by any other method and, in the case of a communication in electronic form, such references are to its being authenticated as specified by the Companies Acts.
- 2.9 A reference to **writing** or **written** includes references to any method of representing or reproducing words in a legible and non-transitory form whether sent or supplied in electronic form or otherwise.
- A references to a notice or document (including, without limitation, a resolution in writing or minutes) include a notice or document recorded or stored in any digital, electronic, electrical, magnetic or other retrievable form or medium and information in visible form whether having physical substance or not.

2.11 A reference to documents or information **being sent or supplied by or to** a company (including the Company) shall be construed in accordance with section 1148(3) of the Act.

### 2.12 A reference to a **meeting:**

- (a) shall mean a meeting convened and held in any manner permitted by these Articles, including a general meeting at which some (but not all) those entitled to be present attend and participate by means of electronic facility or facilities, and such persons shall be deemed to be present at that meeting for all purposes of the Act and these Articles, and **attend**, **participate**, **attending**, **participating**, **attendance** and **participation** shall be construed accordingly; and
- (b) shall not be taken as requiring more than one person to be present if any quorum requirement can be satisfied by one person.
- 2.13 References to a person's **participation** in the business of a general meeting include without limitation and as relevant the right (including, in the case of a corporation, through a duly appointed representative) to speak, vote, be represented by a proxy and have access in hard copy or electronic form to all documents which are required by the Companies Acts or these Articles to be made available at the meeting, and **participate** and **participating** in the business of a general meeting shall be construed accordingly.
- 2.14 Nothing in these Articles precludes the holding and conducting of a general meeting in such a way that persons who are not present together at the same place or places may by electronic means attend and participate in it.

#### 3. Form of resolution

Subject to the Companies Acts, where anything can be done by passing an ordinary resolution, this can also be done by passing a special resolution.

### 4. Limited liability

The liability of the members of the Company is limited to the amount, if any, unpaid on the shares in the Company held by them.

#### 5. Change of name

The name of the Company is "POLESTAR AUTOMOTIVE HOLDING UK PLC". The Company may change its name by resolution of the Board.

### 6. The Office

The Office shall be at such place in England as the Directors shall from time to time appoint.

### 7. Shares

- 7.1 The share capital of the Company on the Date of Adoption is divided into Class A Ordinary Shares, Class B Ordinary Shares, Class C-1 Shares, Class C-2 Shares, Convertible Preferred Shares, Deferred Shares and GBP Redeemable Preferred Shares.
- 7.2 The Board may issue warrants to subscribe for any class of fully paid shares or securities of the Company on such terms and conditions as the Board may from time to time determine. No fraction of any share shall be allotted on exercise of the subscription rights. Subject to the conditions on which the warrants are issued from time to time, bearers of share warrants have the same rights and privileges as they would if their names had been included in the register as holders of the shares represented by their warrants. The Company must not in any way be bound by or recognise any interest in a share represented by a share warrant other than the absolute right of the bearer of that warrant to that warrant.

- 7.3 The Company may in connection with the issue of any shares exercise all powers of paying interest out of capital and of paying commission and brokerage conferred or permitted by the Companies Act.
- 7.4 No person shall become a member until his name shall have been entered into the Register.

### 8. Right to Return of Assets and Distributions

- 8.1 On a return of assets on liquidation or otherwise, the assets of the Company remaining after payment of its debts and liabilities and available for distribution to holders of shares shall be applied in the following manner and order of priority:
  - (a) first, to the holders of the Convertible Preferred Shares (pro rata and pari passu) an amount equal to the Initial Liquidation Preference less such amount as is equal to the Conversion Subscription Value;
  - (b) second, to the holders of the GBP Redeemable Preferred Shares an amount equal to the nominal value of such shares;
  - (c) third, to the holders of the Ordinary Shares pro rata to the number of Ordinary Shares respectively held by them up to an amount of USD 1 million per Ordinary Share;
  - (d) fourth, to the holders of Deferred Shares an amount equal to the nominal value of the Deferred Shares; and
  - (e) fifth:
    - (i) to the holders of the Ordinary Shares pro rata to the number of Ordinary Shares respectively held by them;
    - (ii) to the holders of the Class C Shares pari passu with Ordinary Shares on an as-converted basis *less* the Conversion Price (as defined in Article 13.6 below) per Class C Share,

provided that if the amount which would be received by the holders of the Convertible Preferred Shares if all such shares had been converted in accordance with these Articles would be greater than pursuant to (a) above, the relevant Convertible Preferred Shares shall be deemed for the purposes of the relevant return of capital to be treated pari passu with the holders of Ordinary Shares on an as-converted basis.

- 8.2 Subject to Article 8.3, any dividends (or other distribution) paid by the Company shall be applied among the holders of shares, other than the Class C Shares, the GBP Redeemable Preferred Shares and the Deferred Shares, pro rata to the number of such shares respectively held by them. For the avoidance of doubt, the Class C Shares, the GBP Redeemable Preferred Shares and the Deferred Shares shall not entitle their holders to participate in any dividends or other distributions.
- 8.3 The Convertible Preferred Shares shall not entitle any holder to preferred dividends or accruals save that the holders of Convertible Preferred Shares shall participate in dividends or other distributions on the Class A Ordinary Shares as if such Convertible Preferred Shares had been converted into Class A Ordinary Shares in accordance with these Articles.

### 9. Rights and Conversion of Convertible Preferred Shares

- 9.1 The Convertible Preferred Shares shall carry no voting rights, and shall not entitle their holders to receive notice of, to attend, to speak or to vote at any general meeting of the Company.
- 9.2 On a return of assets on liquidation or otherwise, the assets of the Company remaining after payment of its debts and liabilities and available for distribution to holders of Convertible Preferred Shares shall be applied in the manner and order of priority as set out in Article 8.1.
- 9.3 Any dividends shall be paid to the holders of Convertible Preferred Shares in the manner and order of priority as set out in Articles 8.2 and 8.3.

- 9.4 On an issuance of shares or other securities in connection with a demerger (a "Demerger Issuance"), the holders of Convertible Preferred Shares shall be entitled to participate in such Demerger Issuance as if such Convertible Preferred Shares had been converted into Class A Ordinary Shares in accordance with these Articles.
- 9.5 On a Solvent Reorganisation, the rights of the holders of Convertible Preferred Shares under these Articles shall be preserved in all material respects and the Company shall take all reasonable steps to ensure that the holders of Convertible Preferred Shares shall have equivalent rights in all material respects in respect of any securities and property receivable upon any such Solvent Reorganisation as if such Convertible Preferred Shares had been converted into Class A Ordinary Shares in accordance with these Articles.
- 9.6 Immediately following completion of the Business Combination, each Convertible Preferred Share shall convert into one (1) Class A Ordinary Share (credited as fully paid), provided that:
  - (a) the maximum number of Class A Ordinary Shares to be issued on conversion shall be the maximum number that can be issued so that Volvo (alone or taken together with all other legal entities that, directly or indirectly, are controlled by Geely ("Geely Group")) after such conversion holds, whether directly or indirectly through depositary shares and/or receipts, less than 50% of the aggregate voting rights attaching to the Ordinary Shares (the "Ownership Condition"); or
  - (b) no conversion of a Convertible Preferred Share shall occur in circumstances which would give rise to an obligation on Volvo or any member of the Geely Group to make a mandatory offer under any applicable law or regulation to acquire all of the Class A Ordinary Shares not already held by Volvo or the Geely Group, save with the prior written prior written consent of Volvo or a member of the Geely Group (the "Takeover Condition").
- 9.7 Any Convertible Preferred Shares which are not converted in accordance with Article 9.6 as a result of the Ownership Condition, shall automatically convert into Class A Ordinary Shares at the Preference Conversion Ratio on the last Business Day of each fiscal quarter and on such other dates as determined by the Company to the extent that such conversion can at any time be implemented in compliance with the Ownership Condition and the Takeover Condition.
- 9.8 If a Convertible Preferred Share would otherwise automatically be converted into Class A Ordinary Shares save for the fact that the Takeover Condition is not satisfied, the Company shall take all reasonable steps to procure a waiver prior to any such conversion from the relevant regulatory authority from any obligation on Volvo or any member of the Geely Group to make a mandatory offer following any such conversion. Upon receipt of the requisite waiver, a Convertible Preferred Share shall automatically convert in accordance with Article 9.7. For the avoidance of doubt, where no such waiver is obtained, no such conversion shall take place.
- 9.9 If any Convertible Preferred Share remains unconverted on or following the fifth anniversary of issuance of the Convertible Preferred Shares, a holder of a Convertible Preferred Share shall be entitled, by written notice to the Company, to require that the Convertible Preferred Shares held by it be converted into Class A Ordinary Shares and, subject, to the extent applicable, to the delivery to the Company of a duly stamped instrument of transfer in accordance with these Articles, issued (subject to compliance with the Companies Acts) or transferred to such third party investors as the relevant holder of Convertible Preferred Shares shall direct, provided that such conversion shall not breach the Ownership Condition or the Takeover Condition.
- 9.10 The Company shall, on the relevant conversion date, enter the holder of the converted Convertible Preferred Shares on the register of members as the holder of the appropriate number of Class A Ordinary Shares, subject to the relevant holder delivering its certificate(s) (or an indemnity for lost certificate in a form acceptable to the Board) in respect of any relevant Convertible Preferred Shares so converted in accordance with this Article provided that a failure by such holder to do so shall not prejudice or delay the conversion of the relevant Convertible Preferred Shares into Class A Ordinary Shares. Each conversion of Convertible Preferred Shares shall be deemed to have occurred at 5:00 pm, New York time, on the

applicable conversion date. The Company shall within 10 Business Days of the conversion date forward to such holder by post to his address shown in the register of members (or such other address as the holder may specify by notice to the Company), free of charge, a definitive certificate for the appropriate number of fully paid Class A Ordinary Shares.

- 9.11 The conversion of Convertible Preferred Shares into Class A Ordinary Shares by a holder shall automatically be effected (without the need for any resolution of shareholders or action on the part of the Board) by:
  - (a) the consolidation of the Convertible Preferred Shares to be converted and held by that holder into a single unclassified share; and
  - (b) the subdivision into, and redesignation of, such share into the relevant number of Class A Ordinary Shares (rounded down to the nearest whole number) with the balance being subdivided and converted into Deferred Shares.
- 9.12 The Preference Conversion Ratio shall from time to time be adjusted in accordance with the provisions of this Article:
  - (a) if Convertible Preferred Shares remain capable of being converted into new Class A Ordinary Shares and there is a consolidation and/or sub-division of and/or other alteration of the nominal value of the Class A Ordinary Shares;
  - (b) if Convertible Preferred Shares remain capable of being converted into Class A Ordinary Shares, on an allotment of fully-paid Class A Ordinary Shares pursuant to a capitalisation of profits or reserves (including any share premium account or capital redemption reserve) to holders of Class A Ordinary Shares;
  - (c) if Convertible Preferred Shares remain capable of being converted into Class A Ordinary Shares, on an offer of Class A Ordinary Shares or other share-related securities at a discount by the Company by way of rights or otherwise to all or substantially all of the holders of the Class A Ordinary Shares (a "Discounted Rights Offer").

(each of (a), (b) and (c) a "Convertible Preferred Adjustment Event") the Preference Conversion Ratio shall be adjusted conditional on the Convertible Preferred Adjustment Event occurring such that, in the case of (a) and (b) following adjustment the number of Class A Ordinary Shares into which the outstanding Convertible Preferred Shares will convert is the same as the number of Class A Ordinary Shares as would have resulted had the outstanding Convertible Preferred Shares been converted into Class A Ordinary Shares immediately prior to the time of the Convertible Preferred Adjustment Event at the then prevailing Preference Conversion Ratio and had then been subject to the Convertible Preferred Adjustment Event. In the case of (c), the Preference Conversion Ratio shall be adjusted by multiplying the Preference Conversion Ratio in force immediately prior to such issue or grant by the following fraction:

 $\frac{A+C}{A+B}$ 

where A is the number of Class A Ordinary Shares in issue immediately before announcement of the Discounted Rights Offer: B is the number of Class A Ordinary Shares which the aggregate consideration (if any and before deduction for any commission or expenses paid or incurred by the Company in connection with such issuance) receivable for the Class A Ordinary Shares issued by way of rights, or for the options or warrants or other rights issued by way of rights and for the total number of Class A Ordinary Shares deliverable on the exercise thereof, would purchase at the Fair Market Value per Class A Ordinary Share on the record date of the issuance; C is the number of Class A Ordinary Shares issued or, as the case may be, the maximum number of Class A Ordinary Shares which may be issued upon exercise of such options, warrants or rights calculated as at the date of issue of such options, warrants or rights; and "Fair Market Value" means the volume weighted average price of the Class A Ordinary Shares as reported during the ten (10) trading day period ending on the trading day prior to the date of announcement of the

- Discounted Rights Offer. Any adjustment pursuant to Article 9.12(c) shall only apply to the extent a holder of Convertible Preferred Shares cannot participate in the Discounted Rights Offer due to the Ownership Condition.
- 9.13 If a doubt or dispute arises concerning the determination of an adjustment of the conversion ratio in accordance with Article 9.12, the Board shall refer the matter to the auditors for determination who shall make available to all Shareholders their report and whose certificate as to the amount of the adjustment is, in the absence of manifest error, conclusive and binding on all concerned and their costs shall be met by the Company.

### 10. Rights and Redemption of GBP Redeemable Preferred Shares

- 10.1 The GBP Redeemable Preferred Shares shall carry no voting rights, and shall not entitle their holders to receive notice of, to attend, to speak or to vote at any general meeting of the Company.
- 10.2 On a return of assets on liquidation or otherwise, the assets of the Company remaining after payment of its debts and liabilities and available for distribution to holders of GBP Redeemable Preferred Shares shall be applied in the manner and order of priority as set out in Article 8.1.
- 10.3 Subject to the provisions of the Companies Acts, the Company shall be entitled, at any time, to serve notice on all or some of the holders of the GBP Redeemable Preferred Shares that it wishes to redeem all or some of the GBP Redeemable Preferred Shares in issue at that time on the date falling 14 days after service of such notice (or on such other date as may be agreed between the Company and the holders of the relevant GBP Redeemable Preferred Shares). Notice given under this Article 10.3 shall be irrevocable.
- 10.4 Upon the date on which any GBP Redeemable Preferred Shares are to be redeemed under this Article, each holder of the relevant GBP Redeemable Preferred Shares shall be bound to deliver to the Company at the Office the certificates for such of the GBP Redeemable Preferred Shares as are held by it. Upon such delivery, the Company shall redeem the shares and shall pay to (or to the order of) such holder the nominal value of the relevant GBP Redeemable Preferred Shares. If any holder of GBP Redeemable Preferred Shares whose shares are liable to be redeemed under this paragraph shall fail or refuse to deliver up the certificate for its GBP Redeemable Preferred Shares, the Company may retain the redemption moneys until delivery up of the certificate or an indemnity in respect thereof satisfactory to the Company and shall within seven days thereafter pay the redemption moneys to (or to the order of) such holder. The receipt of the registered holder for the time being of any GBP Redeemable Preferred Shares or, in the case of registered holders, the receipt of any of them for the moneys payable on redemption thereof shall constitute an absolute discharge in respect thereof.

#### 11. Rights of Deferred Shares

- 11.1 The Deferred Shares shall carry no voting rights, and shall not entitle their holders to receive notice of, to attend, to speak or to vote at any general meeting of the Company. Unless the Board determines otherwise, a holder of Deferred Shares shall not be entitled to receive a certificate for such shares.
- 11.2 On a return of assets on liquidation or otherwise, the assets of the Company remaining after payment of its debts and liabilities and available for distribution to holders of Deferred Shares shall be applied in the manner and order of priority as set out in Article 8.1.
- 11.3 The special resolution of the Company adopting these Articles shall be deemed to confer irrevocable authority on the Company, at any time and from time to time, to do all or any of the following without obtaining the sanction and/or consent of the holder(s) of the Deferred Shares (or any of them):
  - (a) from the date falling two years after the Date of Adoption, to appoint any person to execute on behalf of any holder of Deferred Shares a transfer of all or any of its/his Deferred Shares and/or an agreement to transfer the same to such person(s) as the Company may determine and without making any payment to the holder thereof; and

(b)

- (i) from the date falling two years after the Date of Adoption, to purchase and/or cancel any Deferred Shares (in accordance with the Companies Acts) without obtaining the consent of the holder(s) of those Shares and without making any payment to the holder(s) thereof; and
- (ii) for the purposes of any such purchase and/or cancellation, to appoint any person to execute a contract for the sale of any such Shares to the Company on behalf of any holder of Deferred Shares,

and, pending such transfer, purchase and/or cancellation, to retain the certificate(s) (if any) for such Shares.

### 12. Rights and Conversion of Ordinary Shares

- 12.1 On a return of assets on liquidation or otherwise, the assets of the Company remaining after payment of its debts and liabilities and available for distribution to holders of Ordinary Shares shall be applied in the manner and order of priority as set out in Article 8.1.
- 12.2 Any dividend shall be paid to the holders of Ordinary Shares in the manner and order of priority as set out in Article 8.2.
- 12.3 Ordinary Shares shall confer on each holder of Ordinary Shares the right to receive notice of and to attend, speak and vote at all general meetings of the Company in accordance with these Articles.
- 12.4 Each Class B Ordinary Share is convertible into one (1) Class A Ordinary Share at any time at the option of the holder of such Class B Ordinary Share. The right to convert such Class B Ordinary Shares into Class A Ordinary Shares shall be exercisable by the holder of the Class B Ordinary Share delivering a written notice to the Company that such holder elects to convert a specified number of Class B Ordinary Shares into Class A Ordinary Shares. In no event shall Class A Ordinary Shares be convertible into Class B Ordinary Shares.
- 12.5 Any conversion of Class B Ordinary Shares into Class A Ordinary Shares pursuant to these Articles shall be effected by means of the re-designation of each relevant Class B Ordinary Share as a Class A Ordinary Share or by such other method as may be approved by the Board. Such conversion shall become effective forthwith upon entries being made in the Register to record the re-designation of the relevant Class B Ordinary Shares as Class A Ordinary Shares. The Company shall complete the conversion as soon as reasonably practicable, and in any event within 10 Business Days, after the conversion notice is delivered by the holder of the Class B Ordinary Share to the Company.
- 12.6 Save and except for voting rights as set out in Article 70.1 and conversion rights as set out in Articles 12.4 and 12.5, Class B Ordinary Shares and Class A Ordinary Shares shall rank pari passu with one another and shall have the same rights, preferences, privileges and restrictions.

### 13. Rights and Conversion of Class C Shares

Return of capital

13.1 On a return of assets on liquidation or otherwise, the assets of the Company remaining after payment of its debts and liabilities and available for distribution to holders of Class C Shares shall be applied in the manner and order of priority as set out in Article 8.1.

Voting

13.2 Class C Shares shall confer on each holder of Class C Shares the right to receive notice of and to attend, speak and vote at all general meetings of the Company in accordance with these Articles.

### Return of capital

- 13.3 Class C-1 Shares and Class C-2 Shares shall have the same rights, preferences, privileges and restrictions, except that so long as Class C-2 Shares are held by the Sponsor or any of its Permitted Transferees, the Class C-2 Shares:
  - (a) may be converted for cash in accordance with Article 13.6 or on a cashless basis in accordance with Article 13.7;
  - (b) may not be transferred, assigned or sold until thirty (30) days after Closing; and
  - (c) shall not be redeemable by the Company pursuant to Article 13.25,

provided, however, that in the case of (b), the Class C-2 Shares and any Class A Ordinary Shares held by the Sponsor or any of its Permitted Transferees arising upon the conversion of the Class C-2 Shares may be transferred by the holders to a Permitted Transferee.

Conversion of Class C-2 Shares into Class C-1 Shares

- 13.4 Each Class C-2 Share is convertible into one (1) Class C-1 Share at any time at the option of the holder of such Class C-2 Share. The right to convert such Class C-2 Shares into Class C-1 Shares shall be exercisable by the holder of the Class C-2 Share delivering a written notice to the Company that such holder elects to convert a specified number of Class C-2 Shares into Class C-1 Shares. In no event shall Class C-1 Shares be convertible into Class C-2 Shares.
- 13.5 Any conversion of Class C-2 Shares into Class C-1 Shares pursuant to these Articles shall be effected by means of the re-designation of each relevant Class C-2 Share as a Class C-1 Share or by such other method as may be approved by the Board. Such conversion shall become effective forthwith upon entries being made in the Register to record the re-designation of the relevant Class C-2 Shares as Class C-1 Shares. The Company shall complete the conversion as soon as reasonably practicable, and in any event within 10 Business Days, after the conversion notice is delivered by the holder of the Class C-2 Share to the Company.

Conversion for cash

During the Conversion Period, each holder of Class C Shares shall be entitled by notice to the Company (a "Conversion Notice")¹ to require the conversion into Class A Ordinary Shares upon payment to the Company of USD 11.50 per Class A Ordinary Share (subject to the adjustments set out in Articles 13.20 to 13.24 (inclusive) and the last sentence of this Article 13.6) ("Conversion Price") of any or all of the relevant Class C Shares held by them at any time, and those Class C Shares shall convert automatically and with immediate effect on the later of (a) delivery of such notice and (b) payment in full of the aggregate Conversion Price and any and all applicable taxes due in connection with the conversion of the Class C Shares into the Class A Ordinary Shares in lawful money of the United States, in good certified check or good bank draft payable to the Company. If the relevant Conversion Notice does not specify the time on the Conversion Date that the relevant conversion shall occur, the conversion shall take place immediately upon the occurrence of the Conversion Date. The Company in its sole discretion may lower the Conversion Price at any time prior to the Expiration Date (as defined in Article 13.15) for a period of not less than twenty (20) Business Days, provided, that the Company shall provide at least three (3) Business Days prior written notice of such reduction to the Class C Share holders and, provided further that any such reduction shall be identical among all of the Class C Shares. The number of Class A Ordinary Shares into which Class C Shares will convert is equal to the number of Class C Shares being converted multiplied by the Conversion Ratio.

<sup>1</sup> **Note to Draft:** Form of conversion notice to be scheduled.

#### Cashless conversion of Class C-2 Shares

- 13.7 Without prejudice to Article 13.6, during the Conversion Period, each holder of Class C-2 Shares which is the Sponsor or a Permitted Transferee of the Sponsor shall be entitled by notice to the Company (a "Cashless Conversion Notice") to require the conversion into Class A Ordinary Shares of any or all of the relevant Class C-2 Shares held by them at any time on a cashless basis, and those Class C Shares shall convert automatically and with immediate effect on delivery of such Cashless Conversion Notice (the "Conversion Date"). The number of Class A Ordinary Shares into which Class C-2 Shares will convert is equal to the quotient obtained by dividing:
  - (i) the number of converting Class C-2 Shares multiplied by the Conversion Ratio and multiplied by the difference between the Conversion Price and the "Fair Market Value"; by
  - (ii) the "Fair Market Value".

Solely for purposes of this Article 13.7, the "Fair Market Value" shall mean the average last sale price of the Class A Ordinary Share for the ten (10) trading days ending on the third trading day prior to the date on which the Cashless Conversion Notice is sent to the Company.

### Conversion mechanism

- 13.8 The conversion of Class C Shares into Class A Ordinary Shares by a holder pursuant to any provision of these Articles shall automatically be effected (without the need for any resolution of shareholders or action on the part of the Board) by:
  - (a) the consolidation of the Class C Shares to be converted and held by that holder into a single unclassified share; and
  - (b) the subdivision into, and re-designation of, such share into the relevant number of Class A Ordinary Shares (rounded down to the nearest whole number) with the balance being subdivided and converted into Deferred Shares.

### Cashless conversion of Class C Shares in certain circumstances

- 13.9 The Company agrees that as soon as practicable, but in no event later than fifteen (15) Business Days after the Closing, it shall use its commercially reasonable efforts to file with the Securities and Exchange Commission a registration statement for the registration, under the Securities Act, of the Class A Ordinary Shares arising on conversion of the Class C Shares. The Company shall use its best efforts to cause the same to become effective and to maintain the effectiveness of such registration statement, and a current prospectus relating thereto for so long as any Class C Share remain outstanding. If any registration statement for the registration of the Class A Ordinary Shares arising upon the conversion of the Class C Shares pursuant to the Securities Act has not been declared effective by the 60th Business Day following Closing, holders of the Class C Shares shall have the right, during the period beginning on the 61st Business Day after Closing and ending upon such registration statement being declared effective by the Securities and Exchange Commission, and during any other period when the Company shall fail to have maintained an effective registration statement covering the Class A Ordinary Shares arising upon conversion of the Class C Shares, to convert such Class C Shares into Class A Ordinary Shares on a cashless basis. The number of Class A Ordinary Shares into which the relevant Class C Shares will convert is equal to the quotient obtained by dividing:
  - (i) the number of converting Class C Shares, multiplied by the Conversion Ratio and multiplied by the difference between the Conversion Price and the "Fair Market Value"; by
  - (ii) the "Fair Market Value".

Solely for purposes of this Article 13.9, "Fair Market Value" shall mean the volume weighted average price of the Class A Ordinary Shares as reported during the ten (10) trading day period ending on the

trading day prior to the date that notice of conversion is received by the Company from the holder of such Class C Shares. In connection with the conversion of a Class C-1 Share pursuant to this Article 13.9, the Company shall, upon request, provide the holder of Class C-1 Shares with an opinion of counsel for the Company (which shall be an outside law firm satisfactory to the holder of Class C-1 Shares with securities law experience) stating that:

- (iii) the conversion of the Class C-1 Shares pursuant to this Article 13.9 on a cashless basis is not required to be registered under the Securities Act; and
- (iv) the Class A Ordinary Shares arising upon such conversion shall be freely tradable under United States federal securities laws by anyone who is not an affiliate (as such term is defined in Rule 144 under the Securities Act (or any successor rule)) of the Company and, accordingly, shall not be required to bear a restrictive legend.

Except as provided in Article 13.10, for the avoidance of any doubt, unless and until all of the Class C Shares have been converted, the Company shall continue to be obligated to comply with its registration obligations under the first three sentences of this Article 13.9.

- 13.10 If the Class A Ordinary Shares are at the time of any conversion of a Class C Share not listed on a national securities exchange such that it satisfies the definition of a "covered security" under the Securities Act (or any successor rule), the Company may, at its option:
  - (a) require holders of Class C-1 Shares who convert Class C-1 Shares to convert such Class C-1 Shares on a "cashless basis" in accordance with Article 13.9 and the Securities Act (or any successor rule); and
  - (b) in the event the Company so elects, the Company shall:
    - (i) not be required to file or maintain in effect a registration statement for the registration, under the Securities Act, of the Class A Ordinary Shares arising upon conversion of the Class C-1 Shares; and
    - (ii) use its best efforts to register the Class A Ordinary Shares arising upon conversion of the Class C-1 Shares under the blue sky laws of the state of residence of the converting holder of Class C-1 Shares to the extent an exemption is not available.

Issue of share certificates on conversion

13.11 Subject to Article 13.12, as soon as practicable after the conversion of any Class C Share and the clearance of the funds in payment of the Conversion Price (if payment is required pursuant to these Articles), the Company shall issue to the holder of such Class C Share a share certificate for the number of shares of Class A Ordinary Shares to which he, she or it is entitled, and if such Class C Share shall not have been converted in full, a balancing certificate in respect of the Class C Shares that have not been converted.

No conversion unless certain securities laws obligations are satisfied

- 13.12 Notwithstanding Article 13.11 or any other provision of these Articles, the Company shall not be obligated to deliver any Class A Ordinary Shares pursuant to the conversion of Class C Shares and shall have no obligation to settle such Class C Shares conversion unless a registration statement under the Securities Act with respect to the Class A Ordinary Shares arising on conversion of the Class C-1 Shares is then effective and a prospectus relating thereto is current, subject to the Company satisfying its obligations:
  - (a) to, as soon as practicable, but in no event later than fifteen (15) Business Days after Closing, use its commercially reasonable efforts to file with the Securities and Exchange Commission a registration statement for the registration, under the Securities Act, of the Class A Ordinary Shares arising upon conversion of the Class C Shares; and
  - (b) as set out in Article 13.9,

("Registration Condition One"). In addition, no Class C Share shall be convertible unless any Class A Ordinary Share arising upon such Class C Share conversion has been registered, qualified or deemed to be exempt under the securities laws of the state of residence of the holder of the Class C Shares ("Registration Condition Two").

Class A Ordinary Shares to be fully paid

13.13 All Class A Ordinary Shares re-designated upon the proper conversion of Class C Shares in accordance with these Articles shall be credited as fully paid.

Conversion Price reduction

13.14 The Company in its sole discretion may lower the Conversion Price at any time prior to the conversion of the relevant Class C Shares for a period of not less than twenty (20) Business Days, provided, that the Company shall provide at least three (3) Business Days prior written notice of such reduction to the holders of Class C Shares and, provided further that any such reduction shall be identical among all of the Class C Shares.

Conversion Period and expiry

- 13.15 Other than a conversion pursuant to Article 13.26 or 13.27, a Class C Share may be converted only during the period (the "Conversion Period") commencing on the date that is thirty (30) days after Closing, and terminating at 5:00 p.m., New York City time on the earlier to occur of:
  - (a) the date that is five (5) years after Closing;
  - (b) other than with respect to the Class C-2 Shares then held by the Sponsor or any Permitted Transferees with respect to a redemption, the redemption date as fixed by the Company (the "Expiration Date"),

provided, however, that the conversion of any Class C Share shall be subject to the satisfaction of the conditions set out in Article 13.9 with respect to an effective registration statement. Except with respect to the right to receive the Redemption Price (as defined Article) (other than with respect to a Class C-2 Share then held by the Sponsor or any Permitted Transferees with respect to a redemption pursuant to Article 13.25) in the event of a redemption or conversion (as set forth in Articles 13.25 to 13.27), each Class C Share (other than a Class C-2 Share then held by the Sponsor or any Permitted Transferees with respect to a redemption pursuant to Article 13.25) not converted on or before the Expiration Date shall automatically convert into Deferred Shares of the same aggregate nominal value at 5:00 p.m. New York City time on the Expiration Date. The Company in its sole discretion may delay the Expiration Date; provided, that the Company shall provide at least twenty (20) days prior written notice of any such extension to holders of the Class C Shares and, provided further that any such extension shall be identical in duration among all the Class C Shares.

Time to complete conversion

13.16 The Company shall complete any conversion of Class C Shares into Class A Ordinary Shares as soon as reasonably practicable, and in any event within 10 Business Days, after the later of (i) delivery of the relevant conversion notice, if applicable and (ii) in the case of conversion pursuant to Article 13.6, the payment of the Conversion Price by the holder of the Class C Shares to the Company in accordance with these Articles.

Maximum holdings

13.17 A holder of a Class C Share may notify the Company in writing in the event it elects to be subject to the provisions contained in this Article; however, no holder of a Class C Share shall be subject to this Article

unless he, she or it makes such election. If the election is made by a holder, the Company shall not effect the conversion of the holder's Class C Shares into Class A Ordinary Shares, and such holder shall not have the right to convert such Class C Shares into Class A Ordinary Shares, to the extent that after giving effect to such conversion, such person (together with such person's affiliates), to the Company's actual knowledge (without further investigation or inquiry), would beneficially own in excess of 9.8% (the "Maximum Percentage") of the Class A Ordinary Shares outstanding immediately after giving effect to such conversion.

- 13.18 For purposes of Article 13.17, the aggregate number of Class A Ordinary Shares beneficially owned by such person and its affiliates shall include the number of Class A Ordinary Shares arising upon conversion of the Class C Shares with respect to which the determination under the second sentence of Article 13.17 is being made, but shall exclude Class A Ordinary Shares that would arise upon:
  - (a) conversion of the remaining, unconverted portion of the Class C Shares beneficially owned by such person and its affiliates; and
  - (b) exercise or conversion of the unexercised or unconverted portion of any other securities of the Company beneficially owned by such person and its affiliates (including, without limitation, any convertible notes or convertible preferred securities or warrants) subject to a limitation on conversion or exercise analogous to the limitation contained herein.

Except as set forth in this Article 13.18, for purposes of this paragraph, beneficial ownership shall be calculated in accordance with Section 13(d) of the Exchange Act. For purposes of the Class C Shares, in determining the number of outstanding Class A Ordinary Shares, the holder may rely on the number of outstanding Class A Ordinary Shares as reflected in:

- (c) the Company's most recent annual report, quarterly report, current report or other public filing with the Securities and Exchange Commission as the case may be;
- (d) a more recent public announcement by the Company; or
- (e) any other notice by the Company setting forth the number of Class A Ordinary Shares outstanding.

For any reason at any time, upon the written request of the holder of the Class C Shares, the Company shall, within two (2) Business Days, confirm orally and in writing to such holder the number of Class A Ordinary Shares then outstanding. In any case, the number of outstanding Class A Ordinary Shares shall be determined after giving effect to the conversion or exercise of equity securities of the Company by the holder and its affiliates since the date as of which such number of outstanding Class A Ordinary Shares was reported. By written notice to the Company, the holder of a Class C Share may from time to time increase or decrease the Maximum Percentage applicable to such holder to any other percentage specified in such notice; provided, however, that any such increase shall not be effective until the sixty-first (61st) day after such notice is delivered to the Company.

Entry in Register and certificates

13.19 The Company shall, on the Conversion Date, enter the holder of the converted Class C Shares on the Register as the holder of the appropriate number of Class A Ordinary Shares, subject to the relevant holder delivering its certificate(s) (or an indemnity for lost certificate in a form acceptable to the Board) in respect of the relevant Class C Shares so converted in accordance with these Articles, provided that a failure by such holder to do so shall not prejudice or delay the conversion of the relevant Class C Shares into Class A Ordinary Shares. The Company shall within 10 Business Days of the Conversion Date forward to such holder by post to his address shown in the Register (or such other address as the holder may specify by notice to the Company), free of charge, a definitive certificate for the appropriate number of fully paid Class A Ordinary Shares.

Adjustments

13.20

- (a) If the number of outstanding Class A Ordinary Shares is increased by a scrip dividend payable in Class A Ordinary Shares, or by a subdivision of Class A Ordinary Shares or other similar event, then, on the effective date of such scrip dividend, subdivision or similar event, the Conversion Ratio shall be increased in proportion to such increase in the outstanding Class A Ordinary Shares. A rights offering to holders of the Class A Ordinary Shares entitling holders to purchase shares of Class A Ordinary Shares at a price less than the "Fair Market Value" shall be deemed a scrip dividend of a number of Class A Ordinary Shares equal to the product of:
  - (i) the number of Class A Ordinary Shares actually issued in such rights offering (or issuable under any other equity securities issued in such rights offering that are convertible into or exercisable for the Class A Ordinary Shares), multiplied by one (1) minus the quotient of:
    - (A) the price per share of Class A Ordinary Shares paid in such rights offering; divided by
    - (B) the "Fair Market Value".

For purposes of this Article 13.20(a) (i) if the rights offering is for securities convertible into or exercisable for Class A Ordinary Shares, in determining the price payable for Class A Ordinary Shares, there shall be taken into account any consideration received for such rights, as well as any additional amount payable upon exercise or conversion and (ii) "Fair Market Value" means the volume weighted average price of the Class A Ordinary Shares as reported during the ten (10) trading day period ending on the trading day prior to the first date on which the Class A Ordinary Shares trade on the applicable exchange or in the applicable market, regular way, without the right to receive such rights.

- (b) If the Company, at any time while the Class C Shares are on issue, shall pay a dividend or make a distribution in cash, securities or other assets to the holders of Class A Ordinary Shares on account of such Class A Ordinary Shares (or other shares into which the Class C Shares are convertible), other than:
  - (i) as described in Article 13.20(a); or
  - (ii) Ordinary Cash Dividends (as defined below),

then the Conversion Price shall be decreased, effective immediately after the effective date of such Extraordinary Dividend, by the amount of cash and/or the fair market value (as determined by the Board, in good faith) of any securities or other assets paid on each Class A Ordinary Share in respect of such Extraordinary Dividend. For purposes of this Article 13.20(a), "Ordinary Cash Dividends" means any cash dividend or cash distribution which, when combined on a per share basis, with the per share amounts of all other cash dividends and cash distributions paid on the Class A Ordinary Shares during the 365-day period ending on the date of declaration of such dividend or distribution (as adjusted to appropriately reflect any of the events referred to in other subsections of this Article and excluding cash dividends or cash distributions that resulted in an adjustment to the Conversion Price or to the number of shares of Class A Ordinary Shares arising on conversion of each Class C Share) does not exceed \$0.50.

- 13.21 If the number of outstanding Class A Ordinary Shares is decreased by a consolidation, combination, reverse stock split or reclassification of Class A Ordinary Shares or other similar event, then, on the effective date of such consolidation, combination, reverse stock split, reclassification or similar event, the Conversion Ratio shall be decreased in proportion to such decrease in outstanding Class A Ordinary Shares.
- 13.22 Whenever the Conversion Ratio is adjusted, as provided in Articles 13.20(a) and 13.21, the Conversion Price shall be adjusted (to the nearest cent) by multiplying such Conversion Price immediately prior to

such adjustment by a fraction (x) the numerator of which shall be the Conversion Ratio immediately prior to such adjustment, and (y) the Conversion Ratio immediately thereafter.

#### 13.23 In case of any:

- reclassification or reorganization of the outstanding Class A Ordinary Shares (other than a change under Articles 13.20 or 13.21) or (a) that solely affects the nominal value of Class A Ordinary Shares); or
- (b) sale or conveyance to another corporation or entity of the assets or other property of the Company as an entirety or substantially as an entirety in connection with which the Company is dissolved,

the holders of the Class C Shares shall thereafter have the right to purchase and receive, in lieu of the Class A Ordinary Shares into which such Class C Shares would convert pursuant to Article 13.6, the kind and amount of shares of stock or other securities or property (including cash) receivable upon such reclassification, reorganization, merger or consolidation, or upon a dissolution following any such sale or transfer, that the holder of the Class C Shares would have received if such holder had converted his, her or its Class C Share(s) immediately prior to such event (the "Alternative Issuance"), provided that:

- if the holders of the Class A Ordinary Shares were entitled to exercise a right of election as to the kind or amount of securities, cash or other assets receivable upon such consolidation or merger, then the kind and amount of securities, cash or other assets constituting the Alternative Issuance for which each Class C Share shall become convertible shall be deemed to be the weighted average of the kind and amount received per share by the holders of the Class A Ordinary Shares in such consolidation or merger that affirmatively make such election; and
- (d) if a tender, exchange or redemption offer shall have been made to and accepted by the holders of the Class A Ordinary Shares under circumstances in which, upon completion of such tender or exchange offer, the maker, together with members of any group (within the meaning of Rule 13d-5(b)(1) under the Exchange Act (or any successor rule)) of which such maker is a part, and together with any affiliate or associate of such maker (within the meaning of Rule 12b-2 under the Exchange Act (or any successor rule)) and any members of any such group of which any such affiliate or associate is a part, own beneficially (within the meaning of Rule 13d-3 under the Exchange Act (or any successor rule)) more than 50% of the outstanding Class A Ordinary Shares, the holder of a Class C Share shall be entitled to receive as the Alternative Issuance, the highest amount of cash, securities or other property to which such holder would actually have been entitled as a shareholder if such Class C Share holder had converted the Class C Shares prior to the expiration of such tender or exchange offer, accepted such offer and all of the Class A Ordinary Shares held by such holder had been purchased pursuant to such tender or exchange offer, subject to adjustments (from and after the consummation of such tender or exchange offer) as nearly equivalent as possible to the adjustments provided for in Articles 13.20 to 13.22 (inclusive) and this Article; provided, further, that if less than 70% of the consideration receivable by the holders of the Class A Ordinary Shares in the applicable event is payable in the form of common stock in the successor entity that is listed for trading on a national securities exchange or is quoted in an established over-the-counter market, or is to be so listed for trading or quoted immediately following such event, and if the holder properly converts the Class C Shares within thirty (30) days following the public disclosure of the consummation of such applicable event by the Company pursuant to a current report on Form 8-K filed with the Securities and Exchange Commission, the Conversion Price shall be reduced by an amount (in dollars) equal to the difference of (i) the Conversion Price in effect prior to such reduction minus (ii) (A) the Per Share Consideration (as defined below) (but in no event less than zero) minus [(B) the Class C Share value immediately prior to the consummation of the applicable event]. For purposes of calculating such amount, (1) Articles 13.25 to 13.32 shall be taken into account, (2) the price of each Class A Ordinary Share shall be the volume weighted average price of the Class A Ordinary Share as reported during the ten

(10) trading day period ending on the trading day prior to the

effective date of the applicable event, (3) the assumed volatility shall be the 90 day volatility as of the trading day immediately prior to the day of the announcement of the applicable event, and (4) the assumed risk-free interest rate shall correspond to the U.S. Treasury rate for a period equal to the remaining time the Class C Shares are not converted. "Per Share Consideration" means (i) if the consideration paid to holders of the Class A Ordinary Share consists exclusively of cash, the amount of such cash per share of Class A Ordinary Share, and (ii) in all other cases, the volume weighted average price of the Class A Ordinary Share as reported during the ten (10) trading day period ending on the trading day prior to the effective date of the applicable event.

If any reclassification or reorganization also results in a change in shares of Class A Ordinary Shares covered by Article 13.20(a), then such adjustment shall be made pursuant to Articles 13.20(a), 13.21 or 13.22 and this Article. The provisions of this Article shall similarly apply to successive reclassifications, reorganizations, mergers or consolidations, sales or other transfers. In no event will the Conversion Price be reduced to less than the nominal value per share re-designated upon conversion of the Class C Shares.

13.24 Upon every adjustment of the Conversion Price or the Conversion Ratio, the Company shall give written notice thereof to the holders of Class C Shares, which notice shall state the Conversion Price and Conversion Ratio resulting from such adjustment, setting forth in reasonable detail the method of calculation and the facts upon which such calculation is based. Upon the occurrence of any event specified in Articles 13.20 to 13.23 (inclusive), the Company shall give written notice of the occurrence of such event to each holder of Class C Shares, at the last address set forth for such holder in the register of members of the Company, of the record date or the effective date of the event. Failure to give such notice, or any defect therein, shall not affect the legality or validity of such event.

### Redemption or compulsory conversion

- 13.25 Subject to Articles 13.31 and 13.32, not less than all of the outstanding Class C Shares may be redeemed, at the option of the Company, at any time during the Conversion Period, at the office of the Company, upon notice to the holders of the Class C Shares, as described in Article 13.29, at the price of \$0.01 per Class C Share (the "Redemption Price"), provided that the last sales price of the Class A Ordinary Shares reported has been at least \$18.00 per share [(subject to adjustment in compliance with Articles 13.20 to 13.23 (inclusive))], on each of twenty (20) trading days within the thirty (30) trading-day period ending on the third Business Day prior to the date on which notice of the redemption is given and provided that there is an effective registration statement covering the Class A Ordinary Shares arising upon conversion of the Class C Shares, and a current prospectus relating thereto, available throughout the 30-day Redemption Period (as defined in Article 13.29).
- 13.26 Subject to Articles 13.31 and 13.32, the Company, at any time during the Conversion Period upon notice to the holders of the Class C Shares, as described in Article 13.29, may require the holders of Class C Shares to convert all of the outstanding Class C Shares into Class A Ordinary Shares, provided that the last sales price of the Class A Ordinary Shares reported has been at least \$18.00 per share (subject to adjustment in compliance with Articles 13.20 to 13.23 (inclusive)), on each of twenty (20) trading days within the thirty (30) trading-day period ending on the third Business Day prior to the date on which notice of the conversion is given. The number of Ordinary Class A Shares into which the converting Class C Shares will convert is equal to the quotient obtained by dividing:
  - (i) the product of the number of Class C Shares converting, multiplied by the Conversion Ratio, , and multiplied by the difference between the Conversion Price and the "Fair Market Value"; by
  - (ii) the "Fair Market Value".

Solely for purposes of this Article 13.26, the "Fair Market Value" shall mean the average last sale price of the Class A Ordinary Share for the ten (10) trading days ending on the third trading day prior to the date on which the notice of redemption is sent to the holders of Class C Shares;

- (90) days after they are first convertible pursuant to Article 13.6, at the office of the Company, upon notice to the holders of the Class C Shares, as described Article 13.29, into a number of Class A Ordinary Shares determined by reference to the table below, based on the conversion date (calculated for purposes of the table as the period to conversion of the Class C Shares) and the Fair Market Value (as such term is defined in Article 13.26 (the "Alternative Redemption Price"), provided (i) that the last sales price of the Class A Ordinary Shares reported has been at least \$10.00 per share (subject to adjustment in compliance with Articles 13.20 to 13.23 (inclusive)), on the trading day prior to the date on which notice of the redemption is given, (ii) the Class C-2 Shares are converted on the same basis as the outstanding Class C-1 Shares and (iii) there is an effective registration statement covering the Class A Ordinary Shares arising upon conversion of the Class C Shares, and a current prospectus relating thereto, available throughout the 30-day Redemption Period (as defined in Article 13.29) or the Company has elected to require the conversion of the Class C Shares on a "cashless basis" pursuant to Article 13.26. The number of Ordinary Class A Shares into which the converting Class C Shares will convert is equal to the quotient obtained by dividing:
  - (i) the product of the number of Class C Shares converting, multiplied by the difference between the conversion price (taken from the table below) and the "Fair Market Value" (as such term is defined in Article 13.26); by
  - (ii) the "Fair Market Value".

Fair Market Value of Class A Ordinary Shares

| Conversion Date (period to    |                |         | U       |         |         |         |         |         |         |
|-------------------------------|----------------|---------|---------|---------|---------|---------|---------|---------|---------|
| conversion of Class C Shares) | <u>\$10.00</u> | \$11.00 | \$12.00 | \$13.00 | \$14.00 | \$15.00 | \$16.00 | \$17.00 | \$18.00 |
| 57 months                     | 0.257          | 0.277   | 0.294   | 0.31    | 0.324   | 0.337   | 0.348   | 0.358   | 0.365   |
| 54 months                     | 0.252          | 0.272   | 0.291   | 0.307   | 0.322   | 0.335   | 0.347   | 0.357   | 0.365   |
| 51 months                     | 0.246          | 0.268   | 0.287   | 0.304   | 0.32    | 0.333   | 0.346   | 0.357   | 0.365   |
| 48 months                     | 0.241          | 0.263   | 0.283   | 0.301   | 0.317   | 0.332   | 0.344   | 0.356   | 0.365   |
| 45 months                     | 0.235          | 0.258   | 0.279   | 0.298   | 0.315   | 0.33    | 0.343   | 0.356   | 0.365   |
| 42 months                     | 0.228          | 0.252   | 0.274   | 0.294   | 0.312   | 0.328   | 0.342   | 0.355   | 0.364   |
| 39 months                     | 0.221          | 0.246   | 0.269   | 0.29    | 0.309   | 0.325   | 0.34    | 0.354   | 0.364   |
| 36 months                     | 0.213          | 0.239   | 0.263   | 0.285   | 0.305   | 0.323   | 0.339   | 0.353   | 0.364   |
| 33 months                     | 0.205          | 0.232   | 0.257   | 0.28    | 0.301   | 0.32    | 0.337   | 0.352   | 0.364   |
| 30 months                     | 0.196          | 0.224   | 0.25    | 0.274   | 0.297   | 0.316   | 0.335   | 0.351   | 0.364   |
| 27 months                     | 0.185          | 0.214   | 0.242   | 0.268   | 0.291   | 0.313   | 0.332   | 0.35    | 0.364   |
| 24 months                     | 0.173          | 0.204   | 0.233   | 0.26    | 0.285   | 0.308   | 0.329   | 0.348   | 0.364   |
| 21 months                     | 0.161          | 0.193   | 0.223   | 0.252   | 0.279   | 0.304   | 0.326   | 0.347   | 0.364   |
| 18 months                     | 0.146          | 0.179   | 0.211   | 0.242   | 0.271   | 0.298   | 0.322   | 0.345   | 0.363   |
| 15 months                     | 0.13           | 0.164   | 0.197   | 0.23    | 0.262   | 0.291   | 0.317   | 0.342   | 0.363   |
| 12 months                     | 0.111          | 0.146   | 0.181   | 0.216   | 0.25    | 0.282   | 0.312   | 0.339   | 0.363   |
| 9 months                      | 0.09           | 0.125   | 0.162   | 0.199   | 0.237   | 0.272   | 0.305   | 0.336   | 0.362   |
| 6 months                      | 0.065          | 0.099   | 0.137   | 0.178   | 0.219   | 0.259   | 0.296   | 0.331   | 0.362   |
| 3 months                      | 0.034          | 0.065   | 0.104   | 0.15    | 0.197   | 0.243   | 0.286   | 0.326   | 0.361   |
| 0 months                      | _              | _       | 0.042   | 0.115   | 0.179   | 0.233   | 0.281   | 0.323   | 0.361   |

The exact Fair Market Value and Redemption Date (as defined in Article 13.29) may not be set forth in the table above, in which case, if the Fair Market Value is between two values in the table or the Redemption Date is between two redemption dates in the table, the number of Class A Ordinary Shares into which Class C Share will convert will be determined by a straight-line interpolation between the number of shares set forth for the higher and lower Fair Market Values and the earlier and later redemption dates, as applicable, based on a 365- or 366-day year, as applicable. The Class A Ordinary Share conversion prices set out in the table above shall be adjusted as of any date on which the number of shares arising upon conversion of a Class C Share is adjusted pursuant to Articles 13.20 to 13.23 (inclusive)). The adjusted

- Class A Ordinary Share prices in the column headings shall equal the Class A Ordinary Share prices immediately prior to such adjustment, multiplied by a fraction, the numerator of which is the number of shares deliverable upon conversion of a Class C Share immediately prior to such adjustment and the denominator of which is the number of shares deliverable upon conversion of a Class C Share as so adjusted. The number of shares in the table above shall be adjusted in the same manner and at the same time as the Conversion Ratio.
- 13.28 In the event that the Company elects to redeem all of the Class C Shares pursuant to Articles 13.25 or to require compulsory conversion of the Class C Shares pursuant to Article 13.26 or 13.27, the Company shall fix a date for the redemption or conversion (the "Redemption Date"). In the event that the Company elects to require compulsory conversion of all of the Class C Shares pursuant to Article 13.27, the Company shall fix a date for conversion (the "Alternative Redemption Date"). Notice of redemption or conversion shall be mailed by first class mail, postage prepaid, by the Company not less than thirty (30) days prior to the Redemption Date (the "30-day Redemption Period") to the holders of the Class C Shares to be redeemed or converted at their last addresses as they shall appear on the Register. Any notice mailed in the manner herein provided shall be conclusively presumed to have been duly given whether or not the holder received such notice.
- 13.29 The Class C Shares may be converted, for cash in accordance with Article 13.6 (or on a "cashless basis" in accordance with Article 13.26) at any time after notice of redemption shall have been given by the Company pursuant to Article 13.29 and prior to the Redemption Date or the Alternative Redemption Date. In the event that the Company determines to require all holders of Class C Shares to convert their Class C Shares on a "cashless basis" pursuant to Article 13.26, the notice of redemption shall contain the information necessary to calculate the number of Class A Ordinary Shares to be received upon conversion of the Class C Shares, including the Fair Market Value (as such term is defined Article 13.26) in such case. On and after the Redemption Date or the Alterative Redemption Price, as applicable, the holder of the Class C Shares shall have no further rights except to receive the Redemption Price or the Alterative Redemption Price, as applicable.
- 13.30 The redemption rights provided in Article 13.25 shall not apply to the Class C-2 Shares if at the time of the redemption such Class C-2 Shares continue to be held by the Sponsor or its Permitted Transferees. However, once such Class C-2 Shares are transferred (other than to Permitted Transferees under Article 13.1), the Company may redeem the Class C-2 Shares pursuant to Article13.25, provided that the criteria for redemption are met, including the opportunity of the holder of such Class C-2 Shares to convert the Class C-2 Shares prior to redemption pursuant to Article 13.30. Class C-2 Shares that are transferred to persons other than Permitted Transferees shall upon such transfer cease to be Class C-2 Shares and shall become Class C-1 Shares.
- 13.31 If Class C-1 Shares are held by any of the Company's officers or directors, the Class C-1 Shares held by such officers and directors will be subject to the compulsory conversion rights provided in Article 13.26, except that such officers and directors shall only receive "Fair Market Value" ("Fair Market Value" in this Article shall mean the last sale price of the Class C-1 Shares on the Alternative Redemption Date) for such Class C-1 Shares so redeemed.
- 13.32 The Company shall from time to time promptly pay all taxes and charges that may be imposed upon the Company in respect of the re-designation or delivery of Class A Ordinary Shares upon the conversion of the Class C Shares, but the Company shall not be obligated to pay any transfer taxes in respect of the Class C Shares or such Class A Ordinary Shares. The holders of Class C Shares shall not have any duty or obligation to take any action that requires the payment of taxes and/or charges unless and until it is satisfied that all such payments have been made.

## 14. Power to attach rights to shares

- 14.1 Subject to the Companies Acts and to any rights, privileges or restrictions attached to existing shares, any share may be issued with or have attached to it such rights, privileges and restrictions, whether in regard to dividends, voting, repayment or redemption of share capital, or otherwise, as the Company may by ordinary resolution determine, or if no ordinary resolution has been passed or so far as the resolution does not make specific provision, as the Board may determine.
- 14.2 Shares may be issued with a preferential or qualified right to dividends and in the distribution of assets of the Company and with or without any special voting rights.

# 15. Allotment of shares and pre-emption

- 15.1 In accordance with section 551 of the Act, the Directors are generally and unconditionally authorized to allot shares in the Company or grant rights to subscribe for or to convert any security into shares in the Company (the "Share Rights") up to an aggregate maximum number of [•] Ordinary Shares, provided that this authority shall, unless renewed, varied or revoked by ordinary resolution of the Company, expire on the date which is five years from the Date of the Adoption, save that the Directors may allot shares or grant Share Rights in pursuance of an offer or agreement made before such expiry which would or might require shall to be allotted or Share Rights to be granted, notwithstanding that the authority conferred by this Article 15.1 has expired.
- 15.2 The Directors are generally empowered to allot equity securities (as defined in section 560 of the Act) as if section 561(1) of the Act did not apply to any such allotment, provided that this power shall: (i) be limited to the allotment of equity securities up to an aggregate maximum number of [•] Ordinary Shares; and (ii) unless renewed, varied or revoked by ordinary resolution of the Company, expire on the date which is five years from the Date of Adoption, save that the Directors may allot equity securities in pursuance of any offer or agreement made before such expiry which would or might require equity securities to be allotted after such expiry, notwithstanding that the power conferred by this Article 15.2 has expired.
- 15.3 The provisions set forth in Articles 15.1 and 15.2 may be renewed at any meeting of the members of the Company.
- 15.4 Without prejudice to any special rights previously conferred on the holders of any shares or class of shares, and subject to the Companies Acts, these Articles and to any relevant authority of the Company in general meeting required by the Companies Acts, the Board may offer, issue, allot, (with or without conferring rights of renunciation), grant options over or otherwise deal with or dispose of shares or grant rights to subscribe for or convert any security into shares with such preferred, deferred or other special rights, or subject to such restrictions, whether in regard to dividend, return of capital, voting or otherwise, as the Company may from time to time by ordinary resolution determine (or, in the absence of any such determination, as the Board may determine), to such persons, at such times, for such consideration and upon such terms as the Board may decide. No share may be issued at a discount.
- 15.5 If by the conditions of allotment of any shares the whole or part of the issue price thereof shall be payable by instalments, every such instalment shall, when due, be paid to the Company by the person who for the time being and from time to time shall be the registered holder of the shares, or his legal personal representative.
- 15.6 The Board may, at any time after the allotment of any share but before any person has been entered in the Register, recognise a renunciation by the allottee in favour of some other person and accord to the allottee of a share a right to effect such renunciation and/or allow the rights to be represented to be one or more participating securities, in each case upon the subject to such terms and conditions as the Board may think fit to impose.

#### 16. Redeemable shares

Subject to the Companies Acts and to any rights attaching to existing shares, any share may be issued which can be redeemed or is liable to be redeemed at the option of the Company or the holder. The Board may determine the terms, conditions and manner of redemption of any redeemable shares which are issued. Such terms and conditions shall apply to the relevant shares as if the same were set out in these Articles.

## 17. Pari passu issues

If new shares are created or issued which rank equally with any other existing shares, the rights of the existing shares will not be regarded as changed or abrogated unless the terms of the existing shares expressly say otherwise.

## 18. Variation of rights

- 18.1 Subject to the Companies Acts, the rights attached to any class of shares can be varied or abrogated either with the consent in writing of the holders of not less than three-quarters in nominal value of the issued share of that class (excluding any shares of that class held as treasury shares) or with the authority of a special resolution passed at a separate meeting of the holders of the relevant class of shares known as a class meeting.
- 18.2 The provisions of this Article will apply to any variation or abrogation of rights of shares forming part of a class. Each part of the class which is being treated differently is treated as a separate class in applying this Article.
- 18.3 All the provisions in these Articles as to general meetings shall apply, with any necessary modifications, to every class meeting except that:
  - (a) the quorum at every such meeting shall not be less than two persons holding or representing by proxy at least one-third of the nominal amount paid up on the issued shares of the class (excluding any shares of that class held as treasury shares); and
  - (b) if at any adjourned meeting of such holders such quorum as set out above is not present, at least one person holding shares of the class who is present in person or by proxy shall be a quorum.
- 18.4 The Board may convene a class meeting whenever it thinks fit and whether or not the business to be transacted involves a variation or abrogation of class rights.

### 19. Payment of commission

The Company may in connection with the issue of any shares or the sale for cash of treasury shares exercise all powers of paying commission and brokerage conferred or permitted by the Companies Acts. Any such commission or brokerage may be satisfied by the payment of cash or by the allotment of fully or partly paid shares or other securities or the grant of an option to call for an allotment of shares or any combination of such methods.

### 20. Trusts not recognised

Except as otherwise expressly provided by these Articles, required by law or as ordered by a court of competent jurisdiction, the Company shall not recognise any person as holding any share on any trust, and the Company shall not be bound by or required in any way to recognise (even when having notice of it) any equitable, contingent, future, partial or other claim to or interest in any share other than an absolute right of the holder of the whole of the share.

### 21. Share certificates

- 21.1 Every person (except a person to whom the Company is not by law required to issue a certificate and except as otherwise provided in these Articles) whose name is entered in the Register as a holder of any shares shall be entitled, without charge, to receive within the time limits prescribed by the Companies Acts (unless the terms of issue prescribe otherwise) one certificate for all of the shares of that class registered in his or her name
- The Company shall not be bound to issue more than one certificate in respect of shares held jointly by two or more persons. Delivery of a certificate to the person first named in the Register shall be sufficient delivery to all joint holders.
- 21.3 Where a member has transferred part only of the shares comprised in a certificate, the member shall be entitled without charge to a certificate for the balance of such shares. Where a member receives more shares of any class, the member shall be entitled without charge to a certificate for the extra shares of that class.
- 21.4 A share certificate may be issued under Seal (by affixing the Seal to or printing the Seal or a representation of it on the certificate) or signed by at least two Directors or by at least one Director and the Secretary. Such certificate shall specify the number and class of the shares in respect of which it is issued and the amount or respective amounts paid up on it. The Board may be resolution decide, either generally or in any particular case or cases, that any signatures on any share certificates need not be autographic but may be applied to the certificates by some mechanical or other means or may be printed on them or that the certificates need not be signed by any person.
- 21.5 Every share certificate sent in accordance with these Articles will be sent at the risk of the member or other person entitled to the certificate. The Company will not be responsible for any share certificate lost or delayed in the course of delivery.

## 22. Replacement certificates

- 22.1 Any two or more certificates representing shares of any one class held by any member may at the request of the member be cancelled and a single new certificate for such shares issued in lieu without charge on surrender of the original certificates for cancellation.
- 22.2 Any certificate representing shares of any one class held by any member may at the request of the member be cancelled and two or more certificates for such shares may be issued instead.
- 22.3 If a share certificate is defaced, worn out or said to be stolen, lost or destroyed, it may be replaced on such terms as to evidence and indemnity as the Board may decide and, where it is defaced or worn out, after delivery of the old certificate to the Company.
- 22.4 The Board may require the payment of any reasonable exceptional out-of-pocket expenses of the Company incurred in connection with the issue of any certificates under this Article. In the case of shares held jointly by several persons, any such request as is mentioned in this Article may be made by any one of the joint holders.

## 23. Lien on shares not fully paid

The Company shall have a first and paramount lien on every share, not being a fully paid share, for all amounts payable to the Company (whether presently or not) in respect of that share. The Company's lien over a share takes priority over any third party's interest in that share, and extends to any dividend or other money payable by the Company in respect of that share (and, if the lien is enforced and the share is sold by the Company, the proceeds of sale of that share). The Board may at any time, either generally or in any particular case, waive any lien that has arisen or declare any share to be wholly or in part exempt from the provisions of this Article.

## 24. Enforcement of lien by sale

The Company may sell, in such manner as the Board may decide, any share over which the Company has a lien if a sum in respect of which the lien exists is presently payable and is not paid within 14 clear days after a notice has been served on the holder of the share or the person who is entitled by transmission to the share, demanding payment and stating that if the notice is not complied with the share may be sold. For giving effect to the sale the Board may authorise some person to sign an instrument of transfer of the share sold to, or in accordance with the directions, of the buyer. The buyer shall not be bound to see to the application of the purchase money, nor shall the buyer's title to the share be affected by any irregularity or invalidity in the proceedings in reference to the sale.

### 25. Application of proceeds of sale

The net proceeds of any sale of shares subject to any lien, after payment of the costs, shall be applied:

- (a) first, in or towards satisfaction of so much of the amount due to the Company or of the liability or engagement (as the case may be) as is presently payable or is liable to be presently fulfilled or discharged; and
- (b) second, any residue shall be paid to the person who was entitled to the share at the time of the sale but only after the certificate for the shares sold has been surrendered to the company for cancellation, or an indemnity in a form reasonably satisfactory to the directors has been given for any lost certificates, and subject to a like lien for debts or liabilities not presently payable as existed on the share prior to the sale.

#### 26. Calls

- 26.1 Subject to these Articles and the terms on which the shares are allotted, the Board may from time to time make calls on the members in respect of any monies unpaid on their shares (whether in respect of nominal value or premium) and not payable on a date fixed by or in accordance with the terms of issue.
- 26.2 The Company may make arrangements on the issue of shares for a difference between the holders of such shares in the amount of calls to be paid and the time of payment of such calls.
- 26.3 Each member shall (subject to the Company serving upon him or her at least 14 clear days' notice specifying when and where payment is to be made and whether or not by instalments) pay to the Company as required by the notice the amount called on such member's shares.
- 26.4 A call shall be deemed to have been made at the time when the resolution of the Board authorising the call was passed.
- 26.5 A call may be revoked or postponed, in whole or in part, as the Board may decide.
- 26.6 Liability to pay a call is not extinguished or transferred by transferring the shares in respect of which the call is required to be paid.

# 27. Liability of joint holders

The joint holders of a share shall be jointly and severally liable to pay all calls and payments in respect of the share.

### 28. Interest on calls

If a call remains unpaid after it has become due and payable, the person from whom it is due and payable shall pay all expenses that have been incurred by the Company by reason of such non-payment together with interest on the amount unpaid from the day it is due and payable to the time of actual payment at such rate (not exceeding the Bank of England base rate by more than five percentage points) as the Board may decide. The Board may waive payment of the interest or the expenses in whole or in part.

# 29. Power to differentiate

On or before the issue of shares, the Board may decide that allottees or holders of shares can be called on to pay different amounts or that they can be called on at different times.

## 30. Payment of calls in advance

The Board may, if it thinks fit, receive from any member willing to advance the same, all or any part of the monies uncalled and unpaid on the shares held by him or her. Such payment in advance of calls shall, to the extent of the payment, extinguish the liability on the shares on which it is made. The Company may pay interest on the money paid in advance, or so much of it as exceeds the amount for the time being called upon the shares in respect of which such advance has been made, at such rate as the Board may decide. The Board may at any time repay the amount so advanced by giving at least three months' notice in writing to such member of its intention to do so, unless before the expiration of such notice the amount so advanced shall have been called up on the shares in respect of which it was advanced.

### 31. Notice if call or instalment not paid

If any member fails to pay the whole of any call (or any instalment of any call) by the date when payment is due, the Board may at any time give notice in writing to such member (or to any person entitled to the shares by transmission), requiring payment of the amount unpaid (and any accrued interest and any expenses incurred by the Company by reason of such non-payment) by a date not less than 14 clear days from the date of the notice. The notice shall name the place where the payment is to be made and state that, if the notice is not complied with, the shares in respect of which such call was made will be liable to be forfeited.

# 32. Forfeiture for non-compliance

If the notice referred to in Article 31 is not complied with, any share for which it was given may be forfeited, by resolution of the Board to that effect, at any time before the payment required by the notice has been made. Such forfeiture shall include all dividends declared or other monies payable in respect of the forfeited shares and not paid before the forfeiture.

### 33. Notice after forfeiture

When any share has been forfeited, notice of the forfeiture shall be served on the holder of the share or the person entitled to such share by transmission (as the case may be) before forfeiture. An entry of such notice having been given and of the forfeiture and the date of forfeiture shall immediately be made in the Register in respect of such share. However, no forfeiture shall be invalidated by any omission to give such notice or to make such entry in the Register.

# 34. Forfeiture may be annulled

The Board may annul the forfeiture of a share, at any time before any forfeited share has been cancelled or sold, re-allotted or otherwise disposed of, on the terms that payment shall be made of all calls and interest due on it and all expenses incurred in respect of the share and on such further terms (if any) as the Board shall see fit.

# 35. Surrender

The Board may accept the surrender of any share liable to be forfeited and, in any event, references in these Articles to forfeiture shall include surrender.

#### 36. Sale of forfeited shares

- 36.1 A forfeited share shall become the property of the Company.
- 36.2 Subject to the Companies Acts, any such share may be sold, re-allotted or otherwise disposed of, on such terms and in such manner as the Board thinks fit.
- 36.3 The Board may, for the purposes of the disposal, authorise some person to transfer the share in question and may enter the name of the transferee in respect of the transferred share in the Register even if no share certificate is lodged and may issue a new certificate to the transferee. An instrument of transfer executed by that person shall be as effective as if it had been executed by the holder of, or the person entitled by transmission to, the share. The Company may receive the consideration (if any) given for the share on its disposal.

#### 37. Effect of forfeiture

A shareholder whose shares have been forfeited shall cease to be a member in respect of such forfeited shares and shall surrender the certificate for such shares to the Company for cancellation. Such shareholder shall remain liable to pay to the Company all sums which at the date of forfeiture were presently payable by him or her to the Company in respect of such shares with interest (not exceeding the Bank of England base rate by five percentage points) from the date of the forfeiture to the date of payment. The Directors may waive payment of interest wholly or in part and may enforce payment, without any reduction or allowance for the value of the shares at the time of forfeiture or for any consideration received on their disposal.

#### 38. Evidence of forfeiture

A statutory declaration by a Director or the Secretary that a share has been forfeited on a specified date shall be conclusive evidence of the facts stated in it as against all persons claiming to be entitled to the share. The declaration shall (subject to the execution of an instrument of transfer if necessary) constitute a good title to the share. The person to whom the share is transferred or sold shall not be bound to see to the application of the purchase money or other consideration (if any), nor shall his or her title to the share be affected by any act, omission or irregularity relating to or connected with the proceedings in reference to the forfeiture or disposal of the share.

### 39. Form of transfer

- 39.1 Subject to these Articles, each member may transfer all or any of his or her shares by instrument of transfer in writing in any usual form or in any form approved by the Board. Such instrument shall be executed by or on behalf of the transferor and (in the case of a transfer of a share which is not fully paid up) by or on behalf of the transferee. All instruments of transfer, when registered, may be retained by the Company.
- 39.2 The transferor of a share shall be deemed to remain the holder of the share concerned until the name of the transferee is entered in the Register in respect of it.
- 39.3 The directors are authorised to establish such clearing and settlement procedures for the shares of the Company as they deem fit from time to

## 40. Right to refuse registration of transfer

- 40.1 The Board may, in its absolute discretion, refuse to register any transfer of a share (or renunciation of a renounceable letter of allotment) unless:
  - (a) it is for a share which is fully paid up;
  - (b) it is for a share upon which the Company has no lien;

- (c) it is only for one class of share;
- (d) it is in favour of a single transferee or no more than four joint transferees;
- (e) it is duly stamped or is duly certificated or otherwise shown to the satisfaction of the Board to be exempt from stamp duty (if this is required); and
- (f) it is delivered for registration to the Office (or such other place as the Board may determine), accompanied (except in the case of a transfer by a person to whom the Company is not required by law to issue a certificate and to whom a certificate has not been issued or in the case of a renunciation) by the certificate for the shares to which it relates and such other evidence as the Board may reasonably require to prove the title of the transferor (or person renouncing) and the due execution of the transfer or renunciation by him or her or, if the transfer or renunciation is executed by some other person on his or her behalf, the authority of that person to do so.
- 40.2 Transfers of shares will not be registered in the circumstances referred to in Article 76.

### 41. Notice of refusal to register a transfer

If the Board refuses to register a transfer of a share it shall notify the transferee of the refusal and the reasons for it within two months after the date on which the transfer was lodged with the Company or the instructions to the relevant system received. Any instrument of transfer which the Board refuses to register shall be returned to the person depositing it (except if there is suspected or actual fraud). All instruments of transfer which are registered may be retained by the Company.

# 42. No fees on registration

No fee shall be charged for registration of a transfer or other document or instruction relating to or affecting the title to any share or for making any other entry in the Register.

## 43. Other powers in relation to transfers

Nothing in these Articles shall prevent the Board:

- (a) from recognising a renunciation of the allotment of any share by the allottee in favour of another person; or
- (b) (if empowered to do so by these Articles) from authorising any person to execute an instrument of transfer of a share and from authorising any person to transfer that share in accordance with any procedures implemented under Article 24.

## 44. Transmission of shares on death

If a member dies, the survivors or survivor (where the member was a joint holder), and his or her executors or administrators (where the member was a sole or the only survivor of joint holders), shall be the only persons recognised by the Company as having any title to his or her shares. Nothing in these Articles shall release the estate of a deceased member from any liability for any share which has been solely or jointly held by such member.

### 45. Election of person entitled by transmission

45.1 Any person becoming entitled to a share because of the death or bankruptcy of a member, or otherwise by operation of law, may (on such evidence as to his or her title being produced as the Board may require) elect either to become registered as a member or to have some person nominated by him or her registered as a member. If such person elects to become registered himself or herself, he or she shall notify the

Company to that effect. If such person elects to have some other person registered, he or she shall execute an instrument of transfer of such share to that person. All the provisions of these Articles relating to the transfer of shares shall apply to the notice or instrument of transfer (as the case may be) as if it were an instrument of transfer executed by the member and his or her death, bankruptcy or other event had not occurred. Where the entitlement of a person to a share because of the death or bankruptcy of a member or otherwise by operation of law is proved to the satisfaction of the Board, the Board shall within 10 Business Days after proof cause the entitlement of that person to be noted in the Register.

# 46. Rights on transmission

Where a person becomes entitled to a share because of the death or bankruptcy of any member, or otherwise by operation of law, the rights of the holder in relation to such share shall cease. However, the person so entitled may give a good discharge for any dividends and other monies payable in respect of it and shall have the same rights to which he or she would be entitled if the holder of the share, except that he or she shall not be entitled to receive notice of, or to attend or vote at, any meeting of the Company or a separate meeting of the holders of any class of shares of the Company before being registered as the holder of the share. The Board may at any time give notice requiring any such person to elect either to be registered himself or herself or to transfer the share. If the notice is not complied with within 90 days, the Board may withhold payment of all dividends and the other monies payable in respect of such share until the requirements of the notice have been complied with.

## 47. Destruction of documents

- 47.1 The Company may destroy any:
  - (a) instrument of transfer, after six years from the date on which it is registered;
  - (b) dividend mandate or any variation or cancellation of a dividend mandate or any notification of change of name or address, after two years from the date on which it is recorded;
  - (c) share certificate, after one year from the date on which it is cancelled;
  - (d) instrument of proxy which has been used for the purpose of a poll at any time after one year has elapsed from the date of use;
  - (e) instrument of proxy which has not been used for the purpose of a poll at any time after a period of one month has elapsed from the end of the meeting to which the instrument of proxy relates; or
  - (f) other document for which any entry in the Register is made, after six years from the date on which an entry was first made in the Register in respect of it, provided that the Company may destroy any such type of document at a date earlier than that authorised by this Article if a copy of such document is made and retained (whether electronically, by microfilm, by digital imaging or by other similar means) until the expiration of the period applicable to the destruction of the original of such document.
- 47.2 It shall be conclusively presumed in favour of the Company that every:
  - (a) entry in the Register purporting to have been made on the basis of a document so destroyed was duly and properly made;
  - (b) instrument of transfer so destroyed was duly registered;
  - (c) share certificate so destroyed was duly cancelled; and
  - (d) other document so destroyed had been properly dealt with under its terms and was valid and effective according to the particulars in the records of the Company.
- 47.3 This Article shall only apply to the destruction of a document in good faith and without notice of any claim (regardless of the parties to it) to which the document might be relevant. Nothing in this Article shall be

construed as imposing any liability on the Company in respect of the destruction of any such document other than as provided for in this Article which would not attach to the Company in the absence of this Article. References in this Article to the destruction of any document include references to the disposal of it in any manner.

### 48. Sub-division

Any resolution authorising the Company to sub-divide its shares or any of them may determine that, as between the shares resulting from the sub-division, any of them may have any preference or advantage or be subject to any restriction as compared with the others.

#### 49. Fractions

Subject to the Articles, if any shares are consolidated or consolidated and then divided, the Board has power to deal with any fractions of shares which result. If the Board decides to sell any shares representing fractions, it can do so for the best price reasonably obtainable and distribute the net proceeds of sale among members in proportion to their fractional entitlements. The Board can sell those shares to anyone, including the Company if the legislation allows, and may authorise any person to transfer or deliver the shares to the buyer or in accordance with the buyer's instructions. The Buyer shall not be bound to see to the application of the purchase money, nor shall the buyer's title to the share be affected by any irregularity or invalidity in the proceedings in reference to the sale.

## 50. Annual general meetings

An annual general meeting shall be held once a year, at such time (consistent with the terms of the Companies Acts) and place, including partly (but not wholly) by means of electronic facility or facilities, as may be determined by the Board.

## 51. Convening of general meetings

- 51.1 The Board may, whenever it thinks fit, and shall on requisition in accordance with the Companies Acts, proceed to convene a general meeting.
- 51.2 Subject always to Article 60.3, the Board may make whatever arrangements it considers fit to allow those entitled to do so to attend and participate in any general meeting.
- 51.3 The Board shall determine in relation to each general meeting the means of attendance at and participation in the meeting, including whether the persons entitled to attend and participate in the meeting shall be enabled to do so:
  - (a) (subject to Article 60.3) by means of electronic facility or facilities pursuant to Article 52 (and for the avoidance of doubt, the Board shall be under no obligation to offer or provide such facility or facilities, whatever the circumstances); and/or
  - (b) by simultaneous attendance and participation at a satellite meeting place or places pursuant to Article 54.7.
- 51.4 A general meeting may be held at two or more places using any technology that enables members who are not together at the same place to listen, speak and vote at such meeting. Specifically, a general meeting may be held as a physical meeting, a hybrid meeting or an electronic meeting, as may be determined by the Directors in their absolute discretion.
- 51.5 Two or more persons who may not be in the same place as each other attend a general meeting if their circumstances are such that if they have (or were to have) rights to speak and vote at that meeting, they are (or would be) able to exercise them.

- 51.6 A person is able to participate in a meeting if that person's circumstances are such that if he or she has (or were to have) rights in relation to the meeting, he or she is (or would be) able to exercise them.
- 51.7 In determining whether persons are attending or participating in a meeting, other than at a physical place or places, it is immaterial where any of them are or how they are able to communicate with each other.
- 51.8 A person is able to exercise the right to speak at a general meeting when that person is in a position to communicate to all those attending the meeting, during the meeting, any information or opinions which that person has on the business of the meeting.
- 51.9 A person is able to exercise the right to vote at a general meeting when:
  - (a) that person is able to vote, during the meeting (or, in the case of a poll, within the time period specified by the chair of the meeting) on resolutions put to the vote at the meeting; and
  - (b) that person's vote can be taken into account in determining whether or not such resolutions are passed at the same time as the votes of all the other persons attending the meeting.
- 51.10 If, at any general meeting at which members are entitled to participate by means of electronic facility or facilities determined by the Board pursuant to Article 52, any document is required to be on display or to be available for inspection at the meeting (whether prior to or for the duration of the meeting or both), the Company shall ensure that it is available in electronic form to persons entitled to inspect it for at least the required period of time, and this will be deemed to satisfy any such requirement.

## 52. Simultaneous attendance and participation by electronic facilities

Without prejudice to Article 54.7, the Board may resolve to enable persons entitled to attend and participate in a general meeting to do so partly (but not wholly) by simultaneous attendance and participation by means of electronic facility or facilities, and may determine the means, or all different means, of attendance and participation used in relation to the general meeting. The members present in person or by proxy by means of an electronic facility or facilities (as so determined by the Board) shall be counted in the quorum for, and be entitled to participate in, the general meeting in question. That meeting shall be duly constituted and its proceedings valid if the chair is satisfied that adequate facilities are available throughout the meeting to ensure that members attending the meeting by all means (including the means of an electronic facility or facilities) are able to:

- (a) participate in the business for which the meeting has been convened;
- (b) hear all persons who speak at the meeting; and
- (c) be heard by all other persons attending and participating in the meeting.

# 53. Notice of general meetings

A general meeting shall be called by at least such minimum notice as is required or permitted by the Companies Acts. The period of notice shall in either case be exclusive of the day on which it is served or deemed to be served and of the day on which the meeting is to be held and shall be given to all members other than those who are not entitled to receive such notices from the Company. The Company may give such notice by any means or combination of means permitted by the Companies Acts.

# 54. Contents of notice of general meetings

54.1 Every notice calling a general meeting shall specify the place (including any satellite meeting place or places determined pursuant to Article 54.7), date and time of the meeting. There shall appear with reasonable prominence in every such notice a statement that a member entitled to attend and vote is entitled to a proxy or (if he or she has more than one share) proxies to exercise all or any of his or her

rights to attend, speak and vote and that a proxy need not be a member of the Company. Such notice shall also include the address of the website on which the information required by the Act is published, state the procedures with which members must comply in order to be able to attend and vote at the meeting (including the date by which they must comply), provide details of any forms to be used for the appointment of a proxy and state that a member has the right to ask questions at the meeting in accordance with the Act.

- The notice shall specify the general nature of the business to be transacted at the meeting and shall set out the text of all resolutions to be considered by the meeting and shall state in each case whether it is proposed as an ordinary resolution or as a special resolution.
- 54.3 In the case of an annual general meeting, the notice shall also specify the meeting as such.
- 54.4 If pursuant to Article 52 the Board determines that a general meeting shall be held partly by means of electronic facility or facilities, the notice shall:
  - (a) include a statement to that effect;
  - (b) specify the means, or all different means, of attendance and participation thereat, and any access, identification and security arrangements determined pursuant to Article 64; and
  - (c) state how it is proposed that persons attending or participating in the meeting electronically should communicate with each other during the meeting.
- 54.5 The notice shall specify such arrangements as have at that time been made for the purpose of Article 54.7.
- 54.6 For the purposes of determining which persons are entitled to attend or vote at a meeting and how many votes a person may cast, the Company may specify in the notice of meeting a time, not more than 48 hours before the time fixed for the meeting (not taking into account non-working days) by which a person must be entered in the Register in order to have the right to attend or vote at the meeting or appoint a proxy to do so.
- 54.7 Without prejudice to Article 52, the Board may resolve to enable persons entitled to attend and participate in a general meeting to do so by simultaneous attendance and participation at a satellite meeting place or places anywhere in the world. The members present in person or by proxy at satellite meeting places shall be counted in the quorum for, and entitled to participate in, the general meeting in question, and the meeting shall be duly constituted and its proceedings valid if the chair is satisfied that adequate facilities are available throughout the meeting to ensure that members attending at all the meeting places are able to:
  - (a) participate in the business for which the meeting has been convened;
  - (b) hear all persons who speak (whether by the use of microphones, loudspeakers, audio-visual communications equipment or otherwise) in the principal meeting place and any satellite meeting place; and
  - (c) be heard by all other persons so present in the same way, and the meeting shall be deemed to take place at the place where the chair of the meeting presides (the principal meeting place, with any other location where that meeting takes place being referred in these Articles as a satellite meeting). The chair shall be present at, and the meeting shall be deemed to take place at, the principal meeting place and the powers of the chair shall apply equally to each satellite meeting place, including his or her power to adjourn the meeting as referred to in Article 61.
- 54.8 If the general meeting is to be a hybrid meeting or an electronic meeting, the notice shall include a statement to that effect and with details of the electronic facilities for attendance and participation at the meeting or where such details will be made available by the Company prior to the meeting. Unless otherwise specified in the notice of meeting or determined by the chair of the meeting, a general meeting is deemed to take place at the place where the chair of the meeting is at the time of the meeting.

### 55. Omission to give notice and non-receipt of notice

The accidental omission to give notice of any meeting or to send an instrument of proxy (where this is intended to be sent out with the notice) to, or the non-receipt of either by, any person entitled to receive the same shall not invalidate the proceedings of that meeting.

### 56. Postponement of general meeting

If, after the sending of the notice of a general meeting but before the meeting is held, or after the adjournment of a general meeting but before the adjourned meeting is held (whether or not notice of the adjourned meeting is required), the Board, in its absolute discretion, considers that it is impracticable or unreasonable for any reason to hold a general meeting on the date or at the time or place specified in the notice calling the general meeting (including a satellite meeting to which Article 54.7 applies) and/or by means of the electronic facility or facilities specified in the notice, it may postpone the general meeting to another date, time and/or place (or in the case of a general meeting to be held at a principal meeting place and one or more satellite meeting places, to such other places) and/or change the electronic facility or facilities. If such a decision is made, the Board may then change the place (or any of the places in the case of a general meeting to which Article 54.7 applies) and/or the electronic facility or facilities and/or postpone the date and/or time again if it considers that it is reasonable to do so. No new notice of the general meeting need be sent but the Board shall take reasonable steps to ensure that notice of the change of date, time, place (or places, in the case of a general meeting to which Article 54.7 applies) of and/or electronic facility or facilities for the postponed meeting appear at the original time and at the original place (or places, in the case of a general meeting to which Article 54.7 applies) and/or on the original electronic facility or facilities. When a general meeting is so postponed, notice of the date, time and place (or places in the case of a meeting to which Article 54.7 applies), including any electronic facility if applicable, of the postponed meeting shall be given in such manner as the Board may, in its absolute discretion, determine. No business shall be transacted at any postponed meeting other than business which might properly have been transacted at the meeting had it not been postponed. Notice of the business to be transacted at such postponed meeting shall not be required. If a general meeting is postponed in accordance with this Article 56, the appointment of a proxy will be valid if it is delivered and received as required by these Articles not less than 48 hours before the time appointed for holding the postponed meeting. When calculating the 48 hour period mentioned in this Article, the Directors can decide not to take account of any part of a day that is not a working day.

# 57. Quorum at general meeting

No business shall be transacted at any general meeting unless a quorum is present. If a quorum is not present a chair of the meeting can still be chosen and this will not be treated as part of the business of the meeting. At least two members that in aggregate hold at least 51% of the issued shares of the Company who are present in person or by proxy and entitled to attend and to vote on the business to be transacted shall be a quorum for a general meeting for all purposes.

## 58. Procedure if quorum not present

If a quorum is not present within fifteen minutes (or such longer interval as the chair in his or her absolute discretion thinks fit) from the time appointed for holding a general meeting, or if a quorum ceases to be present during a meeting, the meeting shall be dissolved if convened on the requisition of members. In any other case, the meeting shall stand adjourned to such day (not being less than ten clear days after the date of the original meeting), and at such time and place or places, with such means of attendance and participation (including partly but not wholly by means of electronic facility or facilities), as the chair (or, in default, the Board) may determine at his/her/its absolute discretion. If at such adjourned meeting a quorum is not present within fifteen minutes from the time appointed for holding the meeting, one person

entitled to vote on the business to be transacted, being a member or a proxy for a member or a duly authorised representative of a corporation which is a member, shall be a quorum and any notice of an adjourned meeting shall state this.

## 59. Chair of general meeting

The chair of the Board shall preside at every general meeting of the Company. If there is no such chair or if at any meeting he or she shall not be present within five minutes after the time appointed for holding the meeting, or shall be unwilling to act as chair, the deputy chair (if any) of the Board shall, if present and willing to act, preside at such meeting. If more than one deputy chair is present they shall agree amongst themselves who is to take the chair or, if they cannot agree, the deputy chair who has been in office as a director the longest shall take the chair. If no chair or deputy chair shall be so present and willing to act, the Directors present shall choose one of their number to act or, if there be only one Director present, he or she shall be chair if willing to act. If there be no Director present and willing to act, the members present and entitled to vote shall choose one of their number to be chair of the meeting. Nothing in these Articles shall restrict or exclude any of the powers or rights of a chair of a meeting which are given by law. The chair of a general meeting (be it a physical meeting, a hybrid meeting or an electronic meeting) shall, for the purpose of conducting the meeting in orderly manner, have power to take all such steps and actions as he deems appropriate to maintain order during the meeting.

## 60. Entitlement to attend, speak and participate

- 60.1 A Director (and any other person invited by the chair to do so) may attend and speak at any general meeting and at any separate meeting of the holders of any class of shares of the Company, whether or not also a member.
- 60.2 All persons seeking to attend and participate in a general meeting by way of electronic facilities shall be responsible for maintaining adequate facilities to enable them to do so. Subject only to the requirement for the chair to adjourn a general meeting in accordance with the provisions of Article 61.2, any inability of a person or persons to attend or participate in a general meeting by way of electronic facility or facilities shall not invalidate the proceedings of that meeting.
- 60.3 Nothing in these Articles authorises or allows a general meeting to be held exclusively on an electronic basis.

## 61. Adjournments

- 61.1 The chair may, with the consent of a meeting at which a quorum is present, and shall, if so directed by the meeting, adjourn any meeting from time to time (or indefinitely) and from place to place (or, in the case of a meeting held at a principal meeting place and one or more satellite meeting places, such other places) and/or from such electronic facility or facilities for attendance and participation to such other electronic facility or facilities as the meeting shall determine. However, without prejudice to any other power which the chair may have under these Articles (including the power to adjourn a meeting conferred by Article 61.2) or at common law, the chair may, without the need for the consent of the meeting and before or after it has started and irrespective of whether a quorum is present, interrupt or adjourn any meeting from time to time (or indefinitely) and from place to place (or places in the case of a meeting to which Article 54.7 applies) or from electronic facility to electronic facility, or for an indefinite period, if of the opinion that it has become necessary to do so in order:
  - (a) to secure the proper and orderly conduct of the meeting; or
  - (b) to give all persons entitled to do so a reasonable opportunity of attending, speaking and voting at the meeting; or
  - (c) to ensure that the business of the meeting is properly disposed of.

- 61.2 If it appears to the chair that the facilities at the principal meeting place or any satellite meeting place or an electronic facility or facilities or security at any general meeting have become inadequate for the purposes referred to in Articles 52 or 54.7, or are otherwise not sufficient to allow the meeting to be conducted substantially in accordance with the provisions set out in the notice of meeting, then the chair shall, without the consent of the meeting, interrupt or adjourn the general meeting.
- 61.3 All business conducted at a meeting up to the time of any adjournment shall, subject to Article 61.4, be valid.
- The chair may specify that only the business conducted at the meeting up to a point in time which is earlier than the time of the adjournment is valid, if in his or her opinion, to do so would be more appropriate.

### 62. Notice of adjournment

Any adjournment pursuant to Article 61 may, subject to the Act, be for such time and with such means of attendance and participation (including at such place or places and/or by means of such electronic facility or facilities) as the chair (or, in default, the Board) may in his, her or its absolute discretion determine, notwithstanding that by reason of the adjournment some members may be unable to attend and participate in the adjourned meeting. Whenever a meeting is adjourned for 30 days or more or indefinitely, at least seven clear days' notice, specifying the day, the time and the place or places of the adjourned meeting and the means of attendance and participation (including by means of electronic facility or facilities if applicable) as the chair (or, in default, the Board) may in his or her absolute discretion determine, and the general nature of the business to be transacted, shall be given in the same manner as in the case of the original meeting. Save as aforesaid and subject to the Act, no member shall be entitled to any notice of an adjournment or of the business to be transacted at any adjourned meeting. Where a meeting is adjourned indefinitely, form and (where applicable) place for the adjourned meeting shall be fixed by the Directors.

### 63. Business of adjourned meeting

No business shall be transacted at any adjourned meeting other than the business which might properly have been transacted at the meeting from which the adjournment took place.

## 64. Accommodation of members, security arrangements and orderly conduct at general meetings

- 64.1 The Board may, for the purpose of controlling the level of attendance or ensuring the safety of those attending at any place specified for the holding of a general meeting, ensuring the security of the meeting and ensuring the future orderly conduct of the meeting, from time to time make such arrangements as it shall in its absolute discretion consider to be appropriate and may from time to time vary any such arrangements or make new arrangements therefor. Any decision made under this Article 64.1 shall be final and the entitlement of any member or proxy to attend a general meeting at such place (or places, in the case of a meeting to which Article 54.7 applies) shall be subject to any such arrangements as may be for the time being approved by the Board.
- 64.2 The Board may direct that any person wishing to attend any general meeting held at a physical place should provide evidence of identity and submit to such searches or other security arrangements or restrictions (including restrictions in items of personal property to be taken into the meeting) as the Board shall consider appropriate in the circumstances.
- 64.3 If a general meeting is held partly by means of an electronic facility or facilities pursuant to Article 52, the Board and the chair may make any arrangement and impose any requirement or restriction that is:
  - (a) necessary to ensure the identification of those taking part by means of such electronic facility or facilities and the security of the electronic communication; and
  - (b) in its or his or her view, proportionate to those objectives. In this respect, the Board may authorise any voting application, system or facility for attendance and participation as it sees fit.

- 64.4 The Board shall be entitled in its absolute discretion to authorise one or more persons (including the Directors, the company secretary or the chair) to refuse physical or electronic entry to, or eject (physically or electronically) from, any meeting any person who fails to provide such evidence of identity or to submit to such searches or to otherwise comply with such security arrangements or restrictions as are required pursuant to this Article, or who causes the meeting to become disorderly.
- 64.5 Subject to the Act (and without prejudice to any other powers vested in the chair of a meeting) when conducting a general meeting, the chair may make whatever arrangement and take such action or give such directions as he or she considers, in his or her absolute discretion, to be appropriate or conducive to promote the orderly conduct of the meeting, to promote the conduct of the business laid down in the notice of the meeting with reasonable despatch and to maintain good order. The chair's decision on points of order, matters of procedure or on matters arising incidentally from the business of the meeting shall be final and conclusive, as shall his or her determination as to whether any point or matter is of such a nature.
- 64.6 Without affecting the generality of Article 61.4, the Board may, at its absolute discretion, arrange for persons entitled to attend a general meeting to do so by simultaneous attendance and participation by means of electronic facilities at such location or locations determined by the Board from time to time at its absolute discretion.
- All persons seeking to attend and participate in an electronic meeting or a hybrid meeting shall be responsible for maintaining adequate facilities to enable them to do so. Any inability of a person or persons to attend or participate in a general meeting by way of electronic facilities shall not invalidate the proceedings of and/or resolutions passed at that meeting.
- 64.8 A physical meeting may also be held by means of such telephone, electronic or other communication facilities as permit all persons participating in the meeting to communicate with each other simultaneously and instantaneously, and participation in such a meeting shall constitute presence in person at such meeting.

#### 65. Amendment to resolutions

- 65.1 If an amendment to any resolution under consideration is proposed but is ruled out of order by the chair of the meeting in good faith, any error in such ruling shall not invalidate the proceedings on the original resolution.
- 65.2 In the case of a resolution duly proposed as a special resolution, no amendment to it (other than an amendment to correct a patent error) may in any event be considered or voted on. In the case of a resolution duly proposed as an ordinary resolution no amendment to it (other than an amendment to correct a patent error) may be considered or voted on unless either at least 48 hours prior to the time appointed for holding the meeting or adjourned meeting at which such ordinary resolution is to be proposed, notice in writing of the terms of the amendment and intention to move the same has been lodged at the Office or received in electronic form at the electronic address at which the Company has or is deemed to have agreed to receive it or the chair of the meeting in his or her absolute discretion decides that it may be considered or voted on.

# 66. Members' resolutions

- Members of the Company shall have the rights provided by the Companies Acts to have the Company circulate and give notice of a resolution which may be properly moved, and is intended to be moved, at the Company's next annual general meeting.
- 66.2 Expenses of complying with these rights shall be borne in accordance with the Companies Acts.

## 67. Method of voting

67.1 Any resolution put to the vote at a general meeting shall be decided on a poll.

67.2 At general meetings, resolutions shall be put to the vote by the chair of the meeting and there shall be no requirement for the resolution to be proposed or seconded by any person.

### 68. Objection to error in voting

No objection shall be raised to the qualification of any voter or to the counting of, or failure to count, any vote, except at the meeting or adjourned meeting at which the vote objected to is given or tendered or at which the error occurs. Any objection or error shall be referred to the chair of the meeting and shall only vitiate the decision of the meeting on any resolution if the chair decides that the same is of sufficient magnitude to vitiate the resolution or may otherwise have affected the decision of the meeting. The decision of the chair of the meeting on such matters shall be final and conclusive.

## 69. Procedure on a poll

- 69.1 Any poll duly demanded on the election of a chair or on any question of adjournment shall be taken immediately. A poll duly demanded on any other matter shall be taken in such manner (including the use of ballot, voting papers, tickets or electronic means or any combination thereof) and at such time and place, not more than 30 days from the date of the meeting or adjourned meeting at which the poll was demanded, and by such means of attendance and participation (including at such place or places and/or by means of such electronic facility or facilities) as the chair shall direct. The chair may appoint scrutineers who need not be members. It is not necessary to give notice of a poll not taken immediately if the time and place at which it is to be taken are announced at the meeting at which it is demanded. In any other case, at least seven clear days' notice shall be given specifying the time, date and place at the which the poll shall be taken. The result of the poll shall be deemed to be the resolution of the meeting at which the poll was demanded.
- 69.2 On a poll votes may be given in person or by proxy. Members entitled to more than one vote need not, if they vote, use all their votes or cast all the votes they use in the same way.

### 70. Votes of members

- 70.1 Subject to the Companies Acts, to any special terms as to voting on which any shares may have been issued or may for the time being be held and to any suspension or abrogation of voting rights under these Articles, members holding Class A Ordinary Shares and Class B Ordinary Shares shall at all times vote as one class on all matters, and every member holding Class A Ordinary Shares present in person or by proxy shall have one (1) vote for each Class A Ordinary Share it holds, and every member holding Class B Ordinary Shares present in person or by proxy shall have ten (10) votes for each Class B Ordinary Share it holds.
- 70.2 Subject to the Companies Acts, to any special terms as to voting on which any shares may have been issued or may for the time being be held and to any suspension or abrogation of voting rights under these Articles, every member holding Class C Shares present in person or by proxy shall have one (1) vote for each Class C Share it holds.
- 70.3 In any class meeting of: (a) the members holding Class A Ordinary Shares, each Class A Ordinary Share shall be considered as carrying the same number of votes as the other Class A Ordinary Shares; (b) the members holding Class B Ordinary Shares, each Class B Ordinary Share shall be considered as carrying the same number of votes as the other Class B Ordinary Shares; or (c) the members holding Class C Shares, each Class C Share shall be considered as carrying the same number of votes as the other Class C Shares.
- 70.4 If two or more persons are joint holders of a share, then in voting on any question the vote of the senior who tenders a vote, whether in person or by proxy, shall be accepted to the exclusion of the votes of the other joint holders. For this purpose, seniority shall be determined by the order in which the names of the holders stand in the Register.

- Where in England or elsewhere a receiver or other person (by whatever name called) has been appointed by any court claiming jurisdiction in that behalf to exercise powers with respect to the property or affairs of any member on the ground (however formulated) of mental disorder, the Board may in its absolute discretion, upon or subject to production of such evidence of the appointment as the Board may require, permit such receiver or other person on behalf of such member to vote in person, on a poll, by proxy on behalf of such member at any general meeting or to exercise any other right conferred by membership in relation to meetings of the Company. Evidence to the satisfaction of the Board of the authority of the person claiming to exercise the right to vote shall be deposited at the Office, or at such other place as is specified in accordance with these Articles for the deposit of instruments of proxy, at least 48 hours before the time appointed for holding the meeting or adjourned meeting at which the right to vote is to be exercised and, in default, the right to vote shall not be exercisable.
- 70.6 In the case of equality of votes, the chair of the meeting at which the show of hands takes place or at which the poll is demanded shall not be entitled to a casting vote.

### 71. No right to vote where sums overdue on shares

No member may vote at a general meeting (or any separate meeting of the holders of any class of shares), either in person or by proxy, or to exercise any other right or privilege as a member in respect of a share held by him or her unless:

- (a) all calls or other sums presently due and payable by him or her in respect of that share whether alone or jointly with any other person together with interest and expenses (if any) have been paid to the Company; or
- (b) the Board determines otherwise.

## 72. Voting by Proxy

- 72.1 In the case of a proxy relating to shares in the capital of the Company held in the name of a Depositary, the appointment of a proxy shall be in a form or manner of communication approved by the Board, which may include, without limitation, a voter instruction form to be provided to the Company by certain third parties on behalf of the Depositary.
- 52.2 Subject to Articles 72.1, an instrument appointing a proxy shall be in writing in any usual form (or in another form approved by the Board) executed under the hand of the appointor or his or her duly constituted attorney or, if the appointor is a corporation, under its seal or signed by a duly authorised officer or attorney or other person authorised to sign.
- 72.3 Subject to the Companies Acts, the Board may accept the appointment of a proxy received by electronic means on such terms and subject to such conditions as it considers fit. The appointment of a proxy received by electronic means shall not be subject to the requirements of Article 72.1.
- 72.4 For the purposes of Articles 72.1 and 72.3, the Board may require such reasonable evidence it considers necessary to determine:
  - (a) the identity of the member and the proxy; and
  - (b) where the proxy is appointed by a person acting on behalf of the member, the authority of that person to make the appointment.
- 72.5 A member may appoint another person as proxy to exercise all or any of his or her rights to attend and to speak and to vote on a resolution or amendment of a resolution, or on other business arising, at a meeting or meetings of the Company. Unless the contrary is stated in it, the appointment of a proxy shall be deemed to confer authority to exercise all such rights, as the proxy thinks fit.
- 72.6 A proxy need not be a member.

- A member may appoint more than one proxy in relation to a meeting, provided that each proxy is appointed to exercise the rights attached to different shares held by the member. When two or more valid but differing appointments of proxy are delivered or received for the same share for use at the same meeting, the one which is last validly delivered or received (regardless of its date or the date of its execution) shall be treated as replacing and revoking the other or others as regards that share. If the Company is unable to determine which appointment was last validly delivered or received, none of them shall be treated as valid in respect of that share.
- 72.8 Delivery or receipt of an appointment of proxy does not prevent a member attending and voting in person at the meeting or an adjournment of the meeting or on a poll.
- 72.9 The appointment of a proxy shall (unless the contrary is stated in it) be valid for an adjournment of the meeting as well as for the meeting or meetings to which it relates. The appointment of a proxy shall be valid for 12 months from the date of execution or, in the case of an appointment of proxy delivered by electronic means, for 12 months from the date of delivery unless otherwise specified by the Board.
- 72.10 Subject to the Companies Acts, the Company may send a form of appointment of proxy to all or none of the persons entitled to receive notice of and to vote at a meeting. If sent, the form shall provide for three-way voting on all resolutions (other than procedural resolutions) set out in the notice of meeting.

## 73. Receipt of proxy

- 73.1 An instrument appointing a proxy and any reasonable evidence required by the Board in accordance with Article 72.4 shall:
  - (a) subject to Articles 73.1(c) and 73.1(d), in the case of an instrument of proxy in hard copy form, delivered to the office, or another place in the United Kingdom specified in the notice convening the meeting or in the form of appointment of proxy or other accompanying document sent by the Company in relation to the meeting (a "**proxy notification address**") not less than 48 hours before the time for holding the meeting or adjourned meeting at which the person named in the form of appointment of proxy proposes to vote;
  - (b) subject to Articles 73.1(c) and 73.1(d), in the case of an appointment of a proxy sent by electronic means, where the Company has given an electronic address (a "proxy notification electronic address"):
    - (i) in the notice calling the meeting;
    - (ii) in an instrument of proxy sent out by the Company in relation to the meeting;
    - (iii) in an invitation to appoint a proxy issued by the Company in relation to the meeting; or
    - (iv) on a website maintained by or on behalf of the Company on which any information relating to the meeting is required by the Act to be kept, it shall be received at such proxy notification electronic address not less than 48 hours before the time for holding the meeting or adjourned meeting at which the person named in the form of appointment of proxy proposes to vote;
  - (c) in the case of a poll taken more than 48 hours after it is demanded, delivered or received at a proxy notification address or a proxy notification electronic address and not less than 24 hours before the time appointed for the holding of the adjourned meeting or the taking of the poll; or
  - (d) in the case of a poll which is not taken at the meeting at which it is demanded but is taken 48 hours or less after it is demanded, or in the case of an adjourned meeting to be held 48 hours or less after the time fixed for holding the original meeting, received:
    - (i) at a proxy notification address or a proxy notification electronic address in accordance with Articles 73.1(a) or (b);

- (ii) by the chair of the meeting or the secretary or any director at the meeting at which the poll is demanded or, as the case may be, at the original meeting; or
- (iii) at a proxy notification address or a proxy notification electronic address by such time as the chair of the meeting may direct at the meeting at which the poll is demanded. In calculating the periods in this Article, no account shall be taken of any part of a day that is not a working day.
- 73.2 The Board may decide, either generally or in any particular case, to treat a proxy appointment as valid notwithstanding that the appointment or any of the information required under Article 72.4 has not been received in accordance with the requirements of this Article.
- 73.3 Subject to Article 73.2, if the proxy appointment and any of the information required under Article 72.4 is not received in the manner set out in Article 73.1, the appointee shall not be entitled to vote in respect of the shares in question.
- 73.4 Without limitation, the Company may from time to time determine that any such electronic address may be used generally for such matters or specifically for particular meetings or purposes and, if so, the Company may provide different electronic addresses for different purposes. The Company may also impose any conditions on the transmission of and its receipt of such electronic communications including, for the avoidance of doubt, imposing any security or encryption arrangements as may be specified by the Company. If any document or information required to be sent to the Company is sent to the Company by electronic means under this Article, such document or information is not treated as validly delivered to or deposited with the Company if the same is not received by the Company at its designated electronic address in accordance with this Article or if no electronic address is so designated by the Company for the receipt of such document or information.

## 74. Revocation of proxy

A vote given or poll demanded by a proxy shall be valid in the event of the death or mental disorder of the principal or the revocation of the instrument of proxy, or of the authority under which the instrument of proxy was executed, or the transfer of the share for which the instrument of proxy is given, unless notice in writing of such death, mental disorder, revocation or transfer shall have been received by the Company at the Office, or at such other place as has been appointed for the deposit of instruments of proxy, no later than the last time at which an appointment of a proxy should have been received in order for it to be valid for use at the meeting or on the holding of the poll at which the vote was given or the poll taken.

## 75. Corporate representatives

- 75.1 A corporation (whether or not a company within the meaning of the Act) which is a member may, by resolution of its directors or other governing body, authorise such person as it thinks fit to act as its representative (or, as the case may be, representatives) at any meeting of the Company or at any separate meeting of the holders of any class of shares.
- 75.2 Any person so authorised shall be entitled to exercise the same powers on behalf of the corporation (in respect of that part of the corporation's holdings to which the authority relates) as the corporation could exercise if it were an individual member.
- 75.3 The corporation shall for the purposes of these Articles be deemed to be present in person and at any such meeting if a person so authorised is present at it, and all references to attendance and voting in person shall be construed accordingly.
- 75.4 A Director, the Secretary or some person authorised for the purpose by the Secretary may require the representative to produce a certified copy of the resolution so authorising him or her or such other evidence of his or her authority reasonably satisfactory to them before permitting him or her to exercise his or her powers.

75.5 A vote given or a poll demanded by a corporate representative shall be valid notwithstanding that the representative is no longer authorised to represent the member unless notice of the revocation of appointment was delivered in writing to the Company at such place or address and by such time as is specified in Article 74 for the revocation of the appointment of a proxy.

# 76. Failure to disclose interests in shares

- 76.1 If a member, or any other person appearing to be interested in shares held by that member, has been issued with a notice under section 793 of the Act (**section 793 notice**) and has failed in relation to any shares (**default shares**, which expression includes any shares issued after the date of such notice in right of those shares) to give the Company the information required by the section 793 notice within the prescribed period from the service of the notice, the following sanctions shall apply unless the Board determines otherwise:
  - (a) the member shall not be entitled in respect of the default shares to be present or to vote (either in person or by representative or proxy) at any general meeting or at any separate meeting of the holders of any class of shares or on any poll or to exercise any other right conferred by membership in relation to any such meeting or poll; and
  - (b) where the default shares represent at least 0.25% in nominal value of the issued shares of their class (calculated exclusive of any shares held as treasury shares):
    - (i) any dividend or other money payable for such shares shall be withheld by the Company, which shall not have any obligation to pay interest on it, and the member shall not be entitled to elect, pursuant to Article 131, to receive shares instead of that dividend; and
    - (ii) no transfer, other than an excepted transfer, of any shares held by the member shall be registered unless the member himself or herself is not in default of supplying the required information and the member proves to the satisfaction of the Board that no person in default of supplying such information is interested in any of the shares that are the subject of the transfer.
- 76.2 Where the sanctions under Article 76.1 apply in relation to any shares, they shall cease to have effect (and any dividends withheld under Article 76.1(b) shall become payable):
  - (a) if the shares are transferred by means of an excepted transfer but only in respect of the shares transferred; or
  - (b) at the end of the period of seven days (or such shorter period as the Board may determine) following receipt by the Company of the information required by the section 793 notice and the Board being fully satisfied that such information is full and complete.
- 76.3 Where, on the basis of information obtained from a member in respect of any share held by him or her, the Company issues a section 793 notice to any other person, it shall at the same time send a copy of the notice to the member, but the accidental omission to do so, or the non-receipt by the member of the copy, shall not invalidate or otherwise affect the application of Article 76.1.
- 76.4 For the purposes of this Article:
  - (a) a person, other than the member holding a share, shall be treated as appearing to be interested in that share if the member has informed the Company that the person is, or may be, so interested, or if the Company (after taking account of any information obtained from the member or, pursuant to a section 793 notice, from anyone else) knows or has reasonable cause to believe that the person is, or may be, so interested;
  - (b) **interested** shall be construed as it is for the purpose of section 793 of the Act;
  - (c) reference to a person having failed to give the Company the information required by a notice, or being in default as regards supplying such information, includes reference:
    - (i) to the person's having failed or refused to give all of any part of it; and

- (ii) to the person's having given information which he or she knows to be false in a material particular or having recklessly given information which is false in a material particular;
- (d) **prescribed period** means 14 days;
- (e) **excepted transfer** means, in relation to any shares held by a member:
  - (i) a transfer by way of or pursuant to acceptance of a takeover offer for the Company (within the meaning of section 974 of the Act); or
  - (ii) a transfer in consequence of a sale made through any stock exchange on which the Company's shares are normally traded; or
  - (iii) a transfer which is shown to the satisfaction of the Board to be made in consequence of a sale of the whole of the beneficial interest in the shares to a person who is unconnected with the member and with any other person appearing to be interested in the shares.
- 76.5 Nothing contained in this Article shall be taken to limit the powers of the Company under section 794 of the Act.
- 76.6 For the purpose of this Article 76:
  - (a) where any person appearing to be interested in any shares has been served with a section 793 notice and such shares are held by a Depositary, the provisions of this Article 76 shall be deemed to apply only to those shares held by the Depositary in which such person appears to be interested and not (so far as that person's apparent interest is concerned) to any other shares held by the Depositary in which such person does not have an interest and references to default shares shall be construed accordingly;
  - (b) where the shareholder on whom a section 793 notice has been served is a Depositary, the obligations of the Depositary (acting solely in the Depositary's capacity as such) shall be limited to disclosing to the Company such information relating to any person appearing to be interested in the shares held by it as has been recorded by the Depositary and the provision of such information shall be at the Company's cost.

### 77. Power of sale of shares of untraced members

- 77.1 The Company shall be entitled to sell at the best price reasonably obtainable any share of a member, or any share to which a person is entitled by transmission, if and provided that:
  - (a) during the period of 12 years before the date of sending of the notice referred to in Article 77.1(b) no cheque, order or warrant in respect of such share sent by the Company through the post in a pre-paid envelope addressed to the member or to the person entitled by transmission to the share, at his or her address on the Register or other last known address given by the member or person to which cheques, orders or warrants in respect of such share are to be sent has been cashed and the Company has received no communications in respect of such share from such member or person entitled, provided that during such period of 12 years the Company has paid at least three cash dividends (whether interim or final) and no such dividend has been claimed by the person entitled to it;
  - (b) on or after expiry of the said period of 12 years, the Company has given notice of its intention to sell such share by sending a notice to the member or person entitled by transmission to the share at his or her address on the Register or other last known address given by the member or person entitled by transmission to the share and before sending such a notice to the member or other person entitled by transmission, the Company must have used reasonable efforts to trace the member or other person entitled, engaging, if considered appropriate, a professional asset reunification company or other tracing agent and/or giving notice of its intention to sell the share by advertisement in a national newspaper and in a newspaper circulating in the area of the address of the member or person entitled by transmission to the share shown in the Register; and

- (c) during the further period of three months following the date of such notice and prior to the exercise of the power of sale the Company has not received any communication in respect of such share from the member or person entitled by transmission.
- 77.2 To give effect to any sale of shares under this Article, the Board may authorise some person to transfer the shares in question and may enter the name of the transferee in respect of the transferred shares in the Register even if no share certificate has been lodged for such shares and may issue a new certificate to the transferee. An instrument of transfer executed by that person shall be as effective as if it had been executed by the holder of, or the person entitled by transmission to, the shares. The buyer shall not be bound to see to the application of the purchase monies, nor shall his or her title to the shares be affected by any irregularity or invalidity in the proceedings in reference to the sale. If during the period of 12 years referred to in Article 77.1, or during any period ending on the date when all the requirements of Articles 77.1(a) to 77.1(c) have been satisfied, any additional shares have been issued in respect of those held at the beginning of, or previously so issued during, any such period and all the requirements of Articles 77.1(b) to 77.1(c)have been satisfied in regard to such additional shares, the Company shall also be entitled to sell the additional shares.

# 78. Application of proceeds of sale of shares of untraced members

The Company shall account to the member or other person entitled to the share for the net proceeds of a sale under Article 77 by carrying all monies relating to such sale to a separate account. The Company shall be deemed to be a debtor to, and not a trustee for, such member or other person in respect of such monies. Monies carried to such separate account may either be employed in the business of the Company or invested in such investments as the Board may think fit. No interest shall be payable to such member or other person in respect of such monies and the Company does not have to account for any money earned on them.

### 79. Number of directors

- 79.1 Unless otherwise determined by the Company by ordinary resolution, the number of Directors (other than any alternate Directors) shall be at least two and not more than fifteen. As at the Date of Adoption, the number of Directors shall be eight (8) (the "**Initial Directors**"), including a majority of Independent Directors.
- 79.2 For a period of three (3) years post-Closing a majority of the Directors shall be Independent Directors. However, if at any time the number of Independent Directors falls below a majority of the Board, that will not invalidate any resolution or other act of the Board pending the appointment of additional Independent Director(s).
- 79.3 Following the Closing, the Initial Directors shall be divided into three classes of Directors, designated as "Class II", "Class II" and "Class III", respectively (each a "Class"). The Board is authorized to assign members of the Board already in office to such classes at the time the classification becomes effective. The Board is also authorized to assign any persons who take office as Directors after the date hereof to any such Class; provided, however, that the Classes are as close to equal size as possible.
- 79.4 In the event of any increase in the number of Directors, the additional directorships resulting from such increase shall be apportioned by the Board among the Classes of Directors so as to maintain such Classes as nearly equal as possible. No decrease in the number of Directors shall shorten the term of any incumbent Director.
- 79.5 Notwithstanding the foregoing provisions, each Director shall serve until their successor is duly elected and qualified or until their earlier death, resignation or removal.

# 80. Power of company to appoint directors

Subject to these Articles and the Companies Acts, the Company may by ordinary resolution appoint a person who is willing to act to be a Director, either to fill a vacancy or as an addition to the existing Board

but the total number of Directors shall not exceed any maximum number fixed in accordance with these Articles.

# 81. Power of board to appoint directors

Subject to these Articles, the Board shall have power at any time to appoint any person who is willing to act as a Director, either to fill a vacancy or as an addition to the existing Board but the total number of Directors shall not exceed any maximum number fixed in accordance with these Articles.

# 82. Eligibility of new directors

- 82.1 No person, other than a retiring Director, shall be appointed or re-appointed a Director at any general meeting unless:
  - (a) he or she is recommended by the Board; or
  - (b) at least seven but not more than 42 clear days before the date appointed for the meeting the Company has received notice from a member (other than the person proposed) entitled to vote at the meeting of intention to propose a resolution for the appointment or re-appointment of that person, stating the particulars which would, if he or she were so appointed or re-appointed, be required to be included in the Company's register of directors and a notice executed by that person of his or her willingness to be appointed or re-appointed, is lodged at the Office.
- 82.2 For a period of three (3) years post-Closing no person, including a retiring Director, shall be appointed or re-appointed a Director by the Board or at any general meeting unless:
  - (a) he or she would, upon appointment, be an Independent Director; or
  - (b) immediately following that person's appointment as a Director, the Board would include a majority of Independent Directors.
- 82.3 In the event that an Independent Director is removed or required to resign pursuant to Article 84 or 85.1(c) (the "Removed Independent"), for a period of three (3) years post-Closing no person shall be appointed or re-appointed as a replacement for the Removed Independent Director by the Board unless the Board approves such appointment of an Independent Director by simple majority including the affirmative vote of at least two (2) Independent Directors (or if there are fewer than two Independent Directors then in office, all of the Independent Directors, if any).
- 82.4 A Director need not be a member of the Company.

#### 83. Retirement of directors

- 83.1 At the first Annual General Meeting of the Company following Closing, each Director in Class I shall retire from office but shall be eligible for re-appointment by ordinary resolution of the Company at such Annual General Meeting and, in each case, where such Director is so re-appointed, they shall be entitled to serve until the third Annual General Meeting of the Company falling after the first Annual General Meeting, at which stage the Director shall retire from office but shall be eligible for further re-appointment.
- At the second Annual General Meeting of the Company following Closing, each Director in Class II shall retire from office but shall be eligible for re-appointment by ordinary resolution of the Company at such Annual General Meeting and, in each case, where such Director is so re-appointed, they shall be entitled to serve until the third Annual General Meeting of the Company falling after the second Annual General Meeting, at which stage the Director shall retire from office but shall be eligible for further re-appointment.
- 83.3 At the third Annual General Meeting of the Company following Closing, each Director in Class III shall retire from office but shall be eligible for re-appointment by ordinary resolution of the Company at such

Annual General Meeting and, in each case, where such Director is so re-appointed, they shall be entitled to serve until the third Annual General Meeting of the Company falling after the third Annual General Meeting, at which stage the Director shall retire from office but shall be eligible for further re-appointment.

- 83.4 At each succeeding Annual General Meeting of the Company following the third Annual General Meeting of the Company following Closing, Directors shall be elected to serve for a term of three years to succeed the Directors of the class whose terms expire at such Annual General Meeting.
- 83.5 Subject to the provisions of these Articles, a Director shall remain a member of the class of directors to which he or she was assigned in accordance with Article 70.5. The initial terms of each class of directors shall expire as set forth in this Article 83, subject to such director's earlier death, resignation, disqualification or removal.
- 83.6 Where a Director retires at an Annual General Meeting in accordance with Article 83.1, 83.2, 83.3 or otherwise, the Company may at the meeting by ordinary resolution fill the office being vacated by electing the retiring Director. In the absence of such a resolution, the retiring Director shall nevertheless be deemed to have been re-elected except in any of the following cases:
  - (a) where at such meeting a resolution of the Company for the re-election of such Director is put to the meeting and lost;
  - (b) where such Director is ineligible for re-election or has given notice in writing to the Company that he/she is unwilling to be re-elected; or
  - (c) where a resolution of the Company to elect such Director is void by reason of contravention of section 160 of the Act (whereby at a General Meeting a motion for the appointment of two or more persons as Directors by a single resolution must not be made unless a resolution that it should be made has first been agreed to by the meeting without any vote being given against it).
- 83.7 The retirement shall not have effect until the conclusion of the meeting except where a resolution of the Company is passed to elect some other person in the place of the retiring Director or a resolution for the retiring Director's re-election is put to the meeting and lost. Accordingly a retiring Director who is re-elected or deemed to have been re-elected will continue in office without a break.
- 83.8 Without limiting the generality of these Articles, a person ceases to be a Director if that person:
  - (a) ceases to be a director under the Companies Act or applicable winding up laws or regulations, or is prohibited from being a director by applicable laws or regulations;
  - (b) becomes bankrupt or makes any arrangement or composition with the person's creditors generally;
  - (c) becomes a mentally incapacitated person;
  - (d) resigns the office of director by notice in writing of the resignation in accordance with the Companies Act; or
  - (e) for more than 6 months has been absent without the directors' permission from directors' meetings held during that period.

## 84. Removal of directors

84.1 Subject to Article 84.2, in addition to any power of removal conferred by the Companies Acts, the Company may by special resolution, or by ordinary resolution of which special notice has been given in accordance with section 312 of the Act, remove a director before the expiry of his or her period of office (without prejudice to a claim for damages for breach of contract or otherwise) and may (subject to these Articles) by ordinary resolution appoint another person who is willing to act to be a director in his or her place.

84.2 In the first three (3) years following Closing, save as required by the Companies Acts the Board may only convene a general meeting which proposes a resolution (pursuant to Article 84.1 or otherwise) to remove an Independent Director if the Board approves such resolution by simple majority including the affirmative vote of at least two (2) other Independent Directors (or if there are fewer than two Independent Directors then in office excluding the Independent Director proposed to be removed, all of the Independent Directors (other than the Independent Director proposed to be removed), if any).

# 85. Vacation of office by director

- 85.1 Without prejudice to the provisions for retirement contained in these Articles, the office of a Director shall be vacated if:
  - the director resigns by notice in writing delivered to the Secretary at the Office or at an address specified by the Company for the purposes of communication by electronic means or tendered at a Board meeting;
  - (b) the director offers to resign by notice in writing delivered to the Secretary at the Office or at an address specified by the Company for the purposes of communication by electronic means or tendered at a Board meeting and the Board resolves to accept such offer;
  - (c) the Director is requested to resign by a majority of the other Directors by notice in writing addressed to him or her at his or her address as shown in the register of Directors (without prejudice to any claim for damages which the Director may have for breach of any contract between him or her and the Company), provided that in the first three (3) years following Closing this power may only be invoked in respect of an Independent Director if the majority of the other Directors which gives notice includes at least two (2) other Independent Directors (or if there are fewer than two Independent Directors then in office, all of the Independent Directors, if any);
  - (d) the Director ceases to be a Director by virtue of any provision of the Companies Acts, is removed from office pursuant to these Articles or the Act or becomes prohibited by law from being a Director;
  - (e) the Director becomes bankrupt or makes an arrangement or composition with his or her creditors generally;
  - (f) a registered medical practitioner who is treating the Director gives a written opinion to the Company stating he or she has become physically or mentally incapable of acting as a director and may remain so for more than three months, or is or has been suffering from mental or physical ill health and the Board resolves that his or her office be vacated; or
  - (g) the Director is absent (whether or not any alternate Director appointed by the Director attends), without the permission of the Board, from Board meetings for six consecutive months and a notice is served on the Director personally, or at his or her residential address provided to the Company under section 165 of the Act signed by all the other Directors stating that he or she shall cease to be a Director with immediate effect (and such notice may consist of several copies each signed by one or more Directors).
- 85.2 If the office of a Director is vacated for any reason, he or she shall cease to be a member of any committee or sub-committee of the Board.

# 86. Resolution as to vacancy conclusive

A resolution of the Board declaring a Director to have vacated office under the terms of Article 85 shall be conclusive as to the fact and ground of vacation stated in the resolution.

# 87. Appointment of alternate directors

87.1 Each Director may appoint any person (including another Director) to be his or her alternate and may at his or her discretion remove an alternate Director so appointed. Any appointment or removal of an alternate

Director must be by written notice delivered to the Office or at an address specified by the Company for the purposes of communication by electronic means or tendered at a Board meeting or in any other manner approved by the Board. The appointment requires the approval of the Board unless it has been previously approved or the appointee is another Director.

87.2 An alternate Director must provide the particulars, and sign any form for public filing required by the Companies Acts relating to his or her appointment.

## 88. Alternate directors' participation in board meetings

- 88.1 Every alternate Director is (subject to his or her giving to the Company an address [within the United Kingdom] at which notices may be served on him or her (and, if applicable, an address in relation to which electronic communications may be received)) entitled to receive notice of all meetings of the Board and all committees of the Board of which his or her appointor is a member and, in the appointor's absence, to attend and vote at such meetings and to exercise all the powers, rights, duties and authorities of the appointor. Each person acting as an alternate Director shall have a separate vote at Board meetings for each Director for whom that person acts as alternate Director in addition to his or her own vote if also a Director, but shall count as only one for the purpose of determining whether a quorum is present.
- 88.2 Signature by an alternate Director of any resolution in writing of the Board or a committee of the Board will, unless the notice of appointment provides otherwise, be as effective as signature by his or her appointor.

### 89. Alternate director responsible for own acts

Each person acting as an alternate Director will be an officer of the Company, will alone be responsible to the Company for his or her own acts and defaults and will not be deemed to be the agent of the Director appointing them.

## 90. Interests of alternate director

An alternate Director is entitled to contract and be interested in and benefit from contracts or arrangements with the Company, to be repaid expenses and to be indemnified to the same extent as if he or she were a Director. However, no alternate Director is entitled to receive from the Company any fees for his or her services as alternate, except such part (if any) of the fee payable to the alternate's appointor as such appointor may by written notice to the Company direct.

### 91. Revocation of alternate director

An alternate Director will cease to be an alternate Director:

- (a) if the alternate's appointor revokes his or her appointment; or
- (b) if the alternate resigns his or her office by notice in writing to the Company; or
- (c) if the alternate's appointor ceases for any reason to be a Director, provided that if any Director retires but is re-appointed or deemed to be re-appointed at the same meeting, any valid appointment of an alternate Director which was in force immediately before his or her retirement shall remain in force; or
- (d) if any event happens in relation to the alternate which, if the alternate were a Director otherwise appointed, would cause him or her to vacate office.

## 92. Directors' fees

Each of the Directors may be paid a fee at such rate as may from time to time be decided by ordinary resolution of the Company. Any fees payable under this Article shall be distinct from any salary,

remuneration or other amounts payable to a Director under any other provisions of these Articles and shall accrue from day to day.

# 93. Expenses

Each Director may be paid reasonable travelling, hotel and other expenses properly incurred by him or her in or about the performance of their duties as Director, including any expenses incurred in attending meetings of the Board or any committee of the Board or general meetings or separate meetings of the holders of any class of shares or debentures of the Company. Subject to the Act, the Directors shall have the power to make arrangements to provide a Director with funds to meet expenditure incurred or to be incurred by him or her for the purposes of the Company or for the purpose of enabling him or her to perform his or her duties as an officer of the Company or to enable him or her to avoid incurring any such expenditure.

#### 94. Additional remuneration

If by arrangement with the Board any Director shall perform or render any special duties or services outside his or her ordinary duties as a Director and not in his or her capacity as a holder of employment or executive office, he or she may be paid such reasonable additional remuneration (whether by way of salary, commission, participation in profits or otherwise) as the Board may determine.

#### 95. Remuneration of executive directors

The salary or remuneration of any Director appointed to hold any employment or executive office in accordance with these Articles may be either a fixed sum of money, or may altogether or in part be governed by business done or profits made or otherwise determined by the Board, and may be in addition to or instead of any fee payable to him or her for serving as Director under these Articles.

#### 96. Pensions and other benefits

- 96.1 The Board may exercise all the powers of the Company to provide pensions or other retirement or superannuation benefits and to provide death or disability benefits or other allowances or gratuities (whether by insurance or otherwise) for any person who is or has at any time been a Director or employee of:
  - (a) the Company;
  - (b) any company which is or was a holding company or a subsidiary undertaking of the Company;
  - (c) any company which is or was allied to or associated with the Company or a subsidiary undertaking or holding company of the Company; or
  - (d) a predecessor in business of the Company or of any holding company or subsidiary undertaking of the Company. and, in each case, for any member of his or her family (including a spouse or former spouse) and any person who is or was dependent on him or her.
- 96.2 The Board may establish, maintain, subscribe and contribute to any scheme, institution, association, club, trust or fund and pay premiums and, subject to the Companies Acts, lend money or make payments to, guarantee or give an indemnity in respect of, or give any financial or other assistance in connection with any of the matters set out in Article 92.1 above. The Board may procure any of such matters to be done by the Company either alone or in conjunction with any other person. Any Director or former Director shall be entitled to receive and retain for his or her own benefit any pension or other benefit provided under this Article and shall not have to account for it to the Company. The receipt of any such benefit will not disqualify any person from being or becoming a Director of the Company.

#### 97. Powers of the board

- 97.1 Subject to the Companies Acts, these Articles and to any directions given by special resolution of the Company, the business of the Company will be managed by the Board, which may exercise all the powers of the Company, whether relating to the management of the business or not.
- 97.2 No alteration of these Articles and no such direction given by the Company shall invalidate any prior act of the Board which would have been valid if such alteration had not been made or such direction had not been given. Provisions contained elsewhere in these Articles as to any specific power of the Board shall not be deemed to limit the general powers given by this Article.

## 98. Powers of directors if less than minimum number

If the number of Directors is less than the minimum prescribed in Article 79 or decided by the Company by ordinary resolution, the remaining Director or Directors may act only for the purposes of appointing an additional Director or Directors to make up that minimum or convening a general meeting of the Company for the purpose of making such appointment. If no Director or Directors is or are able or willing to act, two members may convene a general meeting for the purpose of appointing Directors. An additional Director appointed in this way holds office (subject to these Articles) only until the dissolution of the next annual general meeting after his or her appointment unless reappointed during the annual general meeting.

### 99. Powers of executive directors

The Board or any committee authorised by the Board may:

- (a) delegate or entrust to and confer on any Director holding executive office (including a Chief Executive or Managing Director) such of its powers, authorities and discretions (with power to sub-delegate) for such time, on such terms and subject to such conditions as it thinks fit; and
- (b) revoke, withdraw, alter or vary all or any of such powers.

## 100. Delegation to committees

- 100.1 The Board may delegate any of its powers, authorities and discretions (with power to sub-delegate) for such time on such terms and subject to such conditions as it thinks fit to any committee consisting of one or more Directors and (if thought fit) one or more other persons provided that:
  - (a) a majority of the members of a committee shall be Directors; and
  - (b) no resolution of a committee shall be effective unless a majority of those present when it is passed are Directors or alternate Directors.
- 100.2 The Board may confer such powers either collaterally with, or to the exclusion of and in substitution for, all or any of the powers of the Board in that respect and may revoke, withdraw, alter or vary any such powers and discharge any such committee in whole or in part. Insofar as any power, authority or discretion is so delegated, any reference in these Articles to the exercise by the Board of such power, authority or discretion shall be construed as if it were a reference to the exercise of such power, authority or discretion by such committee.

## 101. Local management

- 101.1 The Board may establish any local or divisional boards or agencies for managing any of the affairs of the Company in any specified locality, either in the United Kingdom or elsewhere, and appoint any persons to be members of such local or divisional board, or any managers or agents, and may fix their remuneration.
- 101.2 The Board may delegate to any local or divisional board, manager or agent so appointed any of its powers, authorities and discretions (with power to sub-delegate) and may authorise the members of any such local

or divisional board, or any of them, to fill any vacancies and to act notwithstanding vacancies. Any such appointment or delegation under this Article may be made, on such terms conditions as the Board may think fit. The Board may confer such powers either collaterally with, or to the exclusion of and in substitution for, all or any of the powers of the Board in that respect and may revoke, withdraw, alter or vary all or any of such powers.

101.3 Subject to any terms and conditions expressly imposed by the Board, the proceedings of any local or divisional board or agency with two or more members shall be governed by such of these Articles as regulate the proceedings of the Board, so far as they are capable of applying.

### 102. Power of attorney

The Board may, by power of attorney or otherwise, appoint any person or persons to be the agent or attorney of the Company and may delegate to any such person or persons any of its powers, authorities and discretions (with power to sub-delegate), in each case for such purposes and for such time, on such terms (including as to remuneration) and conditions as it thinks fit. The Board may confer such powers either collaterally with, or to the exclusion of and in substitution for, all or any of the powers of the Board in that respect and may revoke, withdraw, alter or vary any of such powers.

### 103. Exercise of voting power

The Board may exercise or cause to be exercised the voting power conferred by the shares in any other company held or owned by the Company, or any power of appointment to be exercised by the Company, in such manner as it thinks fit (including the exercise of the voting power or power of appointment in favour of the appointment of any Director as a director or other officer or employee of such company or in favour of the payment of remuneration to the directors, officers or employees of such company).

# 104. Provision for employees on cessation of business

The Board may, by resolution, sanction the exercise of the power to make provision for the benefit of persons employed or formerly employed by the Company or any of its subsidiary undertakings, in connection with the cessation or the transfer to any person of the whole or part of the undertaking of the Company or that subsidiary undertaking, but any such resolution shall not be sufficient for payments to or for the benefit of directors, former directors or shadow directors.

## 105. Overseas registers

Subject to the Companies Acts, the Company may keep an overseas, local or other register and the Board may make and vary such regulations as it thinks fit respecting the keeping of any such register.

## 106. Borrowing powers

- 106.1 Subject to these Articles and the Companies Acts, the Board may exercise all the powers of the Company to:
  - (a) borrow money;
  - (b) indemnify and guarantee;
  - (c) mortgage or charge all or any part of the undertaking, property and assets (present and future) and uncalled capital of the Company;
  - (d) create and issue debentures and other securities; and
  - (e) give security either outright or as collateral security for any debt, liability or obligation of the Company or of any third party.

#### 107. Board meetings

- 107.1 The Board can decide when and where to have meetings and how they will be conducted. They may also adjourn meetings.
- 107.2 A Board meeting can be called by any Director. The Secretary must call a Board meeting if asked to do so by a Director.

### 108. Notice of board meetings

- 108.1 Notice of a Board meeting shall be deemed to be duly given to a Director if it is given to the Director personally or by word of mouth or given in writing or by electronic means to the Director at his or her last known address or any other address given by him or her to the Company for that purpose.
- 108.2 A Director may waive the requirement that notice be given to him or her of any Board meeting, either prospectively or retrospectively and any retrospective waiver shall not affect the validity of the meeting or of any business conducted at the meeting.
- 108.3 It shall not be necessary to give notice of a Board meeting to a Director who is absent [from the United Kingdom] unless the Director has asked the Board in writing that notices of Board meetings shall during his or her absence be given to him or her at any address [in the United Kingdom] notified to the Company for this purpose, but the Director shall not, in such event, be entitled to a longer period of notice than if he or she had been present [in the United Kingdom] at that address.

### 109. Quorum

- 109.1 The quorum necessary for the transaction of business may be determined by the Board and until otherwise determined shall be three persons, each being a Director. A duly convened meeting of the Board at which a quorum is present shall be competent to exercise all or any of the authorities, powers, and discretions for the time being vested in or exercisable by the Board.
- 109.2 If a Director ceases to be a director at a Board meeting, he or she can continue to be present and to act as a director and be counted in the quorum until the end of the meeting if no other Director objects and if otherwise a quorum of Directors would not be present.
- 109.3 If, within thirty (30) minutes from the time appointed for the meeting of the Directors a quorum be not present, the meeting, if convened upon requisition in accordance with the Companies Act, shall be dissolved; but in any other case it shall stand adjourned to the same day in the next week at the same time and place, or to such other day, time and place as the majority of the Directors may agree. If at such adjourned meeting of the Directors a quorum be not present within thirty (30) minutes from the time appointed for the adjourned meeting, any three Directors present at such adjourned meeting shall be a quorum and may transact the business for which the meeting is called.

## 110. Chair

- 110.1 The Board may appoint one or more of its body as chair or joint chair and one or more of its body as deputy chair of its meetings and may determine the period for which he or she is or they are to hold office and may at any time remove him, her or them from office.
- 110.2 If no such chair or deputy chair is elected, or if at any meeting neither a chair nor a deputy chair is present within ten minutes of the time appointed for holding the same, the Directors present shall choose one of their number to be chair of such meeting. In the event two or more joint chairs or, in the absence of a chair, two or more deputy chairs being present, the joint chair or deputy chair to act as chair of the meeting shall be decided by those Directors present.

## 111. Voting

Questions arising at any Board meeting shall be determined by a majority of votes. In the case of an equality of votes the chair of that meeting shall have a second or casting vote (unless he or she is not entitled to vote on the resolution in question).

### 112. Participation by telephone or other form of communication

- 112.1 Any Director or his or her alternate may validly participate in a meeting of the Board or a committee of the Board through the medium of conference telephone or any other form of communications equipment (whether in use when these Articles are adopted or developed subsequently), provided that all persons participating in the meeting are able to hear and speak to each other throughout such meeting.
- 112.2 A person so participating by telephone or other communication shall be deemed to be present in person at the meeting and shall be counted in a quorum and entitled to vote. Such a meeting shall be deemed to take place where the largest group of those participating is assembled or, if there is no group which is larger than any other group, where the chair of the meeting then is.
- 112.3 A resolution passed at any meeting held in the above manner, and signed by the chair of the meeting, shall be as valid and effectual as if it had been passed at a meeting of the Board (or committee, as the case may be) duly convened and held.

## 113. Resolution in writing

- 113.1 A resolution in writing signed or confirmed electronically by all the Directors for the time being entitled to receive notice of a Board meeting and to vote on the resolution and not being less than a quorum (or by all the members of a committee of the Board for the time being entitled to receive notice of such committee meeting and to vote on the resolution and not being less than a quorum of that committee), shall be as valid and effective for all purposes as a resolution duly passed at a meeting of the Board (or committee, as the case may be).
- 113.2 Such a resolution may consist of several documents or electronic communications in the same form each signed or authenticated by one or more of the Directors or members of the relevant committee.

## 114. Proceedings of committees

All committees of the Board shall, in the exercise of the powers delegated to them and in the transaction of business, conform with any mode of proceedings and regulations which the Board may prescribe and subject to this shall be governed by such of these Articles as regulate the proceedings of the Board as are capable of applying.

## 115. Minutes of proceedings

- 115.1 The Board shall keep minutes of all shareholder meetings, all Board meetings and meetings of committees of the Board. The minutes must include the names of the Directors present.
- 115.2 Any such minutes, if purporting to be signed by the chair of the meeting at which the proceedings were held or by the chair of the next meeting or the Secretary, shall be evidence of the matters stated in such minutes without any further proof.

## 116. Validity of proceedings

All acts done by a meeting of the Board, or of a committee of the Board, or by any person acting as a Director, alternate Director or member of a committee shall be valid even if it is discovered afterwards that there was some defect in the appointment of any person or persons acting, or that they or any of them were or was disqualified from holding office or not entitled to vote, or had in any way vacated office.

#### 117. Transactions or other arrangements with the company

- 117.1 Subject to the Companies Acts and provided he or she has declared the nature and extent of his or her interest in accordance with the requirements of the Companies Acts, a Director who is in any way, whether directly or indirectly, interested in an existing or proposed transaction or arrangement with the Company may:
  - be a party to, or otherwise interested in, any transaction or arrangement with the Company or in which the Company is otherwise (directly or indirectly) interested;
  - (b) act by himself or herself or through his or her firm in a professional capacity for the Company (otherwise than as auditor) and he or her, or his or her firm, shall be entitled to remuneration for professional services as if he or she were not a Director;
  - (c) be or become a director or other officer of, or employed by, or a party to a transaction or arrangement with, or otherwise interested in, any body corporate in which the Company is otherwise (directly or indirectly) interested; and
  - (d) hold any office or place of profit with the Company (except as auditor) in conjunction with his or her office of Director for such period and upon such terms, including as to remuneration as the Board may decide.
- 117.2 A Director shall not, save as he or she may otherwise agree, be accountable to the Company for any benefit which he or she derives from any such contract, transaction or arrangement or from any such office or employment or from any interest in any such body corporate and no such contract, transaction or arrangement shall be liable to be avoided on the grounds of any such interest or benefit nor shall the receipt of any such remuneration or other benefit constitute a breach of duty under section 176 of the Act.

# 118. Authorisation of Directors' conflicts of interest

- 118.1 The Board may, in accordance with the requirements set out in this Article, authorise any matter or situation proposed to them by any Director which would, if not authorised, involve a Director (an **Interested Director**) breaching his or her duty under the Act to avoid conflicts of interest.
- 118.2 A Director seeking authorisation in respect of a conflict of interest shall declare to the Board the nature and extent of his or her interest in a conflict of interest as soon as is reasonably practicable. The Director shall provide the Board with such details of the matter as are necessary for the Board to decide how to address the conflict of interest together with such additional information as may be requested by the Board.
- 118.3 Any authorisation under this Article will be effective only if:
  - (a) to the extent permitted by the Act, the matter in question shall have been proposed by any Director for consideration in the same way that any other matter may be proposed to the Directors under the provisions of these Articles;
  - (b) any requirement as to the quorum for consideration of the relevant matter is met without counting the Interested Director and any other interested Director; and
  - (c) the matter is agreed to without the Interested Director voting or would be agreed to if the Interested Director's and any other interested Director's vote is not counted.
- 118.4 Any authorisation of a conflict of interest under this Article must be recorded in writing (but the authority shall be effective whether or not the terms are so recorded) and may (whether at the time of giving the authorisation or subsequently):
  - (a) extend to any actual or potential conflict of interest which may reasonably be expected to arise out of the matter or situation so authorised;

- (b) provide that the Interested Director be excluded from the receipt of documents and information and the participation in discussions (whether at meetings of the Directors or otherwise) related to the conflict of interest;
- (c) impose upon the Interested Director such other terms for the purposes of dealing with the conflict of interest as the Directors think fit:
- (d) provide that, where the Interested Director obtains, or has obtained (through his or her involvement in the conflict of interest and otherwise than through the interested Director's position as a Director) information that is confidential to a third party, he or she will not be obliged to disclose that information to the Company, or to use it in relation to the Company's affairs where to do so would amount to a breach of that confidence; and
- (e) permit the Interested Director to absent himself or herself from the discussion of matters relating to the conflict of interest at any meeting of the Directors and be excused from reviewing papers prepared by, or for, the Directors to the extent they relate to such matters.
- 118.5 Where the Directors authorise a conflict of interest, the Interested Director will be obliged to conduct himself or herself in accordance with any terms and conditions imposed by the Directors in relation to the conflict of interest.
- 118.6 The Directors may revoke or vary such authorisation at any time, but this will not affect anything done by the Interested Director, prior to such revocation or variation, in accordance with the terms of such authorisation.
- 118.7 A Director is not required, by reason of being a Director (or because of the fiduciary relationship established by reason of being a director), to account to the Company for any remuneration, profit or other benefit which he or she derives from or in connection with a relationship involving a conflict of interest which has been authorised by the directors or by the Company in general meeting (subject in each case to any terms, limits or conditions attaching to that authorisation) and no contract shall be liable to be avoided on such grounds.

#### 119. Directors' permitted interests

- 119.1 A Director cannot vote or be counted in the quorum on any resolution relating to any transaction or arrangement with the Company in which the Director has an interest and which may reasonably be regarded as likely to give rise to a conflict of interest but can vote (and be counted in the quorum) on the following:
  - (a) any security, guarantee or indemnity for any money or any liability which the Director, or any other person, has lent or obligations the Director or any other person has undertaken at the request, or for the benefit, of the Company or any of its subsidiary undertakings;
  - (b) any security, guarantee or indemnity to any other person for a debt or obligation which is owed by the Company or any of its subsidiary undertakings, to that other person if the Director has taken responsibility for some or all of that debt or obligation. The Director can take this responsibility by giving a guarantee, indemnity or security;
  - (c) a proposal or contract relating to an offer of any shares or debentures or other securities for subscription or purchase by the Company or any of its subsidiary undertakings, if the Director takes part because he or she is a holder of shares, debentures or other securities, or if he or she takes part in the underwriting or sub-underwriting of the offer;
  - (d) any arrangement for the benefit of employees of the Company or any of its subsidiary undertakings which only gives him or her benefits which are also generally given to employees to whom the arrangement relates;

- (e) any arrangement involving any other company if the Director (together with any person connected with the Director) has an interest of any kind in that company (including an interest by holding any position in that company or by being a shareholder of that company). This does not apply if he or she knows that he has a Relevant Interest.
- (f) a contract relating to insurance which the Company can buy or renew for the benefit of the Directors or a group of people which includes Directors; and
- (g) a contract relating to a pension, superannuation or similar scheme or a retirement, death, disability benefits scheme or employees' share scheme which gives the Director benefits which are also generally given to the employees to whom the scheme relates.
- 119.2 A Director cannot vote or be counted in the quorum on a resolution relating to the Director's own appointment or the settlement or variation of the terms of his or her appointment to an office or place of profit with the Company or any other company in which the Company has an interest.
- 119.3 Where the Directors are considering proposals about the appointment, or the settlement or variation of the terms or the termination of the appointment of two or more Directors to other offices or places of profit with the Company or any company in which the Company has an interest, a separate resolution may be put in relation to each Director and in that case each of the Directors concerned shall be entitled to vote and be counted in the quorum in respect of each resolution unless it concerns his or her own appointment or the settlement or variation of the terms or the termination of his or her own appointment or the appointment of another director to an office or place of profit with a company in which the Company has an interest and the Director seeking to vote or be counted in the quorum has a Relevant Interest in it.
- 119.4 A company shall be deemed to be one in which the Director has a **Relevant Interest** if and so long as (but only if and so long as) the Director is to his or her knowledge (either directly or indirectly) the holder of or beneficially interested in 1% or more of any class of the equity share capital of that company (calculated exclusive of any shares of that class in that company held as treasury shares) or of the voting rights available to members of that company. In relation to an alternate Director, an interest of his or her appointor shall be treated as an interest of the alternate Director without prejudice to any interest which the alternate Director has otherwise. Where a company in which a Director has a Relevant Interest is interested in a contract, the Director also shall be deemed interested in that contract.
- 119.5 If a question arises at a Board meeting about whether a Director (other than the chair of the meeting) has an interest which is likely to give rise to a conflict of interest, or whether he or she can vote or be counted in the quorum, and the Director does not agree to abstain from voting on the issue or not to be counted in the quorum, the question must be referred to the chair of the meeting. The chair's ruling about the relevant Director is final and conclusive, unless the nature and extent of the Director's interests have not been fairly disclosed to the Directors. If the question arises about the chair of the meeting, the question must be directed to the Directors. The chair cannot vote on the question but can be counted in the quorum. The Directors' resolution about the chair is final and conclusive, unless the nature and extent of the chair's interests have not been fairly disclosed to the Directors.

# 120. General

For the purposes of Articles 117 to 119 inclusive (which shall apply equally to alternate Directors):

- 120.1 An interest of a person who is connected (which word shall have the meaning given to it by section 252 of the Act) with a Director shall be treated as an interest of the Director.
- 120.2 A contract includes references to any proposed contract and to any transaction or arrangement or proposed transaction or arrangement whether or not constituting a contract.
- 120.3 A conflict of interest includes a conflict of interest and duty and a conflict of duties.

120.4 Subject to the Companies Acts, the Company may by ordinary resolution suspend or relax the provisions of Articles 117 to 119 to any extent or ratify any contract not properly authorised by reason of a contravention of any of the provisions of Articles 117 to 119.

## 121. Power to authenticate documents

Any Director, the Secretary or any person appointed by the Board for the purpose shall have power to authenticate any documents affecting the constitution of the Company and any resolution passed by the Company or the Board or any committee, and any books, records, documents and accounts relating to the business of the Company, and to certify copies or extracts as true copies or extracts. Where any books, records, documents or accounts are not at the Office, the local manager or other officer of the Company who has their custody shall be deemed to be a person appointed by the Board for this purpose. A document purporting to be a copy of a resolution, or an extract from the minutes of a meeting, of the Company or the Board or any committee which is so certified shall be conclusive evidence in favour of all persons dealing with the Company that such resolution has been duly passed or, as the case may be, that any minute so extracted is a true and accurate record of proceedings at a duly constituted meeting.

### 122. Use of seals

- 122.1 The Board shall provide for the safe custody of the Seal. A Seal shall not be used without the authority of the Board or of a committee of the Board so authorised.
- 122.2 Subject as otherwise provided in these Articles, every document which is sealed using the Seal must be signed by at least one authorised person in the presence of a witness who attests the signature. An authorised person for this purpose is any Director, the Secretary or any other person authorised by the Directors for the purpose of signing documents to which the Seal is applied.
- 122.3 The Seal shall be used only for sealing securities issued by the Company and documents creating or evidencing securities so issued. Any such securities or documents sealed with the Seal shall not require to be signed unless the Board decides otherwise or the law otherwise requires.
- 122.4 The Board may decide who will sign an instrument to which a Seal is affixed (or in the case of a share certificate, on which the Seal may be printed) either generally or in relation to a particular instrument or type of instrument and may also determine either generally or in a particular case that a signature may be dispensed with or affixed by mechanical means.

## 123. Declaration of dividends

Subject to the Act and these Articles, the Company may by ordinary resolution declare dividends to be paid to members according to their respective rights and interests in the profits of the Company. However, no dividend shall exceed the amount recommended by the Board.

# 124. Interim dividends

Subject to the Act, the Board may declare and pay such interim dividends (including any dividend at a fixed rate) as appears to the Board to be justified by the profits of the Company available for distribution. If the Board acts in good faith, it shall not incur any liability to the holders of shares for any loss that they may suffer by the lawful payment of any interim dividend on any other class of shares ranking with or after those shares.

### 125. Calculation and currency of dividends

Except as provided otherwise by the rights attached to shares, all dividends:

(a) shall be declared and paid accordingly to the amounts paid up (otherwise than in advance of calls) on the shares on which the dividend is paid;

- (b) shall be apportioned and paid proportionately to the amounts paid up on the shares during any portion or portions of the period in respect of which the dividend is paid, but if any share is issued on terms that it shall rank for dividend as from a particular date, it shall rank for dividend accordingly; and
- (c) may be declared or paid in any currency. The Board may decide the rate of exchange for any currency conversions that may be required and how any costs involved are to be met.

#### 126. Amounts due on shares can be deducted from dividends

The Board may deduct from any dividend or other money payable to any person on or in respect of a share all such sums as may be due from him or her to the Company on account of calls or otherwise in relation to the shares of the Company. Sums so deducted can be used to pay amounts owing to the Company in respect of the shares.

#### 127. Dividends not in cash

The Board may, by ordinary resolution of the Company direct, or in the case of an interim dividend may without the authority of an ordinary resolution direct, that payment of any dividend declared may be satisfied wholly or partly by the distribution of assets, and in particular of paid up shares or debentures of any other company, or in any one or more of such ways. Where any difficulty arises regarding such distribution, the Board may settle it as it thinks fit. In particular, the Board may:

- (a) issue fractional certificates (or ignore fractions);
- (b) fix the value for distribution of such assets or any part of them and determine that cash payments may be made to any members on the footing of the values so fixed, in order to adjust the rights of members; and
- (c) vest any such assets in trustees on trust for the person entitled to the dividend.

#### 128. No interest on dividends

Unless otherwise provided by the rights attached to the share, no dividend or other monies payable by the Company or in respect of a share shall bear interest as against the Company.

### 129. Method of payment

- 129.1 The Company may pay any dividend, interest or other sum payable in respect of a share in cash or by direct debit, bank transfer, cheque, dividend warrant, or money order or by any other method, including by electronic means, as the Board may consider appropriate.
- 129.2 The Company may send such payment by post or other delivery service (or by such means offered by the Company as the member or person entitled to it may agree in writing) to the registered address of the member or person entitled to it (or, if two or more persons are holders of the share or are jointly entitled to it because of the death or bankruptcy of the member or otherwise by operation of law, to the registered address of such of those persons as is first named in the Register) or to such person and such address as such member or person may direct in writing.
- 129.3 Every cheque, warrant, order or other form of payment is sent at the risk of the person entitled to the money represented by it, shall be made payable to the person or persons entitled, or to such other person as the person or persons entitled may direct in writing. Payment of the cheque, warrant, order or other form of payment (including transmission of funds through a bank transfer or other funds transfer system or by such other electronic means as permitted by these Articles or in accordance with the facilities and requirements of the relevant system concerned) shall be good discharge to the Company. If any such cheque, warrant, order or other form of payment has or shall be alleged to have been lost, stolen or destroyed the Company shall not be responsible.

- 129.4 Any joint holder or other person jointly entitled to a share may give an effective receipt for any dividend, bonus, return of capital or other monies payable in respect of such share.
- 129.5 If a holder (or joint holder) does not specify an address, or does not specify an account or such other details and in each case that information is necessary in order to make a payment of a dividend, interest or other sum by the means by which in accordance with this Article the Board have decided that a payment is to be made or by which the holder (or joint holder) has validly elected to receive payment or the payment cannot be made by the Company using the details provided by the holder (or joint holders), the dividend, interest or other sum shall be treated as unclaimed for the purposes of these Articles.
- 129.6 The Board may, at its discretion, make provisions to enable any member as the Board shall determine to receive duly declared dividends in a currency or currencies other than sterling. For the purposes of the calculation of the amount receivable in respect of any dividend, the rate of exchange to be used to determine the foreign currency equivalent of any sum payable as a dividend shall be such rate or rates and the payment shall be on such terms and conditions as the Board may in its absolute discretion determine.

### 130. Uncashed dividends

If cheques, warrants or orders for dividends or other sums payable in respect of a share sent by the Company to the person entitled to them are returned to the Company or left uncashed on two consecutive occasions or, following one occasion, reasonable enquires have failed to establish any new address to be used for the purpose, the Company does not have to send any dividends or other monies payable in respect of that share due to that person until he or she notifies the Company of an address to be used for the purpose.

#### 131. Unclaimed dividends

All dividends, interest or other sums payable and unclaimed for 12 months after having become payable may be invested or otherwise made use of by the Board for the benefit of the Company until claimed. The Company shall not be a trustee in respect of such unclaimed dividends and will not be liable to pay interest on it. All dividends that remain unclaimed for 12 years after they were first declared or became due for payment shall (if the Board so resolves) be forfeited and shall cease to remain owing by the Company.

### 132. Scrip dividends

Subject to the Act, the Board may, by ordinary resolution of the Company and subject to such terms and conditions as the Board may determine, offer to any holders of ordinary shares (excluding any member holding shares as treasury shares) the right to elect to receive ordinary shares, credited as fully paid, instead of cash in respect of the whole (or some part, to be determined by the Board) of any dividend specified by the ordinary resolution. The following provisions shall apply:

- (a) the said resolution may specify a particular dividend, or may specify all or any dividends declared within a specified period or periods but such period may not end later than the third anniversary of the date of the meeting at which the ordinary resolution is passed;
- (b) the entitlement of each holder of ordinary shares to new ordinary shares shall be such that the relevant value of the entitlement shall be as nearly as possible equal to (but not greater than) the cash amount (disregarding any tax credit) of the dividend that such holder would have received by way of dividend. For this purpose **relevant value** shall be calculated by reference to the average of the middle market quotations for the ordinary shares on the Exchange, for the day on which the ordinary shares are first quoted "ex" the relevant dividend and the four subsequent dealing days, or in such other manner as the Board may determine on such basis as it considers to be fair and reasonable. A certificate or report by the Company's auditors as to the amount of the relevant value in respect of any dividend shall be conclusive evidence of that amount;

- (c) no fractions of a share shall be allotted. The Board may make such provisions as it thinks fit for any fractional entitlements including provisions where, in whole or in part, the benefit accrues to the Company and/or under which fractional entitlements are accrued and/or retained and in each case accumulated on behalf of any member and such accruals or retentions are applied to the allotment by way of bonus to or cash subscription on behalf of any member of fully paid ordinary shares and/or provisions where cash payments may be made to members in respect of their fractional entitlements;
- (d) the Board shall, after determining the basis of allotment, notify the holders of ordinary shares in writing of the right of election offered to them, and specify the procedure to be followed and place at which, and the latest time by which, elections must be lodged in order to be effective. No such notice need to be given to holders of ordinary shares who have previously given election mandates in accordance with this Article and whose mandates have not been revoked. The accidental omission to give notice of any right of election to, or the non-receipt (even if the Company becomes aware of such non-receipt) of any such notice by, any holder of ordinary shares entitled to the same shall neither invalidate any offer of an election nor give rise to any claim, suit or action;
- (e) the Board shall not proceed with any election unless the company has sufficient reserves or funds that may be capitalised, and the Board has authority to allot sufficient shares, to give effect to it after the basis of the allotment is determined;
- (f) the Board may exclude from any offer or make other arrangements in relation to any holders of ordinary shares where the Board considers that the making of the offer to them or in respect of such shares would or might involve the contravention of the laws of any territory or that for any other reason the offer should not be made to them or in respect of such shares;
- (g) the Board may establish or vary a procedure for election mandates in respect of future rights of election and may determine that every duly effected election in respect of any ordinary shares shall be binding on every successor in title to the holder;
- (h) the dividend (or that part of the dividend in respect of which a right of election has been offered) shall not be payable on ordinary shares in respect of which an election has been duly made (elected ordinary shares) and instead additional ordinary shares shall be allotted to the holders of the elected ordinary shares on the basis of allotment determined as stated above. For such purpose the Board may capitalise, out of any amount for the time being standing to the credit of any reserve or fund (including any share premium account or capital redemption reserve) or of any of the profits which could otherwise have been applied in paying dividends in cash as the Board may determine, a sum equal to the aggregate nominal amount of the additional ordinary shares to be allotted on such basis and apply it in paying up in full the appropriate number of unissued ordinary shares for allotment and distribution to the holders of the elected ordinary shares on such basis. The Board may do all acts and things considered necessary or expedient to give effect to any such capitalisation;
- (i) the Board may decide how any costs relating to the new shares available in place of a cash dividend will be met, including to deduct an amount from the entitlement of a holder of ordinary shares under this Article;
- (j) the additional ordinary shares so allotted shall rank *pari passu* in all respects with each other and with the fully paid ordinary shares in issue on the record date for the dividend in respect of which the right of election has been offered, except that they will not rank for any dividend or other distribution or other entitlement which has been declared, paid or made by reference to such record date; and
- (k) the Board may terminate, suspend, or amend any offer of the right to elect to receive ordinary shares in lieu of any cash dividend at any time and generally may implement any scrip dividend

scheme on such terms and conditions as the Board may determine and take such other action as the Board may deem necessary or desirable in respect of any such scheme.

#### 133. Capitalisation of reserves

The Board may, with the authority of an ordinary resolution of the Company:

- (a) subject as provided in this Article, resolve to capitalise any undivided profits of the Company not required for paying any preferential dividend (whether or not they are available for distribution) or any sum standing to the credit of any reserve or fund of the Company which is available for distribution or standing to the credit of the share premium account or capital redemption reserve or other undistributable reserve;
- (b) appropriate the sum resolved to be capitalised to the members in proportion to the nominal amounts of the shares (whether or not fully paid) held by them respectively which would entitle them to participate in a distribution of that sum if the shares were fully paid and the sum were then distributable and were distributed by way of dividend and apply such sum on their behalf either in or towards paying up the amounts, if any, for the time being unpaid on any shares held by them respectively, or in paying up in full unissued shares or debentures of the Company of a nominal amount equal to that sum, and allot the shares or debentures credited as fully paid to those members or as they may direct, in those proportions, or partly in one way and partly in the other, provided that:
  - (i) the share premium account, the capital redemption reserve, any other undistributable reserve and any profits which are not available for distribution may, for the purposes of this Article, only be applied in paying up in full shares to be allotted to members credited as fully paid;
  - (ii) the Company will also be entitled to participate in the relevant distribution in relation to any shares of the relevant class held by it as treasury shares and the proportionate entitlement of the relevant class of members to the distribution will be calculated accordingly; and
  - (iii) in a case where any sum is applied in paying amounts for the time being unpaid on any shares of the Company or in paying up in full debentures of the Company, the amount of the net assets of the Company at that time in not less than the aggregate of the called up share capital of the Company and its undistributable reserves as shown in the latest audited accounts of the Company or such other accounts as may be relevant and would not be reduced below that aggregate by the payment of it;
- (c) resolve that any shares so allotted to any member in respect of a holding by him or her of any partly paid shares shall, so long as such shares remain partly paid, rank for dividends only to the extent that such partly paid shares rank for dividends;
- (d) make such provision by the issue of fractional certificates (or by ignoring fractions or by accruing the benefit of it to the Company rather than to the members concerned) or by payment in cash or otherwise as it thinks fit in the case of shares or debentures becoming distributable in fractions;
- (e) authorise any person to enter on behalf of such members concerned into an agreement with the Company providing for either:
  - (i) the allotment to them respectively, credited as fully paid up, of any shares or debentures to which they may be entitled on such capitalisation; or
  - (ii) the payment up by the Company on behalf of such members by the application of their respective proportions of the reserves or profits resolved to be capitalised, of the amounts or any part of the amounts remaining unpaid on their existing shares, (any agreement made under such authority being effective and binding on all such members); and
- (f) generally do all acts and things required to give effect to such resolution.

#### 134. Record dates

- 134.1 Notwithstanding any other provision of these Articles but without prejudice to the rights attached to any shares and subject always to the Act, the Company or the Board may by resolution specify any date (**record date**) as the date at the close of business (or such other time as the Board may determine) on which persons registered as the holders of shares or other securities shall be entitled to receipt of any dividend, distribution, interest, allotment, issue, notice, information, document or circular. Such record date may be before, on or after the date on which the dividend, distribution, interest, allotment, issue, notice, information, document or circular is declared, made, paid, given, or served.
- 134.2 In the absence of a record date being fixed, entitlement to any dividend, distribution, interest, allotment, issue, notice, information, document or circular shall be determined by reference to the date on which the dividend is declared, the distribution allotment or issue is made or the notice, information, document or circular made, given or served.

### 135. Inspection of records

No member (other than a Director) shall have any right to inspect any accounting record or other document of the Company unless he or she is authorised to do so by law, by order of a court of competent jurisdiction, by the Board or by ordinary resolution of the Company.

### 136. Account to be sent to members

- 136.1 In respect of each financial year, a copy of the Company's annual accounts, the strategic report, the Directors' report, the Directors' remuneration report, the auditor's report on those accounts and on the auditable part of the Directors' remuneration report shall be sent or supplied to:
  - (a) Every member (whether or not entitled to receive notices of general meetings);
  - (b) Every holder of debentures (whether or not entitled to receive notice of general meetings);
  - (c) Every other person who is entitle to receive notice of general meetings; not less than 21 clear days before the date of the meeting at which copies of those documents are to be laid in accordance with the Act.
- 136.2 This Article does not require copies of the documents to which it applies to be sent or supplied to:
  - (a) A member or holder of debentures of whose address the Company is unaware; or
  - (b) More than one of the joint holders of shares or debentures.
- 136.3 The Board may determine that persons entitled to receive a copy of the Company's annual accounts, the strategic report, the Directors' report, the Directors' remuneration report, the auditor's report on those accounts and on the auditable part of the Directors' remuneration report are those persons entered on the Register at the close of business on a day determined by the Board, provided that the day determined by the Board may not be more than 21 days before the day that the relevant copies are being sent.
- 136.4 Where permitted by the Act, a strategic report with supplementary material in the form and containing the information prescribed by the Act may be sent or supplied to a person so electing in place of the documents required to be sent or supplied by Article 136.1.

## 137. Service of Notices

- 137.1 The Company can send, deliver or serve any notice or other document, including a share certificate, to or on a member:
  - (a) personally;

- (b) by sending it through the postal system addressed to the member at the member's registered address or by leaving it at that address addressed to the member;
- (c) where appropriate, by sending or supplying it in electronic form to an address notified by the member to the Company for that purpose;
- (d) where appropriate, by making it available on a website and notifying the member of its availability in accordance with this Article; or
- (e) by any other means authorised in writing by the member.
- 137.2 In the case of joint holders of a share:
  - (a) service, sending or supply of any notice, document or other information on or to one of the joint holders shall for all purposes be deemed a sufficient service on, sending or supplying to all the joint holders; and
  - (b) anything to be agreed or specified in relation to any notice, document or other information to be served on, sent or supplied to them may be agreed or specified by any one of the joint holders and the agreement or specification of the first named in the Register shall be accepted to the exclusion of that of the other joint holders.
- 137.3 Where a member (or, in the case of a joint holders, the person first named in the Register) has a registered address outside the United Kingdom but has notified the Company of an address within the United Kingdom at which notices, documents or other information may be given to him or her or has given to the Company an address for the purposes of communications by electronic means at which notices, documents or other information may be served, sent or supplied to him or her, the member shall be entitled to have notices served, sent or supplied to him or her at such address or, where applicable, the Company may make them available on a website and notify the holder of that address. Otherwise no such member shall be entitled to receive any notice, document or other information from the Company.
- 137.4 If on three consecutive occasions any notice, document or other information has been sent to any member at the member's registered address or the member's address for the service of notices (by electronic means or otherwise) but has been returned undelivered, such member shall not be entitled to receive notices, documents or other information from the Company until he or she shall have communicated with the Company and supplied in writing a new registered address or address within the United Kingdom for the service of notices or has informed the Company of an address for the service of notices and the sending or supply of documents and other information in electronic form. For these purposes, any notice, document or other information served, sent or supplied by post shall be treated as returned undelivered if the notice, document or other information is served, sent or supplied back to the Company (or its agents) and a notice, document or other information served, sent or supplied in electronic form shall be treated as returned undelivered if the Company (or its agents) receives notification that the notice, document or other information was not delivered to the address to which it was served, sent or supplied.
- 137.5 The Company may at any time and in its sole discretion choose to serve, send or supply notices, documents or other information in hard copy form alone to some or all of the members.

## 138. Notice on person entitled by transmission

The Company may give notice to the person entitled to a share because of the death or bankruptcy of a member or otherwise by operation of law, by sending or delivering it in any manner authorised by these Articles for the giving of notice to a member, addressed to that person by name, or by the title of representative of the deceased or trustee of the bankrupt or representative by operation of law or by any like description, at the address (if any) within the United Kingdom supplied for the purpose by the person claimed to be so entitled or to which notices may be sent in electronic form. Until such an address has been so supplied, a notice may be given in any manner in which it might have been given if the death or bankruptcy or operation of law had not occurred.

#### 139. Record date for service

Any notice, document or other information may be served, sent or supplied by the Company by reference to the register as it stands at any time not more than 15 days before the date of service, sending or supplying. No change in the register after that time shall invalidate that service, sending or supply. Where any notice, document or other information is served on, sent or supplied to any person in respect of a share in accordance with these Articles, no person deriving any title or interest in that share shall be entitled to any further service, sending or supplying of that notice, document or other information.

### 140. Evidence of service

- 140.1 Any notice, document or other information, addressed to a member at the member's registered address or address for service in the United Kingdom shall, if served, sent or supplied by first class post, be deemed to have been served or delivered on the day after the day when it was put in the post (or, where second class post is employed, on the second day after the day when it was put in the post). Proof that an envelope containing the notice, document or other information was properly addressed and put into the post as a prepaid letter shall be conclusive evidence that the notice was given.
- 140.2 Any notice, document or other information not served, sent or supplied by post but delivered or left at a registered address or address for service in the United Kingdom (other than an address for the purposes of communications by electronic means) shall be deemed to have been served or delivered on the day on which it was so delivered or left.
- 140.3 Any notice, document or other information, if served, sent or supplied by electronic means shall be deemed to have been received on the day on which the electronic communication was sent by or on behalf of the Company notwithstanding that the Company subsequently sends a hard copy of such notice, document or other information by post. Any notice, document or other information made available on a website shall be deemed to have been received on the day on which the notice, document or other information was first made available on the website or, if later, when a notice of availability is received or deemed to have been received pursuant to this Article. Proof that the notice, document or other information was properly addressed shall be conclusive evidence that the notice by electronic means was given.
- 140.4 Any notice, document or other information served, sent or supplied by the Company by means of a relevant system shall be deemed to have been received when the Company or any sponsoring system-participant acting on its behalf sends the issuer-instruction relating to the notice, document or other information.
- 140.5 Any notice, document or other information served, sent or supplied by the Company by any other means authorised in writing by the member concerned shall be deemed to have been received when the Company has carried out the action it has been authorised to take for that purpose.

## 141. Notice when post not available

If at any time by reason of the suspension, interruption or curtailment of postal services within the United Kingdom the Company is unable effectively to convene a general meeting by notices sent through the post, the Company need only give notice of a general meeting to those members with whom the Company can communicate by electronic means and who have provided the Company with an address for this purpose. The Company shall also advertise the notice in at least one national newspaper published in the United Kingdom and make it available on its website from the date of such advertisement until the conclusion of the meeting or any adjournment of it. In any such case the Company shall send confirmatory copies of the notice by post to those members to whom notice cannot be given by electronic means if, at least seven days prior to the meeting, the posting of notices to addresses throughout the United Kingdom again becomes practicable.

### 142. Indemnity and insurance

### 142.1 In this Article:

- (a) companies are **associated** if one is a subsidiary of the other or both are subsidiaries of the same body corporate;
- (b) a **relevant officer** means any Director or other officer or former director or other officer of the Company or an associated company (including any company which is a trustee of an occupational pension scheme (as defined by section 235(6) of the Act), but excluding in each case any person engaged by the Company (or associated company) as auditor (whether or not he or she is also a director or other officer), to the extent he or she acts in his or her capacity as auditor); and
- (c) relevant loss means any loss or liability which has been or may be incurred by a relevant officer in connection with that relevant officer's duties or powers in relation to the company, any associated company or any pension fund or employees' share scheme of the company or associated company.
- 142.2 Subject to Article 142.3, but without prejudice to any indemnity to which a relevant officer is otherwise entitled:
  - (a) each relevant officer shall be indemnified out of the Company's assets against all relevant loss including any liability incurred by the officer in defending any civil or criminal proceedings, in which judgment is given in the officer's favour or in which the officer is acquitted or the proceedings are otherwise disposed of without any finding or admission of any material breach of duty on the officer's part or in connection with any application in which the court grants the officer, in his or her capacity as a relevant officer, relief from liability for negligence, default, breach of duty or breach of trust in relation to the Company's (or any associated company's) affairs; and
  - (b) the Company may provide any relevant officer with funds to meet expenditure incurred or to be incurred by him or her in connection with any proceedings or application referred to in Article 142.2(a) and otherwise may take any action to enable any such relevant officer to avoid incurring such expenditure.
- 142.3 This Article does not authorise any indemnity which would be prohibited or rendered void by any provision of the Companies Acts or by any other provision of law.
- 142.4 The Directors may decide to purchase and maintain insurance, at the expense of the Company, for the benefit of any relevant officer in respect of any relevant loss.

### 143. Forum Selection

- 143.1 Unless the Company consents in writing to the selection of an alternative forum, the Courts of England and Wales shall, to the fullest extent permitted by law, be the sole and exclusive forum for:
  - (a) any derivative action or proceeding brought on behalf of the Company;
  - (b) any action, including any action commenced by a member of the Company in its own name or on behalf of the Company, asserting a claim of breach of any fiduciary or other duty owed by any director, officer or other employee of the Company (including but not limited to duties arising under the Act);
  - (c) any action arising out of or in connection with these Articles or otherwise in any way relating to the constitution or conduct of the Company; or
  - (d) any action asserting a claim against the Company governed by the "Internal Affairs Doctrine" (as such concept is recognised under the laws of the United States of America).

- 143.2 Unless the Company consents in writing to the selection of an alternative forum, the federal district courts of the United States of America shall be the sole and exclusive forum for the resolution of any complaint asserting a cause of action arising under the Securities Act or the Exchange Act.
- 143.3 For the avoidance of doubt, nothing contained in this Article 143 shall apply to any action or suits brought to enforce a duty or liability created by the Securities Act or the Exchange Act, or any claim for which the federal district courts of the United States of America are, as a matter of the laws of the United States, the sole and exclusive forum for determination of such a claim.

# 144. Winding up

- 144.1 If the Company is wound up, the liquidator may, with the authority of a special resolution and any other authority required by law, divide among the members in specie the whole or any part of the assets of the Company. This applies whether the assets shall consist of property of one kind or different kinds. For this purpose, the liquidator may set such value as the liquidator considers fair on any asset or assets and may determine how to divide it between the members or different classes of members. The liquidator may, with the authority of a special resolution and any other authority required by the law, transfer all or any part of the assets to trustees on such trusts for the benefit of members as the liquidator decides. Where the liquidator divides or transfers any assets in pursuance of the powers in this Article, no member shall be required to accept any asset in respect of which there is a liability.
- 144.2 Article 144.1 is without prejudice to any right or power that the liquidator may have, in the absence of the rights expressly conferred by Article 144.1, to divide or transfer the assets in specie as contemplated in Article 144.1 without a special resolution.